### RUSSIAN ACCENT IN RUSSIAN NATIVE SPEAKERS OF DANISH AS A SECOND AND FOREIGN LANGUAGE

Master thesis by Elena Dmitrievna Sokolova

> Supervisor Jakob Steensig

**Aarhus University** 

**Department of Aesthetics and Communication** 

February 2012

#### Summary

The research goal of this master thesis was two-fold 1) to find out what the most typical foreign accent features in Russian native speakers of Danish as a foreign or second language are on the segmental level and in the word stress assignment; and 2) to find out whether a special introductory phonetic training (SIPT) anticipating the main language course can mitigate the degree of a global foreign accent in late native Russian learners of Danish.

In order to reach these goals, I formulated predictions about eventual typical features of the Russian accent in Danish applying the method of contrastive analysis (Archibald, 1998; Lado, 1957; Whitman, 1970) of the Russian and Danish phonemic inventories, based on the distinctive features phonological theory (Chomsky and Halle, 1968), and compared the peculiarities of word stress assignment in Danish and Russian. The formulated hypotheses were then tested by means of a case-study method, namely an error analysis of recorded reading samples of two word lists (one with vowel and consonant targets (151 target words) and the other one with diphthongs and word stress targets (51 target words)) read by 18 adult subjects. Half of the subjects have studied Danish as a second language at Danish language schools and did not have any SIPT (the D-group); and half of the subjects have studied Danish as a foreign language at Moscow State Linguistic University and had SIPT (the R-group). All the subjects speak Russian as their L1 and have a high command of Danish. I transcribed the recorded samples and systematized all the errors separately for the D-and R-groups.

The error analysis substantiated my theoretical assumptions based on the theory of equivalence classification (Flege, 1987), Speech Learning Model (Flege, 1987) and the theory of spelling interference (Miglio & Fukazaw, 2006; Ehri & Wilce 1980) and I have verified the hypotheses about the following typical features of the Russian accent in the studied aspects:

- qualitative reduction of [a] to [v]\* or [ə]\*; [ε] to [ə]\*; [ä] to [v]\* according to the degrees of reduction typical of the allophones of Russian /a/ and /ε/;
- fewer quality distinctive properties of the back vowels and [i] vs. [e];
- shortening of long vowels (however, the latter depends on the type of the instruction learners receive; in the current study those subjects who had SIPT with a focus on the distinction between long and short vowels performed better in the reading task for the long targets);
- consonantization of non-syllabic elements [w] and [v] in the Danish diphthongs as [v]\* and [r]\*/[v]\* respectively;
- > monophthongization of diphthongs, especially [v]-diphthongs;
- > disaspiration of  $[b^h]$ ,  $[g^h]$  and [ds];
- ➤ voicing of segments [b], [d], [s] and [g] after a vowel;
- dentalization of Danish /d/, /s/, /t/ /n/;
- double primary stress is typically either ignored or set in a word with two primary stresses as if there were a secondary and main stress in this word;

secondary word stress is often ignored in non-compound words, and in compounds with more than two stems Russian native speakers tend to "save" the primary stress for the last stem in the word.

Additionally, other typical features, not predicted in the contrastive study, were discovered empirically as the results of the error analysis:

- front labialized [ø] is often mispronounced as [y]\*, especially under the influence of the spelling interference;
- > sounds  $[\alpha]$  and  $[\alpha]$  are generically susceptible to *narrowing* in terms of height to such qualities as  $[\phi]^*$ ,  $[\alpha]^*$ ,  $[\gamma]^*$  and  $[\Theta]^*$  as well as a tongue retraction;
- the nucleus of the diphthong may be exposed to the same qualitative errors as the corresponding vowel quality;
- > prefixed words may have a broken word stress.

The hypotheses about the lengthening of short vowels and about the velarization of /l/ were falsified. Moreover, my prediction about a more consonant-like pronunciation of [g] could neither be verified, but is sooner falsified, since the main accent feature for [g] was its omission rather than [g]-like production. Finally, one of my hypotheses was that Russian natives would palatalize /b, d, g/ especially before Danish front /i/, / $\epsilon$ /, /y/, /e/. The error analysis showed that this assumption was right for /g/, but I should admit that also sonorant segments, voiced [v] and voiceless [h] may be exposed to the palatalization. The reading task could not reveal characteristics of the pronunciation of [ə], which as was argued, should be studied not in isolated words, but rather in a spontaneous speech task. Generally, the current study claims that the discovered features are at least true under the conditions of the reading task. Further research is needed to test them in a spontaneous speech task.

As far as the second goal is concerned, it was reached by means of global accent ratings of the readings of a small text by 27 Russian natives (12 from the D-group and 15 from the R-group) and 4 Danish native controls. The global accent ratings were done by four native raters with linguistic backgrounds and four native raters without any linguistic background, according to a 5-point scale. The difference in the mean scores of the two groups (1.80 for the D-group and 2.17 for the R-group) was proved to be statistically significant as the result of a T-test run on statistical significance. It was calculated that considering the experiment conditions and taking into account the linguistic portraits of the two target groups, the probability that SIPT (focused both on the segmental and prosodic aspects of the Danish pronunciation) mitigates a foreign accent in Russian native speakers of Danish, is 94.7%.

It should be noted that the results of the thesis have a practical significance for both teachers of Danish working with Russian natives, and for the Russian native learners of Danish, as a set of guidelines about weak points of the Russian natives' pronunciation in Danish. It can serve a basis for the development of a pronunciation course focused on the segmental level and word stress assignment, and can be considered as a recommendation to introduce SIPT for beginners with Danish as their L2 and Russian as L1.

## **Table of Contents**

1. Intro	oduction	9
1.1. P	Problem overview	9
1.2. C	Goals of the project	10
2. Fore	eign accent as a research issue	11
2.1. F	Foreign accent, interference or transfer?	11
2.2. Т	Theoretical framework	13
2.2.1.	. Distinctive features phonological theory	13
2.2.2.	. Previous studies on "accent factors"	14
2.2.3.	. Speech Learning Model (SLM) and the theory of equivalence classificat	tion16
2.2.4.	. Theory of spelling interference	19
2.3. H	Hypotheses and methodological framework	20
2.4. N	Notation conventions for the current project	23
3. Contr	rastive study of the Danish and Russian phonological systems	on the
segment	tal level and the level of word stress	
3.1. Vov	wels	25
3.1.1.	. Danish vs. Russian vowels: distinctive features of the two phonemic invent	tories 25
3.1.2.	. Russian vowel reduction degrees and their possible transfer into Danish	34
3.2. Dip	phthongs	37
3.3. Di	istinctive articulation features of Russian and Danish consonant planes	honemic
3.4 Wo	ard strong in Danish and Pussian	
5.4. W0	ating ampirical data	
4. Collec		45
4.1. Pre	Word list 1	
4.1.1.	Word list 2	43
4.1.2.	Text sample	40
4.1.3.	vcedure	47
4.2.110	D-vs -R taxonomy	47 A7
422	Recording Procedure	/ 
4 3 Sub	hiects	49
5 Data	analycic	
5.1 Tra	anary 515	51
53 Err	or analysis	

5.3.1. Vowel features	
5.3.1.1. Front and central full vowels	53
5.3.1.1.1. Qualitative features	53
5.3.1.1.2. Quantitative features	60
5.3.1.2. Back vowels	62
5.3.1.1.1. Qualitative features	63
5.3.1.1.2. Quantitative features	66
5.3.1.3. Shortening of long vowels	68
5.3.1.3. Neutral [ə]	68
5.3.1.4. Diphthong features	
5.3.1.4.1. Consonantization of the glide	70
5.3.1.4.2. Nucleus replacement	71
5.3.1.4.3. Monophthongization	71
5.3.2. Consonant features	72
5.3.2.1. Disaspiration	73
5.3.2.2. Voicing, assimilation and dentalization	75
5.3.2.3. Palatalization and velarization	79
5.3.2.4. Other consonant features	
5.3.4. Word stress features	
5.3.4.1. Double primary word stress	
5.3.4.2. Secondary stress in compound and non-compound words.	
5.3.4.3. Broken word stress in prefixed words	
6. Global accent rating	86
6.1. Rating procedure and rating method	87
6.2. Raters	87
6.3. Results	
Final conclusions and discussions	
References	
Appendices	
Appendix 1. Questionnaire for subjects	
Appendix 2. Word List (WL1). Vowel and consonant segments	
Appendix 3. Target vowel segments in WL1	
Appendix 4. Target consonant segments in WL1	111
Appendix 5. Word list 2 (WL2). Words stress and diphthongs	
Appendix 6. Target words for the analysis of word stress assignment ir	n WL2112
Appendix 7. Target diphthongs in WL2	113

Appendix 8. Transcription results for the vowel targets in the R-group114
Appendix 9. Transcription results for the vowel targets in the D-group115
Appendix 10. Transcription results for the diphthong targets in the R-group117
Appendix 11. Transcription results for the diphthong targets in the D-group119
Appendix 12. Transcription results for the consonant targets in the R-group120
Appendix 13. Transcription results for the consonant targets in the D-group122
Appendix 14. Transcription results for the word stress assignment in the R-group124
Appendix 15. Transcription results for the word stress assignment in the D-group125
Appendix 16. Global accent rating sheet
Appendix 17. Correspondence between PPNs and numbers in126
the global accent rating sheet
Appendix 18. Correspondence between transcription symbols
in IPA, Den Danske Ordborg and Dania127
Appendix 19. Transcriptions of the vowel and consonant targets in the R-group128
Appendix 20. Transcriptions of the vowel and consonant targets in the D-group135
Appendix 21. Transcriptions of the diphthong and word142
stress targets in the R-group
Appendix 22. Transcriptions of the diphthong and word stress targets in the D-group144
Appendix 23. T-test on the statistical significance of the difference in mean scores of the D- and R-groups
Appendix 24. Levels of the Danish language programme for adult foreigners according to the Common European Language Framework
Appenix 25. Recordings

#### 1. Introduction

#### 1.1. Problem overview

The topic of this master thesis is "Russian accent in Russian native speakers of Danish as a second and foreign language". The point of departure for the choice of the topic has been a difficulty, which as my experience of a teacher of Danish a second language shows, the majority of late Russian native learners of Danish as both second and foreign language have to face while mastering the Danish pronunciation. The latter in most cases is a much more time- and effort demanding aspect of Danish studies, compared to the vocabulary and grammar learning or reaching a general fluency of speech in Danish. The most illustrative example, I have witnessed myself, of what this unsolved difficulty may result in, is a situation when a late learner has spent years on learning Danish, but when he or she starts communicating with a Dane, they fail to understand each other. One reason for that is the late learner's heavy accent. Another one is that the Danes are in general less accustomed to hear a foreign variant of their native language, and as a consequence have less "practice" in distinguishing a foreign variant of Danish. Such an experience of not being-understood may develop into a psycholinguistic barrier in future in the case of Danish end-learners or put an obstacle on the studying process in the case of beginners by forming a negative perception of the Danish language.

Numerous research experiments, which I discuss in sections 2.1., 2.2., have addressed the fundamental idea that phonologically the foreign accent basis lies on the word segmental level, i.e. in the major dissimilarities between two phonemic inventories, namely the articulatory properties of the L1 vs. L2 vowels and consonants. Among major segmental dissimilarities between Russian and Danish are the absence in Russian of the distinction long-vs.-short vowels, absence of the aspiration for the stops /p/, /t/, /k/; absence in Danish of the consonant categories voiced vs. voiceless and lateralized vs. non-lateralized inherent to Russian consonants, as well as a smaller range of vowels in Russian, compared to Danish.

These are only few examples of the differences on the segmental level. No native-like pronunciation, according to Birdsong (2007: 117), is possible on the global sentence level, if an articulatory word level is affected by a foreign accent. However, no foreign accent-free pronunciation can be realized only with an accurate articulation on the segmental level, since a native-like pronunciation is characterized by a whole set of features including prosody and syllable structure. I argue however, that the core of the foreign accent is on the segmental level.

One of the methodological principles of accent studies most widely applied during the recent decades (Flege, 2002; Best et al. 2001; Flege et al. 1995; Flege 1981a; Ingram & Park, 1998; McAllister et al., 2002; Missaglia, 1999) is the one, which implies testing narrow segmental and suprasegmental foreign accent features proceeding from more abstract predictions about difficulties which non-native speakers may have, as the result of filtering the sound and prosody systems of their second language (L2) through their first language (L1) corresponding systems. No previous research has addressed the phenomenon of Russian accent in Danish, neither on the abstract level of basic dissimilarities, nor on a more precise level of acoustic or prosodic features of the Russian accent. Therefore, this research paper is to tackle the very fundamental aspects of the Russian accent in Danish. I shall address the issue of the Russian accent by first making more abstract predictions about how dissimilarities between Danish and Russian phonemic inventories may be reflected in the Russian accent, and then shall test my predictions empirically.

The current project will be the first one in the field of accent studies examining the combination Russian (L1 in this study) - Danish (L2 in this study). It may also have significance for future studies in the field of foreign accent in Danish learners with other Slavonic languages as their L1s. Moreover, the results of this thesis can make a considerable contribution to the methodology of the Danish language teaching by finding the weak sides of the Russian natives' articulation in Danish, which could be used as a first-priority aspect in the phonetic training.

#### **1.2.** Goals of the project

A lot of research work has been done in the field of studying foreign accents from the point of view of factors that influence the degree of a foreign accent, such as the age of L2 learning, nature of L2 phonetic input, length of residence in an L2-speaking country, gender and motivation, type of language formal instruction, and amount of native language use (Piske et al., 2001). These factors can be referred to as common linguistic and extra-linguistic ones, very often interrelated with sociolinguistic conditions, and that are of a more general character. Such an approach to accent studies seems to be reasonable, but addresses the questions: "Why do people speak with accent? What could been done to give next generations a chance to minimize the accent?"

The main goal of the current project, however, is to answer the following questions:

1) What are the most typical accent features in Russian native speakers with Danish as a foreign and second language on the segmental level and in the word stress assignment?

2) Can a special introductory phonetic training anticipating the main language course mitigate the degree of a global foreign accent<sup>1</sup> in late native Russian learners of Danish?

In order to answer these two questions I shall resolve the two tasks.

- Describe major phonological features of the Russian accent in Danish on the segmental level (pronunciation of vowels and consonants), and on the segmental level - accent properties connected with the assignment of the word stress.
- 2) Analyze a global accent degree in two different groups of subjects: 1) in those who have received a special introductory phonetic training before their main language course and have studied Danish as a foreign language in Russia; and 2) in those who have been taught Danish pronunciation as an integrated part of their language course and studied Danish as a second language in Denmark. I shall further in section 2.3. discuss my methodological decision to examine these two groups and explain why in the current project these two groups are considered initially equal in terms of the foreign accent factors.

#### 2. Foreign accent as a research issue

#### 2.1. Foreign accent, interference or transfer?

Before examining methodological issues of the current project, it seems natural to define basic terminological conventions regarding the term "accent" as the object of our primary research. The term "accent" or alternatively "foreign accent" in the context of second language acquisition and teaching theories means a set of phonological characteristics of a non-native pronunciation. In a broad sense, not in the meaning "…property of a syllable which makes it stand out in an utterance relative to its neighboring syllables…" in various domains: word accent (also word stress or lexical stress), phrase stress or sentence accent (http://www.britannica.com/EBchecked/topic/2866/accent), accent is "the cumulative auditory effect of those features of pronunciation which identify where a person is from regionally or socially" (Crystal 2003: 3). In our case, we shall deal with the regional accent, namely the Russian one as a set of phonological properties that make Danish speech sound

<sup>&</sup>lt;sup>1</sup> "The degree to which an L2 speaker's productions are perceived to differ from those of a native speaker" (Riney et al. 2000: 713).

Russian-like due to the L1 (Russian) pronunciation habits both, on the level articulation, acoustics and prosody.

A foreign accent is also often discussed in connection with such phenomena as language transfer or language interference. The language transfer is usually defined as "the influence of a person's first language on the language being acquired." (Crystal 2003: 471). The language transfer is a more general notion compared to a foreign accent and usually stands for applying rules and knowledge of the L1 grammar, vocabulary, spelling or even punctuation in the L2. As a rule, a language transfer takes place as the result of insufficient knowledge of an L2, absence of a native-like command of an L2, or merely lack of authentic input. An example could be for instance a situation when the L1 syntactic system dominates over that of the L2, and an L2 learner tends to apply syntactic patterns untypical of the L2, which can lead to a non-native syntax and even semantic errors. In this case, the transfer of the L1 elements and structures will have a negative nature, because the influence of the L1 will lead to major or minor violations of the L2 norms or usage.

A notion similar to the negative transfer is a linguistic interference. Grosjean (1992) distinguishes between a static and dynamic interference. The static interference "describes the relatively permanent influence of one of the bilingual's languages on the other" (Malmkjær 2001: 69). According to Grosjean (1992), the common areas of static interference are accent, intonation and the pronunciations of individual sounds, such as a constant devoicing of final voiced English constants by Russian natives. The dynamic interference usually implies a temporarily transferred feature, only occasionally both in a written and spoken language.

The linguistic interference is most often mentioned in connection with the interlanguage theory, best presented by Larry Selinker (1992). An interlanguage is a transitional stage in the L2 learning, when the learners an L2, according to Selinker (1996: 97) "produce structures that exist neither in their first language, nor in the language they are learning and which (it seems) no native speaker of any language ever produces". It is possible to assume that the interlanguage structures also imply pronunciation patterns, characterized by mixed properties of the L1 and the L2, due to the fact that learners process and percept the L2 system through his or her solid and well-established L1 system, which puts an obstacle on the way to a native-like pronunciation in his or her L2.

As was mentioned above, the language transfer often has a negative effect on an L2 learner's linguistic performance. However, the language transfer can also have a positive effect through borrowings, for instance. Such a transfer can undoubtedly help a learner in the comprehension and production in his or her L2.

As we can see, the phonological transfer may be defined as applying the L1 pronunciation habits in the L2, in other words, the process of bringing L1 phonological features into the L2. Compared to a foreign accent, the phonological transfer, especially in its negative realization is an inevitable process in the L2 learning, rather than a result, while a foreign accent is one of its areas of realization resulting in a set of end-point features inherent to this or that phonological transfer as a process.

Since my project is aimed at figuring out what the main peculiarities of the Russian speech in Danish on the above mentioned levels are, I shall further use the term "foreign accent", not transfer or interference, and for the methodological convenience shall figuratively consider the Russian accent as a separate "variant of foreign Danish" with specific properties. I shall further also use the term global foreign accent in the meaning of "the degree to which a L2 speaker's productions are perceived to differ from those of a native speaker" (Riney et al. 2000: 713).

#### 2.2. Theoretical framework

Though this project is the first one regarding Russian accent in Danish, it is of course not the first one in the field of foreign accent studies in general. Therefore, the current investigation will rely on many of the recent studies in the field of foreign accent research, phonological theory and theory of the second language phonology.

#### 2.2.1. Distinctive features phonological theory

Since my project addresses a foreign accent as a research issue, it implies that I deal with two phonological systems. Therefore, I shall further - particularly in the contrastive study of the Danish and Russian phonological systems (see section 3) on the levels of segments and word stress - proceed from the distinctive feature phonological theory (Hall, 2001).

This theory was developed by Jakobson and his colleagues (1954) and further elaborated by Chomsky and Halle (1968) (http://clas.mq.edu.au/phonetics/phonology/features/index.html#distinctive). The theory implies "that speech sounds are composed of smaller abstract categories called distinctive features..." (Mielke & Hume 2006: 723). The term "distinctive features" as Mielke & Hume (2006: 723) put it, is applied in the sense of properties that "are used to define natural classes

of sounds, describe sound patterns, and to form contrasts". Both Chomsky and Halle, and Jakobson et al. (1954) argue that, distinctive features are defined in terms of some phonetic property (Hall 2001: 3). However, according to Hall (2001: 4), Chomsky and Halle (1968) take the point of view that the features are defined solely on articulatory terms, while Jakobsonian approach suggests that distinctive features have primarily acoustic definitions. The latter approach has been recently supported by, for example, Flemming (1995), Boersma (1998), Steriade (2000) who argue that acoustic and auditory form a basis of the distinctive features.

In my project, I support Chomsky and Halle (1968)'s approach to the definition of distinctive features and shall further proceed from the distinctive articulatory properties of the sounds in Russian and Danish (see section 3). I argue that the latter is more relevant for my foreign accent study, and I have two arguments for that.

Firstly, this thesis will have the largest practical significance for the late learners of Danish as a second and foreign language and their teachers if in connection with the analysis of typical accent features I proceed from the articulatory phonetic basis for distinctive features. This is reasonable from the practical point of view: in a classroom, or individually by Russian learners, the highlighted accent properties could be first and foremost tackled and eliminated or "improved" on an articulatory level, and only then on the level of acoustics. It is hard to imagine that a learner of a foreign or a second language would first operate with acoustic features of the L2 sounds – he or she would rather like to learn how to use his or her articulatory apparatus for the production of this or that L2 sound.

Secondly, in my thesis I set a particular focus on the production of sounds, i.e. how the Russian learners' foreign accent manifests itself in the language production, and examine what articulatory mistakes stand behind the accent. I believe that an accent study based on the distinctive acoustic features would be rather more relevant for the analysis of a foreign accent in terms of how accent correlates with learners' auditory perception of an L2, but the latter is not the topic of this thesis.

#### 2.2.2. Previous studies on "accent factors"

One of the fundamental aspects in accent studies is the so-called *factors' issue*. As I have mentioned above, foreign accent factors are not the point of departure in our research. However, it is impossible to gather and analyze data without taking into consideration the linguistic backgrounds of the subjects, which are indirectly correlated with the factors'

determining the degree of accent. Therefore, below I shall give a short overview of the accent factors for the purpose of composing questionnaires for participants in the case-study as a part of the project.

The key-factors that have been in the focus of recent studies are the following:

- *age of L2 learning* (Bongaerts et al., 1997; Thompson, 1991; Lund, 2003; Munro et al., 1996), and in this connection the support (Patkowski, 1990) and counter-evidence (Flege, 1987) for the Critical period hypothesis (CPH)<sup>2</sup>; however, in general, the age-factor is rather the result of multiple factors that co-vary with the age at which an L2 learning began.
- *length of residence in an L2-speaking country* as a not necessarily significant factor (Flege, 1988; Thompson, 1991; Elliott, 1995; Moyer, 1999), and usually a less important predictor of the L2 accent degree (Flege & Fletcher, 1992), but the one, that can be a crucial contribution to the accent decrease at the initial stage of L2 learning (Riney & Flege, 1998);
- *motivation* such as a professional motivation, integrative motivation or a strength of concern for the L2 pronunciation accuracy which do not automatically lead to an accent-free pronunciation in the L2 (Piske et al., 2001);
- *innate aptitude for the oral mimicry* (Purcell & Suter, 1980) that may positively influence the decrease in accent and facilitate the L2 learning;
- type of language formal instruction (Piske et al., 2001) as an insignificant factor except for a special training in pronunciation for late learners (Bongaerts et al., 1997; Moyer, 1999), of which prosody-centered training (suprasegmental) was found to have improved pronunciation more efficiently than segmental training (Missaglia, 1999);
- *amount of native language use* as a minor, but still an important factor in decreasing a foreign accent (Flege et al., 1999b; Thompson, 1991).

<sup>&</sup>lt;sup>2</sup> According to Bongaerts et at. (1995) the Critical period hypothesis was first proposed by Wilder Penfield and Lamar Roberts, and was popularized by Eric Lenneberg in 1967 with *Biological Foundations of Language*. According to Lenneberg (1967: 180), "there are maturational constraints on the time a first language can be acquired. First language acquisition relies on neuroplasticity. If language acquisition does not occur by puberty, some aspects of language can be learnt but full mastery cannot be achieved".

Thus, applying CPH to a SLA theory, we can say that if a child learns hie or her L2 before the critical age for L1, he or she will have a native command of the both; whereas the later in life L2 is studied, the fewer chances are that a L2's learner will have a native-like command of L2.

#### 2.2.3. Speech Learning Model (SLM) and the theory of equivalence classification

Other research areas within accent studies of the last decades have been centered around the correlation between the degree of a foreign accent and speech production (Anderson-Hsieh et al., 1992), the degree of a foreign accent and comprehension in the L2 (Anderson-Hsieh, 1988) as well as between the degree of a foreign accent and perception in the L2 (Flege & MacKay, 2004; Baker et al., 2002). In all these research areas, the point of departure was dissimilarities between the L1 and L2 phonological systems (Baker et al., 2002; McAllister et al., 2002; Grønnum, 2008; Flege, 1981; Flege et al., 2003; Strange, 2007; etc.).

In this thesis, I have the same point of departure. I shall further - in the error analysis on the segmental level - refer to and apply one of the most frequently used models to explain an accent basis, called *Speech Learning Model (SLM)*, developed by James, E. Flege. "Focusing at the segmental level, SLM attributes foreign (non-native-like) accent to the learner's tendency to classify into a pre-existing phonic category an L2 sound that is acoustically similar to an L1 sound" (Birdsong 2007: 100). The core term of Flege's model is *"equivalence classification*" and the idea behind this term is that

..."equivalent" or "similar" sounds are difficult to acquire because a speaker perceives and classifies them as equivalent to those in the L1 and no new phonetic category is established, whereas "new" (dissimilar or different) sounds are easier to learn because the speaker perceives these differences and establishes new phonetic categories (Major 38: 2001).

In his SLM, Flege has gone further than the contrastive analysis hypothesis (CAH) (Lado, 1957). CAH stated that "cross-language differences result in learning difficulty..." and that "...that learners of an L2 will have more difficulty learning a new sound that has no equivalent in the L1 than in learning an L2 speech sound that resembles (but is not physically identical) an L1 sound" (Aoyama et al. 2004: 235).

Flege (1987) did not deny CAH, but he has claimed that it is only relevant at the initial stages of L2 learning. Since the current study addresses the issue of accent in advanced learners, I shall apply contrastive analysis (Archibald, 1998; Lado, 1957; Whitman, 1970) as a point of departure but in the error analysis refer to Flege's SLM.

Flege (1995) suggests that in the second language learning both at the level of perception and production, L1 sounds phonetically dissimilar to L2 sounds will be learned and perceived with more accuracy in the long-term of L2 learning, in advanced learners. At initial stages, a L2 learner may be misled by a "temptation" to ignore accuracy in production and perception of L2 sounds, which are most similar to those of his or her L1. Psycholinguistically, this may be caused by a L2 learner's strive to speak L2 fluently, resulting in ignoring very minor dissimilarities of the most-similar sounds. Additionally, similar sounds are much more difficult to be phonologically processed in search of acoustic and articulation differences because of the established and automatized pronunciation and perception habits developed for the L1 phonological system.

According to Flege (1995), similar segments are filtered through the phonetic categories of the L1 sound inventory, and "...several different L1 speech sounds might be used as substitutes..." (Aoyama et al. 2004: 245), compared to the dissimilar or non-existing-ones in the L1, for which a new phonetic category will be created during a second language learning.

I used Venn diagrams to depict the idea of the accuracy progress in the pronunciation of similar and dissimilar sounds according to SLM in the following way: see **Figure 1**.



Figure 1. Accuracy progress in the pronunciation of similar and dissimilar sounds according to SLM

Let us illustrate SLM with an example of Russian and Danish. In the case of *accent 1* let us take, for instance, Danish  $/\delta$ / as in *gade, hedde, bide*, which does not exist in Russian, neither as a separate phoneme nor as an allophone. The articulation of this sound is certainly a challenge for a native Russian, since it does not admit any "equivalence classification", and when it is pronounced for the first time, it resembles something between /l/ and dental-alveolar /d/. In the case of *accent 2*, let us take Danish vowel phoneme /o/ as in *bopæl, olie, god.* The latter one resembles a sound between Russian /o/ and /u/, but usually Russian natives pronounce it as the one closer to the Danish /u/, i.e. as a more closed sound. The accuracy of articulation in the case of this Danish segment is not a goal of an utmost importance for a Russian, since in most cases a context can "compensate" the missing phonemic distinction.

However, this negligence as the result of a 99% -"equivalence classification" is exactly what makes the accented articulation of the Danish sound plausible. Thus, *accent 1* and *accent 2* have completely different "origins", and during the initial stage of Danish acquisition a Russian speaker will be weaker at pronouncing the challenging /ð/, rather than /o/, but gradually as mastering the peculiarities of /ð/'s articulation, on the "tabula rasa", his or her accuracy of /o/ will be lacking behind, and will take much more time to become similar to the native-like pronunciation.

Though numerous studies of the last decades carried out by Flege himself (1987, 1995, 1999, 2002) and other researchers (Best, 1995; Best et al., 2001; Ingram & Park, 1998; etc.) support the idea of a greater achievement for dissimilar sounds, especially in advanced learners. There are studies, which cast doubt on SLM. For example, Kim (1994) found "that both advanced and beginning speakers of Korean learners of English performed better for the similar sounds" (Major 2001: 39). Earlier, Bohn & Flege (1992) in their study of the production of new and similar vowels by adult German learners of English showed that some of the speakers performed better with the similar sounds.

In this connection, Major (Major & Kim, 1996) proposed another model and suggested to speak not in terms of the difficulty of acquiring sounds, but rather in terms of the rate of acquisition. He claimed that, "dissimilar phenomena are acquired at faster rates than similar phenomena..." (Major 2001: 39). This laid the ground for the Similarity Differential Rate Hypothesis (SDRH) that has been supported by a series of studies: Chabanova, 1997; DeGaytan, 1997; Riney & Flege, 1998; Major & Kim, 1996). The latter was focused on Korean learners of English and showed that "sound [j] was produced better by both beginning and advanced students (beginners did better) than the dissimilar sound [z], but comparing the beginning and advanced students it was clear that the rate of acquisition for the dissimilar sound was faster than for the similar sound..." (Major 2001: 39).

What is interesting is that even though SDRH and SLM look at accent from different angels, neither of them deny the fact that dissimilar sounds are an advantage for a L2 learner in the end, either in terms of forming new categorical properties (SLM) of the dissimilar sounds, or in terms of the rate of acquisition (SDRH). Both approaches give grounds for emphasizing the dissimilarities and similarities between the Russian and Danish segments. However, I shall further refer mainly to Flege's SLM and his theory of "equivalence classification". I believe that Flege's approach, substantiated by the above-mentioned studies, forms a more relevant theoretical basis for my project because

- I study a foreign accent in late advanced learners both in similar and dissimilar sounds, but do not compare the rate of acquisition of similar and dissimilar sounds in advanced and begging students;
- This thesis should describe accent features rather than investigating what sounds are acquired at a faster rate;
- Flege's theory of equivalence classification explains mental processes that stand behind the acquisition of the L2 phonemic inventory and the way new pronunciation habits develop in late advanced learners.

#### 2.2.4. Theory of spelling interference

Another major theoretical point of departure for my accent description is the theory of *spelling interference* (Miglio & Fukazaw, 2006; Ehri & Wilce 1980) in L2 learners (here I apply the term "second language in the meaning a language learned after L1"). According to this theory pronunciation mistakes, and as a consequence a speech with a foreign accent" could not be generically ascribed to L1 interference, but more specifically to the spelling interference from the L1" (Miglio & Fukazawa 2006: 4145). Miglio & Fukazawa (2006) substantiated their idea empirically having made a research on American learners of Spanish. The two authors found out recurrent patterns in the subjects' pronunciation errors. These patterns were the result of the Spanish learners' word recognition on analogy in a word list reading task, when they had unconsciously associated the letters from the Latin alphabet,

used in Spanish to represent Spanish sounds, with English sound. That is to say, in the words pronounced with an accent, they processed the spelling as if it represented English sounds. Thus, a spelling interference is a complex "phenomenon whereby the spelling of the word…" in L2 "…triggers a correspondence between…" an L2 spelling symbol "…and the pronunciation of the same symbol in the native language…" (Miglio & Fukazawa 2006: 4145).

The reason why I consider this theoretical approach particularly relevant for my research is that firstly, learning Danish pronunciation through a written language, i.e. by using the principle from letter (symbol) - to sound is a very common practice among Danish learners with Russian as L1, as my teaching experience shows. In other words, Russian learners tend to pronounce Danish words in the way closest to the spelling, assumedly due to a closer letter-to-sound correspondence in the Russian morphophonemic spelling system. Secondly, this learning through a written language creates a basis for a spelling interference situation. The latter takes place when a Danish spelling symbol triggers a Russian and Danish letters a, o, y, k; or 2) because the Danish Latin spelling symbol triggers the "corresponding" Cyrillic symbol, and as a result the Danish sound is pronounced in a Russian-like manner (e.g. in the case of Danish p, t, d, g, h etc.). I shall further in 5.3. refer to the *spelling interference theory* in order to explain the origin of some Russian accent features discussed.

#### 2.3. Hypotheses and methodological framework

As was mentioned the aim of this thesis is twofold: 1) to describe the typical features of the Russian foreign accent in Russian natives with Danish as a foreign and second language; 2) to find out whether a special phonetic training anticipating the main language course plays an accent-mitigating role. To do that, I have formulated a series of preliminary hypotheses that I am going to proceed from. They are the following.

- 1) Russian accent in Danish would have the following typical features on the segmental level and in the word stress assignment:
  - excessive and unnecessary qualitative reduction of unstressed vowels;
  - shortening of the long Danish vowels and lengthening of the short ones;
  - fewer quality distinctive properties of the back vowels and front /i/ and /e/;
  - monophthongization of diphthongs;

- consonantization of the glide in diphthongs;
- disaspiration of /p/, /t/, k/;
- velarization of /l/;
- voicing of non-aspirated consonants /b/, /d/, /g/, also of the intervocalic [s].
- palatalization of /b/, /d/, /g/ especially before Danish front /i/, / $\phi$ /, / $\epsilon$ /, /y/ and /e/.
- dentalization of /d/, /s/, /t/ and /n/;
- /r/-assimilation to a thrilling /r/; "consonantization" of [g];
- [ts]-overtone in the Danish  $/t/^3$ .
- avoidance of the secondary word stress;
- replacement of one of the double word stresses by a secondary one.
- 2) Russian native learners of Danish as a foreign language, exposed to a special introductory phonetic training (SIPT) before the main language course, would have a lower degree of the global accent than Russian native learners of Danish as a second language, who studied pronunciation as an integrated part of the major language instruction and did not have any SIPT before the main language course.

To narrow and precise the hypotheses for the segmental level and word stress assignment I need a preliminary theoretical study in order to examine dissimilarities and similarities between the two phonological inventories in terms of segmental characteristics and word stress features. Methodologically, I shall carry out this study in the form of a traditional inter-linguistic contrastive analysis (CA) (Archibald, 1998; Lado, 1957; Whitman, 1970). The theoretical CA should result in more precise assumptions and predictions about accent features, which then will be either verified or falsified empirically by means of a casestudy method - namely the error analysis of the recorded samples of word lists read by Russian learners of Danish as a foreign and second language. Moreover, the eventual "stumbling blocks" for Russian learners, highlighted in the CA will help to make methodologically more solid word lists for the reading task. I shall study the case of adult learners of Danish, who started learning Danish as a second or a foreign language after age 17. There will be two groups of subjects.

<u>Group 1 (further referred to as "R-group"):</u> Russian native speakers, who are studying Danish as their minor foreign language at Moscow State Linguistic University, and

<sup>&</sup>lt;sup>3</sup> See section 3.3.

have an advanced command of the language. This group received SIPT before the main curriculum, in the first semester.

<u>Group 2 (further referred to as "D-group"):</u> Russian native speakers, who studied or are still studying Danish as a second language at language schools and centers in Denmark. They have an advanced command of Danish. In the case of this group of subjects, the pronunciation training has been integrated in the module teaching, but was not taught as a separate introduction course prior to the main language course<sup>4</sup>.

In order to substantiate or falsify my hypothesis about the accent mitigating role of SIPT, I methodologically *imply* that the R-group would run the same chances to be rated approximately in the same way as the D-group. On the one hand, the longer mean length of the language instruction in the case of the participants with the special introductory phonetic training would compensate for the fact that they have not lived in a language environment where Danish is an official (and a majority language), as opposed to the other group with Danish as a second language. On the other hand, the subjects with Danish as a second language by default would assumedly have more chances to put their language skills in language practice with native speakers mitigate the Russian accent by a larger native input exposure and an active use of Danish outside language schools. The latter should compensate for their not having SIPT. Therefore, I initially *imply* the two groups to be equal. However, if the R-group receives higher mean scores, this would assumedly mean that SIPT was a decisive factor, but this will be tested in the case study. I shall further in section 4.3. present the subjects of both groups and their linguistic portraits in a more detailed way.

The case study will consist of two major stages, which correspond to the goals of the project:

**Stage 1:** Error analysis of the recorded samples of word lists read aloud by subjects from both groups. Subjects will be offered to read two lists of isolated Danish words, covering the full Danish phonemic inventory (see sections 4.1.1., 4.1.2. for a more detailed account on preparing the reading materials). The recorded samples of word lists will be then transcribed with the help of the International Phonetic Alphabet (further IPA), and the mispronounced sounds will be exposed to the error analysis in order to systematize errors and figure out the typical features of the Russian accent in Danish in the target case-study groups.

<sup>&</sup>lt;sup>4</sup> Only few language schools in Denmark offer special pronunciation training prior to the first study module, and such courses are usually designed for students with L1s typologically very different from Danish, such as Chinese, for instance.

**Stage 2:** Rating by Danish native expert and non-expert speakers of the Russian subjects' global accent degree by means of assessing the recorded readings of a small text, according to a 5-point rating scale. The obtained ratings will be then averaged separately for each group in order to find out whether SIPT plays an accent-mitigating role. See sections 6.1., 6.2. for more details on the rating procedure.

For methodological reasons, Danish raters will be from two groups: those who do not have any special linguistic background - who did not major/are not majoring in Danish or any foreign languages and who do not have any language-teaching experience ("non-experts"); and those who major/majored in Danish or foreign languages, or have a language-teaching experience ("experts"). My decision to choose raters with these two different backgrounds was based on the previous research, mainly on Flege's (1984) and Thompson's (1991) studies. These studies showed that linguistically "inexperienced raters are more stringent in rating the degree of accent" (Thompsons 1991: 198). In order to make the global accent rating procedure as objective as possible, I decided to choose both expert and non-expert raters.

Methodologically, I decided to choose the reading tasks for data collection for both stage 1 and stage 2, rather than spontaneous speech tasks. I have a series of arguments for that. Firstly, by preparing lists of isolated words I could cover all the main allophones of the Danish language. Spontaneous speech samples could not ensure that in all the samples, all the target sounds would occur at least once. Secondly, the reading task allowed us to choose target words in a way that target sounds would occur in the most illustrative positions from the point of view of a potential accent basis for these sounds according to the Russian phonological properties. Thirdly, a word lists reading task ensures that all the subjects are in equal conditions and would have the same level of task difficulty, whereas in the case of a spontaneous speech task subjects could on purpose avoid using words containing sounds, which are particularly difficult for them. Finally, the same reading materials for all subjects give a chance to compare the results across the two target groups (those with SIPT and without it).

#### 2.4. Notation conventions for the current project

In the theoretical contrastive analysis, I shall apply the modern Russian literary language as a reference example for the Russian language, and the pronunciation of the "standard" Copenhagen dialect as a reference example for the Danish language. The Russian literary language is considered (Cubberley, 2002) today a standard variant of Russian spoken by educated Russians across the Russian Federation. It is the official language of the Russian government, radio and television. All the subjects participating in our case study are all speakers of the modern Russian literary language.

All the Russian examples are incorporated in the paper by means of transliteration in the Latin alphabet. It should be noted that we use the modern conventionalized rules of Cyrillic-Latin transliteration designed by Yermolovich (2005).

The phonetic transcriptions of the recorded word lists samples, as well as all the examples given in the text of the thesis will be given in accordance with the IPA (Handbook of the International Phonetic Association, 1999). I decided to use the IPA both for Russian and Danish in order to unify the transcription notation systems under one convention. Moreover, the IPA is more efficient in depicting the sounds of the Russian variant of Danish where phonemes and their allophones existing neither in Danish nor in Russian will definitely turn up. In this case the number of transcription signs of, for instance, the Dania Phonetic Alphabet (further DPA) would not be able to reflect all the non-native like variants and "new-generated" sounds (a learner's interlanguage sounds) inherent to the pronunciation of Danish words. Therefore, the IPA with its larger inventory seems to be a more methodologically reasonable solution, and a tool allowing a more precise transcription.

However, since most of the Danish dictionaries apply the Dania Phonetic Alphabet (further DPA), the correspondence between the IPA and Dania is presented in Appendix 18. Other notational conventions used in the project are the following:

Russian examples: *italics*, *non-bold type* 

Danish examples: *italics, bold type*.

English translation: non-italics, non-bold type.

# **3.** Contrastive study of the Danish and Russian phonological systems on the segmental level and the level of word stress

#### 3.1. Vowels

#### 3.1.1. Danish vs. Russian vowels: distinctive features of the two phonemic inventories

In this section, I investigate what the differences between the two vowels inventories are on the segmental level, mainly what the distinctive features of the vowel phonemes in both languages are. By the term *distinctive features*, I shall further imply the definition given in section 2.2.1. (I examine only articulatory features, as was mention in 2.2.1). Additionally, since this study addresses the Danish pronunciation in late Russian native learners, I consider it relevant to supplement the articulation distinctive features (as a basis for comparison) firstly, with the length - as a phonologically meaningful quantitative feature of Danish vowels. Secondly, I shall also discuss the weakening of vowels (qualitative reductions in the unstressed syllables) as a vowel feature highly relevant for the Russian accent in Danish, which will be addressed in section 3.1.2.

In Russian, there is a crucial interaction between consonant and vowel segments. The articulation properties of the latter are in most cases dependent on the quality of a following or preceding consonant, for example, on whether this consonant is hard or palatalized. Moreover, there is a major influence of the word stress on the quality of the phoneme. The former and the latter characteristics result in a rich allophony. Therefore, when discussing Russian vowel phonemes, linguists (Avanesov, 1956; Bondarko, 1977) cannot ignore the allophony, and as a rule, the overview of the Russian vowel phonemic inventory is focused on the presentation of the positional variants of vowel phonemes and description of phonemic rows.

The aim of our contrastive study is not to describe how the positional principle works for the allophonization in Russian and Danish, but rather to highlight distinctive articulation properties of the Russian vowels compared to the Danish ones.

However, I shall further sometimes refer to the Russian vowel allophony, because the Russian positional principle may have an effect on the accent in Danish. I shall mention two main positional domains of the Russian phonemes (not taking into account the isolated position) and their subtypes:

#### A. STRONG POSITION:

stressed vowels at the beginning of the word before a hard consonant;

stressed vowels at the beginning of the word before a palatalized consonant; stressed vowels after a hard consonant before a palatalized consonant; stressed vowels after a hard consonant not before a palatalized consonant; stressed vowels after a palatalized consonant before a palatalized consonant; stressed vowels after a palatalized consonant not before a palatalized consonant;

#### B. <u>WEAK POSITION</u> (reduced)

unstressed vowels in the first pre-stressed syllable; unstressed vowels in the second pre-stressed syllable; unstressed vowels in the first post-tonic syllable.

The positions A and B play a distinctive role in defining the allophonic properties of 5 Russian stressed distinctive vowel phonemes /a/, / $\epsilon$ /, / i/, /u/ and /o/. There are still discussions concerning the sound [i]. Moscow linguistic school considers this sound to be the allophone of /i/ in the position after hard (non-palatalized) consonants, while St. Petersburg linguistic school regards them as two separate phonemes. In this thesis, I support the first point of view, because [i] and [i] exist in complementary distribution and are never interchanged.

Regardless of the positional characteristics, these five strong stressed full phonemes /a/,  $\epsilon$ /, / i/, /u/ and /o/ have their distinctive contrastive articulation properties in respect to their place of articulation (see **Table 1**):

- height (vertical dimension) of the body of the tongue in relation to the hard palate (close, mid and open);
- participation of the lips in their articulation (labialized/rounded, unlabialized/unrounded).

Phoneme	Height	Backness <sup>5</sup>	Rounded-ness	Examples in IPA
1. /i/	close	front	_	osina [ɐˈsʲ <b>i</b> nə]
				(an aspen)
2. / <b>u</b> /	close	back	+	<i>ruki</i> [ˈr <b>u</b> k <sup>j</sup> ɪ]
				(hands)
3. /ɛ/	open-mid	near-front	_	<i>belka</i> [ˈb <sup>i</sup> elkə]

<sup>&</sup>lt;sup>5</sup> Grey filed shows articulation properties, which are not phonemically distinctive.

				(a squirrel)
4. /o/	mid	back	+	korobka
				[keropkə]
				(a box)
5. / <b>a/</b>	open	mid/near-back	_	gladky [glaṯk <sup>j</sup> ɪj]
				(smooth)

#### **Table 1. Russian full vowels**

In the discussion of Russian full vowel phonemes, it is only relevant to speak about full phonemes in relation to the stressed vowels. Such a feature can be ascribed to the influence of the word stress on the quality of the Russian vowels. What will distinguish allophones' reference to their phonemes, and therefore the meaning of the word on the articulatory level, is the labialization and height.

Let us take minimal pairs. In (1) the only meaningful distinction between mid  $\epsilon$  and mid /o/ is provided by labialization, where [e] in *sel* is an allophone of  $\epsilon$  after a palatalazied consonant, and [ $\theta$ ] is an allophone of /o/ after a palatalized consonant. In (2) the phonemic difference is preserved by the vowels' height – close /i/ and open /a/.

(1)	<i>sel</i> [s <sup>j</sup> el] <i>sit</i> (past, pf, 3sing, male)	<i>syol</i> [s <sup>i</sup> <b>θ</b> ] rural settlements (neuter, pl., Gen.)		
(2)	<i>tik</i> [t <sup>j</sup> <b>i</b> k] a tic (Nom., sing., male)	<i>tak</i> [tak] so		

It is worth noting, that the height and labialization remain, according to Avanesov (103: 1956), fundamental and constitutive distinctive features of all the allophones of one phoneme, which make them belong to the same phoneme irrespective of the immediate distribution and phonological processes such as assimilation.

What is remarkable is that backness - the position of the tongue in relation to the back of the mouth (and in this connection division into front, central and back) - traditionally pointed out in the distinctive features theory as a property determining the vowel's quality - is not a constituent and distinctive articulation property of the Russian full vowel phonemes. It is a complimentary one, and, according to Avanesov (1956: 89), characterizes the quality of an allophone rather than that of a phoneme. If we take, for example, the open /**a**/, the position of the tongue in relation to the back of the mouth is vague in an isolated position or at the beginning of a word. Since the body of the tongue is more or less flat, its articulation involves neither moving the body of the tongue forward nor stretching it backwards to the velum. Only in definite realizations, for example after hard velar /k/, /g/ as in *kartofel* [kar'tof<sup>i</sup>el<sup>j</sup>], (a potato), *gadat* [ge'dat<sup>j</sup>] (to tell smb's fortune) the body of the tongue or mainly the back part of it stretches a little bit backwards to the velum. Another good example is the front /*i*/. When occurring after a hard vowel as in *bistro* [bistrə] (quickly), *ryskat* [rigkat<sup>j</sup>] (to prowl) it is realized as the mid allophone /*i*/, and the same allophone can be more front after a dental consonant as in *dynya* [din<sup>j</sup>ə] (melon), or becomes more back and even diphnongized after a labial consonant plus [ł] *plyt* [płit<sup>j</sup>] (to swim). Thus, it is possible to conclude that backness in Russian is inherent to allophones and depends on whether a vowel sound is followed and/or precedied by a consonant phoneme, and if so - on the qualities of the following and/or preceding consonants (hard/palatalized).

As far as such a phonological quantitative characteristic as length is concerned, it is not distinctive or contrastive in Russian. It only plays an emphatic role when a particular vowel is in the focus for the purpose of a more distinct pronunciation, during a syllable reading or in particular positions, as for instance, the case with the phoneme /a/ before /l/. The stressed vowels are usually slightly longer, but the latter is not a phonological feature in Russian and has nothing to do with the contrast long vs. short phonemes.

Compared to the Russian vowel phonemic inventory the Danish one is much more varied. This is probably a major first-sight distinction between the two systems. Russian with its five full phonemes (occurring in stressed positions) is lacking considerably behind in the size of the vowel phonemic inventory compared to Danish with 20 full phonemes (occurring in stressed syllables). The latter lays the foundation for a non-native-like pronunciation in the Russian learners.

What is common of two systems is that the place of articulation plays a contrastive and distinctive role in determining phonemic characteristics. As well as in Russian, the backness as a property of Danish vowels is a distinctive and contrastive quality of the allophones rather than phonemes. This quality is positionally determined and may slightly vary from an allophone to allophone, but not beyond the limits of the contrastive phonemic reference. I shall further refer to the traditional cardinal classification of the vowel phonemes with two distinctive articulatory phonemic contrastive features height and roundedness, just as I did with the Russian vowel phonemes. There is another approach to the articulatory classification

of vowels, first elaborated by Ladefoged (1971) and then widely used by, for instance, Basbøll (2005) for the classification of vowel "space" characteristics. It implies the division of phonemes into labial, palatal, velar, pharyngeal, approximant and front. Methodologically, I argue in favor of the traditional cardinal classification, which in our study can provide a common ground for comparison of the two inventories: namely because the traditional cardinal classification is much more frequently used in the Russian phonology (Avanesov 1956; Bondarko, 1977).

Let us now look at the Danish inventory of vowel phonemes, according to Grønnum (2005: 62) with our own examples. See **Table 2**.

It is worth noting that the length, as a contrastive property of Danish vowel phonemes, is considered by some linguists, such as Basbøll (2005) to be syllable-related and suprasegmental by nature in Danish. It is true that long Danish vowels occur typically in an open syllable, thus, to a certain extent, are syllable-related and syllable-determined, and sometimes even dependent on "morpho-syntactic conditions" (Grønnum 2005: 251). Therefore, the vowel length is prosodic (suprasegmental) by nature. However, it seems reasonable and relevant to consider this quantitative property of Danish phonemes to be relevant for the current segment-related contrastive study, and further set shortening of the Danish vowels in the focus of the error analysis. The first reason for that is that this vowel quantitative phonological characteristic does not exist in Russian. Therefore, it would be a particular challenge in pronunciation. And secondly, as my teaching experience shows, the vowel length (suprasegmental by nature) is most effectively assimilated by Russian learners when it is discussed as a "segmental" feature and trained in minimal pairs. Therefore, here the vowel length will be included in the error analysis along with the segmental properties of the vowels.

Vowel phoneme Height		Backness <sup>6</sup>	Roundedness	Examples in IPA
1. /i:/	close	front	-	pige [ˈb̥ʰiːə]
2. /i/	close	front	-	ligge [ˈliɡ̊ə]
3. /eː/	close-mid	front	-	alene [ạˈlęːnə]
4. /e/	close-mid	front	-	det [de]
5. /εː/	close-mid	front	-	næse [ˈneːsə]
6. /ε/	close-mid	front	-	pædagog [þʰed̥ạ̯ˈɡ̊ou̯²]
7. /aː/	open-mid	front	-	bade [ˈŷạːð̪ʲə]
8. /a/	open-mid	front	-	<i>mad</i> [mąǧץ]
9. /yː/	close	front	+	<i>dyne</i> [ˈd̥yːnə]
10. /y/	close	front	+	<i>cykel</i> [ˈsyɡ̊əl]
11. /øː/	close-mid	front	+	<b>købe</b> [ˈɡ̊ʰø̞ːb̥ə]
12. /ø/	close-mid	front	+	kysse [ˈɡ̊ʰø̯sə]
13. /œ:/	open-mid	front	+	gøre [ˈģœ̞ːʌ̯]
14. /œ/	open-mid	front	+	høns [hæns]
15. /u:/	close	back	+	<i>bruge</i> [ˈb̥ʁฺ <b>u</b> ːə]
16. /u/	close	back	+	<i>huske</i> [husgႆə]
17. /oː/	close-mid	back	+	skole [ˈsɡ̊o̪ːlə]
18. /o/	close-mid	back	+	olie ['qljə]
19. /ɔː/	open-mid	back	+	<b>dåse</b> [ˈd̥ <b>ҙ</b> ːsə]
20. /ɔ/	open-mid	back	+	komme [ˈġʰʌᢩmə]

#### Table 2. Danish short and long full vowel phonemes according to Grønnum (2005: 62)

Another syllable-related distinctive characteristic of Danish vowels is called  $st\phi d$ . The latter is, according to Basbøll (2005: 83), "a syllabic prosody, a laryngealization – a kind of creaky voice...- often beginning somewhere near the middle of certain syllables..." Stød is lexically distinctive in Danish: e.g. *aftale*, vb. ['äw, $ds\epsilon$ ?lə] (to agree) and *aftale* sb. ['äw, $ds\epsilon$ :lə] (an agreement). The general rule is that in order to receive stød a stressed syllable should have a long vowel or a short vowel plus a sonorant consonant, in other words should have a "stød-basis" (Basbøll 2005: 84).

Even though stød is a semantically meaningful phonological feature in Danish, it cannot be heard in all the geographical variants of Danish. According to Fischer (1992), it is

<sup>&</sup>lt;sup>6</sup> Grey filed shows articulation properties, which are not phonemically distinctive.

possible to speak about a stød-border, which lies north of Rømø via Tønder to Haderslev and then from Fåborg northwards up to Præstø and farther to Bornholm (see **Figure 2**). To the south of this border stød cannot be heard at all, and to the north of the border, it exists, but not in all variants (Kirk 2008: 75).



Figure 2. Geografical occurrence of stød in modern Danish according to Heger (1992: 125)

The ability or inability to produce stød as a prosodic and non-segment, but segmentrelated distinctive property of a syllable will not be taken into account in this thesis, neither theoretically, nor empirically for certain pragmatic reasons. Arguing idealistically, it is possible to assume that if the speech of the Danes without stød is considered by the Danes who produce stød to sound native-like, then other phonological features of the segmental and suprasegmental levels are more "significant" than stød in terms of assigning native-likeness. The latter however does not mean that it is all the same whether the learners of Danish acquire stød or not (Kirk 2008: 75). Moreover, in the context of Danish as a foreign or second language the acquisition of a native-like or at least regular production of stød is not a common practice, even in the learners of Danish with a high language command. As Henrichsen (2009) claims, "two specific aspects of Danish pronunciation are perceived by L2 learners as particularly hard to master, the stress assignment, and the stød"). I suggest to assume that an advanced and intermediate learners are aware of this phonological feature, but in a real language use it turns out to be a secondary priority compared to the segment articulation. Thus, stød will not be in the pragmatic focus of my research.

Earlier in section 2.2.4., I spoke about the influence of the Danish written language on Danish acquisition by Russian learners. In Danish language schools (as my own teaching experience and internship show) and in Russia (according to the information from teachers of Danish as a foreign language at Moscow State Linguistic University, for instance<sup>7</sup>) the instruction in segments' pronunciation is usually letter-to-sound-related, and vowel qualities in this connection are presented and trained proceeding from their immediate phonetic distribution. Among these distribution factors are the following ones: 1) whether a vowel is preceded or followed by /r/; 2) whether the syllable is opened or not; 3) as well as more rarely on whether a vowel is followed by a particular consonant. Since the error analysis of the reading samples in the current study will address errors on the phonetic level, it seems relevant to supplement the above presented Danish vowel phonemic overview with the illustration of the Danish vowel allophony, according to (Grønnum 2001: 45), depending on the immediate phonetic distribution. Moreover, this allophony illustration reflects the pronunciation guidelines, which the learners of Danish usually receive. See Table 3 for the Danish vowel allophony illustration. The vowel qualities with stød are not included in the allophony illustration.

phonemes	allophonic manifestation						
	before r	after r	otherwise				
/i:/	[iː] svire	[iː] prise	[iː] <i>mile</i>				
/i/	[i] <i>birk</i>	[i] <i>ridt</i>	[i] <i>mit</i>				
/e:/	[eː] <i>mere</i>	[ɛː] <i>kredse</i>	[ęː] <i>mele</i>				
/e/	[e] <b>Per</b>	[ε] <b>brik</b>	[e̯] <i>midt</i>				
		[ɛː] <b>k<i>ræse</i></b>	[eː] <i>mæle</i>				
/ɛː/	[εː] <i>være</i>	[äː] before [ð̪ˠ] græde					
/ɛ/	[ɛ] <b>bær</b>	[a] <b>bræk</b>	[e] <i>mæt</i>				
/a:/	[äː] <i>vare</i>	[äː] <i>rase</i>	[ε:] <i>male</i>				
		[ä] <i>brak</i>	[a̯] <i>mat</i>				
			[ä] before labial and dorsal, e.g.				
/a/	[ä] <i>var</i>		lak, lam				
/y:/	[yː] <i>fyre</i>	[yː] <i>rype</i>	[y:] <i>syne</i>				
/y/	[y] <b>dyrk</b>	[y] <b>rytter</b>	[y] tyst				
/ø:/	[øː] <b>køre</b>	[œː] <i>røbe</i>	[øː] <i>føne</i>				

<sup>&</sup>lt;sup>7</sup> One of the basic course books on pronunciation applied at Moscow State Linguistic University is *Min udtale* by Søgaard (1999).

		[œ] ryste	[ø] øst
	[øː] <i>mørne</i>	[æ] before [j] <i>drøj</i>	[œ] before [j] <i>tøj</i>
/ø/		[œ] before [w] <i>røv</i>	[ø] before [w] øvrig
/œ:/	[œː] <i>gøre</i>		[œː] <i>høne</i>
/œ/	[œ] <i>gør</i>	[Œ] <b>grøn</b>	[œ] <i>høns</i>
/u:/	[uː] <i>kure</i>	[oː] <i>ruse</i>	[uː] <i>mule</i>
/u/	[u] skurk	[o̯] <i>brusk</i>	[u] <i>mut</i>
/o:/	[oː] <i>more</i>	[oː] <i>rose</i>	[oː] <i>mole</i>
	before C [o]	final [o] <i>ro( 'bot)</i>	final [o] <i>foto</i>
	sort	before C [ɔ̯] <i>rust</i>	before C [o] ost
/0/		Ĩ	1
/ɔː/	[ɔ̣ː] <i>båre</i>	[ɔ̣ː] <i>råbe</i>	[ɔ̣ː] <i>måle</i>
		[ʌ̯] k <i>rop</i> [ə̯]	[ʌ̯] <i>kop</i>
/3/	[ɔ̞] <i>vor</i>	before [w] <i>rov</i>	[ɔ̞] berore [w] <i>tov</i>

Table 3. The illustration of Danish vowel allophony according to (Grønnum 2001: 245)in IPA symbols

Proceeding from the contrastive overview of the articulatory distinctive features of the Russian and Danish vowel phonemic inventories, it possible to make the following conclusions about eventual "stumbling blocks" for the Russian learners.

• The Russian vowel phonemic inventory is considerably less rich than the Danish one that makes a first challenge for the Russian learners of Danish in terms of the acquisition of new phonemes and their allophones. See Figure 3.



Figure 3. Russian (in blue) and Danish (in red) vowel phonemes: cardinal scheme

• It is possible to assume that according to SLM model, back close /u/ would be probably the most native-like of all phonemes, while the distinction between /i/ and

/e/ would be a challenge due to the fact that the main allophone [i] of the Russian /i/ after a hard consonant will be most similar to the Danish /e/, even though [i] is rather a central one.

- According to SLM, absolutely new phonemes /y/, /ø/, /œ/ and correspondingly their allophones will be produced in a more native-like manner, though with a less distinction between /y/-/ø/.
- As far as the Danish /a/ is concerned, its main allophone [a] will be presumably mispronounced as a more closed allophone, while allophone [a] will be more nativelike.
- The most challengeable will be the distinctive articulation of back vowels /o/ and /ɔ/, and allophones of the latter.
- The major quantitative distinction is the absence in Russian of the contrastive opposition long vs. short vowels. As a result, the production of long vowel sounds will be a weaker point of a Russian native learner, while the short vowel sounds would be assumedly less difficult since they exist in Russian. It should be noted, that this "equivalence classification effect" in case of the long vowels according to Flege's SLM would be probably a vivid exception, since the principle "new and thus easier" will not assumedly function, as it would with new phonemes. The reason for that is the prosodic nature of the vowel length, and thus it requires much more training and adaption to the new pronunciation habits.

#### 3.1.2. Russian vowel reduction degrees and their possible transfer into Danish

As was mentioned earlier, the Russian vowel phonemes are subject to a qualitative reduction in unstressed positions. I shall further use the term "reduction" in the sense "qualitatively weakened, but not omitted". This term should be distinguished from the term "reduction" often applied in Danish literature (Kirk 2008: 130) in the meaning that a sound (as it comes from Kirk's examples, both consonant and vowel sounds) is not pronounced, i.e. reduced to zero.

Russian full vowel phonemes are always subject to at least minor qualitative reduction if they are unstressed. In Danish, the weakening of unstressed vowels is to a certain extent different from that one in Russian. If in Russian reduction is more or less systematic, in Danish it is highly dependent on an individual's speech rate, pronunciation habits, emotions, and stylistic features (formal/informal). Generally speaking, according to Grønnum (2001: 153)

...vowels are pronounced more distinctly in stressed syllables...than in unstressed syllables... In Danish the difference in vowel quality (here the difference between stressed and unstressed vowels is meant) is not so marked, except for the fact that we have one vowel, [ə], which only occurs in unstressed syllables. However, in other languages... certain vowels are obviously centralized in unstressed syllables. (Grønnum 2001: 153).

Thus, as far as the qualitative reduction is concerned, in Danish it is more relevant to speak about it implying the reduction to neutral [ə] for instance in final positions as in *sine* ['si:nə], *snakke* ['snäĝə], *hoppe* ['hə̯bə]. In general, the neutral [ə] in final positions would be a typical characteristic of Danish and generally in final position there are very few phonological oppositions between vowels left. The qualitative reduction in pre-stressed syllables would preserve more of a vowel quality in terms of phonological oppositions, but still would be often dependent on an individual speech rate and style.

Grønnum's reference to the languages where unstressed vowels are centralized is particularly to the point for Russian. From the point of view of accent, the vowel reduction is a very significant characteristic of the Russian vowel system, and my hypothesis, which will be tested in the error analysis, is that the general mechanisms of vowel reduction may be more or less transferred into Danish. Therefore, below I shall give an overview of the vowel reduction mechanism in Russian.

The Russian vowel reduction, according to Bondarko (1977), may be of two different degrees: *Degree 1*, and *Degree 2*. *Degree 1* is typical of the first pre-tonic syllable and is usually considered to be less qualitatively severe. *Degree 2* occurs in all other unstressed syllables. Now let us see in **Table 4** how these two degrees of reduction work with particular phonemes and give examples.

Phoneme	Stressed	First pre-tonic	Other unstressed syllables			
	position	syllable				
		(Degree 1)	(Degree 2)			
			a) second pre-tonic syllable;			
			b) all post-tonic ones			
/i/	<i>ig ·ry</i> (games)	<i>ig</i> • <i>ra</i> (a game)	a) pri-ras-tat (to grow/increase,			
	[ 1gr+ ]	[Ig ra]	ipf.) [pr <sup>j</sup> ırɐˈstat <sup>j</sup> ];			
		NB! after C /i/ can	b) vy• <b>ra</b> •schi•vat (to cultivate,			
		be reduced to [1].	ipf.) [vɨrɐɛːɪvət <sup>i</sup> ]			
/u/	$ug \cdot ol$ (a corner)	$u \cdot gly$ (corners)	a) u+tram+ <b>bo</b> +va+ny			
	[ ugəl]		(to tamp: participle, past indf.,			
		NB! after $C^j / u / can$	passive) [otrem'bovənij];			
		be reduced to [u].	b) po• <b>mo</b> •gut			
			(will help: 3 pl., future indf.)			
			[pɐˈmogʊt]			
/ɛ/	be·gal (ran: past,	<i>be</i> • <i>zhat</i> (to jog/to	a) <i>pe+re+me+na</i> (a change/ a			
	ipf, 3sing, male) ['b <sup>j</sup> egəl]	run, ipi.) [biˈzat <sup>j</sup> ]	break) [p <sup>j</sup> ər <sup>j</sup> ıˈm <sup>j</sup> ɛnə];			
			b) <i>o</i> . <i>le</i> . <i>nem</i> (a deer: sing,			
			instrumental) [vl <sup>j</sup> en <sup>j</sup> əm]			
/0/	$do \cdot ro \cdot ga$ (a road)	<i>ko</i> . <i>rabl</i> (a ship)	a) $do \cdot go \cdot vor$ (an agreement)			
	[dɐˈrogə]	[ke rabl']	[dəgɐˈvor];			
		NB! after C <sup>j</sup> /o/ can	b) <b>ry</b> .boj (fish, sing, instrumental			
		be reduced to [1].	[ˈrɨbəj]			
/a/	<i>po•da•rok</i> (a gift)	$o \cdot da \cdot ryen \cdot y$	a) <i>pa</i> . <i>ra</i> . <i>lel</i> (a parallel)			
	[pəˈdarək]	[ədɐˈrʲonɨj]	[pərɐˈl <sup>j</sup> el <sup>j</sup> ]			
			b) <i>vo•<b>ro</b>•bush•ka</i> (a sparrow, sing.			
		NB! after C' /a/ can be reduced to [1].	diminuitive) [veˈrobʊshkə]			
		e.g.				

Table 4.	The	degrees	of	vowel	redi	iction	in	Russian
		achieco	<b>U</b> .					

As is seen from **Table 4**, *Degree 1* and *Degree 2* reductions in the case of phonemes /i/ and /u/ are slightly centralized compared to the stressed /i/, /u/. The Russian  $\epsilon$ / in *Degree 1* reduction becomes more front and more closed and may only have allophonic variants [1] after C<sup>j</sup> or [i] after C. In *Degree 2* reduction,  $\epsilon$ / becomes somewhat similar to the Danish
neutral [ə]. However, the latter is more closed, whereas the Russian is more open. What is remarkable about /o/ and /a/ is that they merge into [v] in *Degree 1* reduction or [ə] in *Degree 2* reduction. Another important point is that in a weak position only close and mid vowel are distinguished.

The question is what both degrees of reduction mean for the Danish pronunciation. Firstly, if Russian the vowel reduction mechanism is transferred into Danish, mostly susceptible to the reduction of height will be Danish open unstressed phonemes /a/, /o/ and mid / $\epsilon$ / and their corresponding allophones both in the post-tonic and pre-tonic positions. Since the Russian mid "merged" phoneme / $\epsilon$ / of the first pre-tonic is more closed than these three Danish vowel phonemes, the principle "do not forget to open your mouth wider" will be a prevailing one in the correction of Russian native learners, because the due height of these three Danish vowels will not be probably realized in articulation.

As for the Danish /o/, I assume that it would be in a more advantageous position compared to /a/, and /ɔ/, because it is initially more closed than the Russian /o/, and if reduced in height will be even more native-like. However, the latter is not claimed but only hypothetically suggested and will be studied as a part of error analysis. The weakening of the vowels will be further considered as an important factor in choosing target Danish words for the reading task. For more details on the reading materials and the principle of target word selection, see sections 4.1.1., 4.1.2.

## **3.2. Diphthongs**

According to Jones & Ward (2011), Russian has diphthongs, and they all are falling diphthongs<sup>8</sup>, i.e. they all end in non-syllabic [j/i]:

[aj] as in *gajka* (a screw);

[oj] as in *bojko* (readily/briskly);

[ij] as in krasnyj (red);

[uj] as in *bujnyj* (turbulent/lush);

[ej] as in *lejka* (a watering pot).

There are no diphthongs in Russian with the initial component  $[\varepsilon]$  because in Russian  $[\varepsilon]$  cannot occur before a soft consonant, and a semivowel [j] plays the same role as the soft

<sup>&</sup>lt;sup>8</sup> In falling diphthongs, the first element is syllabic, while the second one is non-syllabic (Heger, 1992).

consonant /j/ does. (Jones & Ward 2011: 75). Jones & Ward (2011) also argue that the first part of diphthongs is subject to the same allophony as their constituent vowels, as well as to the reduction in the unstressed positions in a fast speech.

What is common of Danish (Grønnum 2001: 255) and Russian (Hickey, 1986) diphthongs is that they are phonetic by nature. However, compared to Russian, Danish beyond falling diphthongs ending in [w], [j], [g], e.g., [gw], [yw], [ew], [gw], [gw], [gw], [gi], [aj], [aj], [aj], [aj], [aj], [aj], [aw/au], [ig], [eg], [eg], [gg], [gg],

It is possible to assume that since no falling diphthongs ending [w] and [v] are found in Russian, this kind of diphthongs would be a potential difficulty for Russian natives. I assume that this may result in the spelling-induced consonantization of the glides [w] and [v], according to the theory of spelling interference mentioned in section 2.2.4., when the corresponding letters v and r would trigger sounds [v] and [v]. However, if we take into consideration, that learners are aware of the vocalization of /r/ after a vowel, then, another scenario may take place and the result would be a total omission of the element [v] in diphthongs and, thus, [v]-diphthongs may be subject to monophthongization. These two hypotheses will be tested in the error analysis of the reading samples.

# **3.3.** Distinctive articulation features of Russian and Danish consonant phonemic inventories

Both Russian and Danish inventories of consonant phonemes are characterized by a series of similar distinctive articulatory features: these two are the manner and place of articulation. What makes the two inventories distinctive, is their particular supplementary articulation properties standing in binary or non-binary oppositions, such as a binary opposition palatalization vs. non-palatalization as well as voiced vs. voiceless in Russian; aspiration (occurs only in the syllable-initial position) vs. non-aspiration in Danish.

**Table 5** gives an overview (Basbøll 2005; Avanesov 1956) of Danish (in red) and Russian (in blue) consonant phonemes. The consonant phonemes are presented in a single table for the purpose of comparison. The Russian palatalization is shown by "<sup>j</sup>" in accordance with the IPA, (except from /c:/, /s/, /z/, /z:/ that have separate symbols for the pairs palatalized non-palatalized) and the absence of palatalization by the absence of this sign. For

<sup>&</sup>lt;sup>9</sup> I do not take into the analysis diphthongs with  $st\phi d$  by methodological reasons mentioned in section 3.1.

<sup>&</sup>lt;sup>10</sup> In rising diphthongs, the first element is non-syllabic, while the second one is syllabic (Heger, 1992).

the Danish consonants the position-determined aspiration is shown, following Basbøll's (2005: 64) example, by means of a post-posted hyphen, that correspond to their ability to be aspirated only in a syllable initial position. Since almost all Danish consonant are voiceless this will be a default quality in the table, the few voiced Danish consonant phonemes will be shown with "<sup>v</sup>" only for the purpose of comparison, though this taxonomy is not typical of the IPA. **Table 5** also shows the main allophones (in parentheses) of the Danish phonemes, because in the error analysis the main consonant qualities, but not phonemes will be analyzed (as was the case with vowel allophony in section 3.1.)

As is seen from **Table 5**, consonant phonemes in both languages have articulation features within the common contrastive system of place (bilabial, labial-dental, dental, alveodental, alveolar, alveopalatal, palatal, velar, pharyngeal and glottal) and manner of articulation (plosive, fricative, non-lateral, affricate(d), nasal, lateral, trilling and gliding).

Additionally, aspiration inherent to Danish syllable-initial /p, t, k/ is often described as a supplementary contrastive feature of consonants, however, Grønnum (2005) suggests an alternative way of defining aspiration as an articulation feature subject to neutralization in a syllable final position. I shall use the term binary opposition regarding the aspiration since it has the representation of two members and these members are phonemically distinctive (compare minimal pairs *bind* [bin] - *pind* [b<sup>h</sup>in], *kø* [g<sup>h</sup>ø<sup>?</sup>] - *gø* [g̊ø<sup>?</sup>], *betaget* [b<code>e</code> d͡sasð<code>y</code>] -

Place/ manner of articulation	bilabial	labial- dental	dental	alveo- dental	alveolar	alveo- palatal	palatal	velar	uvular /phary ngeal	glottal
plosive	/p-/ (bʰ) /b-/ (b) /p/ /pʲ/ /b/ /bʲ/	/f/ (f) /f/ /f <sup>i</sup> /	/t/ /d/		/ <mark>d/ (d)</mark> /d <sup>i</sup> / /t <sup>i</sup> /			/k-/ (ồgh) /g/ (ồg) /k/ /k/ /g/ /gi <sup>j</sup>		
fricative		/v/		/s/ /s <sup>j</sup> / /z/ /z <sup>j</sup> /	/s/ (s)	/c:/ /ş/ /z/ /z:/		/x/ /x <sup>j</sup> /		h (h)
non-lateral							/ɕ/		(Ĕ) (Ř) ou \L <sub>A</sub> \	
approxi- mant		/v <sup>v</sup> / (v) or (w/u̯)			/ð <sup>v</sup> / (₫ <sup>ɣ</sup> )			/j <sup>v</sup> / (j/i̯)		

affricate(d)			/t-/				
		/ts/	(ds)	/tc/			
nasal	/m/ (m)	/n/	/n/ (n)			/ŋ/	
	$/m//m^{j}/$	/n <sup>j</sup> /				(ŋ)	
lateral		/}/	/l/ (l)				
			/l <sup>j</sup> /				
trilling				/ <b>r</b> / / <b>r</b> <sup>j</sup> /			
gliding					/j/		

 Table 5. Russian (in blue) and Danish (in red + with main allophones in parentheses)

 consonant phonemes

*bedaget* [beˈdasðv]) not as in English, for example, where the aspiration is not a phonemic quality. What is in fact more important for the current study is the absence of such a phonemically contrastive feature as aspiration in Russian. The latter leads to the assumption that this distinctive phonemic feature may turn out to be a potential ground for a foreign accent with the disaspiration of /p/, /t/, /k/, as well as confusion of the aspirated and non-aspirated consonant qualities, i.e. the assignment of aspiration where it should *not* be, and vice versa.

The major distinctive phonemic contrast (in supplement to place and manner of articulation) of the Russian consonant phonemes (and their allophones) is a binary opposition palatalized vs. non-palatalized. Compared to the Danish opposition aspirated vs. nonaspirated, it is more representative in terms of the number of phonemes that belong either to palatalized or hard consonants (only  $\hat{f_c}$ ,  $\hat{c_i}$ ,  $\hat{j}$  and  $\hat{z_i}$  are always palatalized, while  $\hat{f_s}$ ,  $\hat{s}$ , and /z/ are always hard). The palatalization is characterized by the movement of the central part of the body of the tongue towards the hard palate, which gives a soft sounding of the consonant. On the level of articulatory habits, the Russian palatalization can bring additional Russian-like pronunciation in Danish. The error analysis will show the exact influence of the distinction hard-vs.-soft on the Danish pronunciation. Russian learners are usually aware of the absence of palatalized consonant in Danish, however, as my teaching experience showed this sometimes leads to an opposite effect, and learners tend to pronounce Danish consonants similarly to Russian hard consonants, which is wrong. The latter often results in an unnecessary velarization, probably also because Russian is often characterized (Avanesov, 1956) as a velarized language. My hypothesis is that the transfer of the Russian opposition palatalized/non-palatalized into Danish may result in an excessive velarization of the Danish /l/, as well as palatalization of /b/, /d/, /g/ especially before Danish front /i/, /ɛ/, /y/, /e/. This hypothesis will be in the focus of the error analysis.

Another major distinction is the dental articulation of a series of allophones of the Russian hard phonemes /t/, /d/, /s/, /n/, /z/, /t͡s/. The dental articulation of English alveolar sounds is traditionally typical of the Russian native speakers of English. The same would probably be true of the Danish allophones of the similar fricative phonemes /d/, /s/, /t/, /n/ alveolar by origin. What is remarkable is that the dentals' paired soft phonemes /t<sup>j</sup>/, /d<sup>j</sup>/, /s<sup>j</sup>/, /n<sup>j</sup>/, /z<sup>j</sup>/ in Russian are alveolar. The latter may result in the transfer of dentalization into Danish, but this will be testen in the error analysis.

As for the Danish uvular /r/ and /ŋ/, new to the Russian natives, /r/ will probably be exposed to the slightest assimilation with or replacement by hardly similar Russian thrilling phonemes /r/ and /r<sup>j</sup>/, while /ŋ/ will have a major advantage, according to SLM, since no similar sound is found in Russian. However, the allophone [g] of the Danish /r/, a nonsyllabic vowel segment by nature, will probably sound more consonant-like, because 1) there is no vocalization as a regular phonological process in Russian; and 2) the effect of spelling interference mentioned in section 2.2.4. would probably trigger [ $\psi$ ], and not [g].

Danish  $\langle \delta \rangle$  has always been a partcular challenge for Russian learners at the beginners' level. However, in most cases  $\langle \delta \rangle$  is *a* most vivid demonstration of Flege's SLM. The intermediate and advanced Russian learners of Danish usually pronounce it native-like. However, the pronunciation of the lateral [1] insetad of  $\langle \delta \rangle$  may take place. The latter happens, because learners do not put forward the front part of the body of the tongue with the tip of the tongue touching lower teeth, but lift it towards the hard palate instead, with the tip of the tongue touching the aleveolar ridge and the lateral wings of the tongue slightly down. This makes  $\langle \delta \rangle$  sound like a lateral fricative rather than a non-lateral alveolar.

One of the most contrastive Danish consonant phonemes is the aspirated voicelss affricated alveolar /t/. I assume that since in terms of articulation it is closest to the Russian  $/\hat{ts}/$ , this would lead to a typical  $/\hat{ts}/$ -overtone in the sounding of the Danish /t/ according to the principle of equivalence classification discussed in section 2.2.3., since the Danish aspirated affricate /t/ (with as the main allophone [ $\hat{qs}$ ]) resembles most the Russian  $/\hat{ts}/$ .

Compared to Danish, Russian is enormously rich in voiced consonants and this will definitely put a voiced trace on the Danish articulation of Russian speakers in case of non-aspirated Danish /b, d, g/. Danish [s] in the intervocalic position may be prone to be voiced to [z], which is not found as an allophone of /s/ in Danish.

Thus, proceeding form the above given contrastive overview, I hypothesize that the Russian foreign accent in the articulation of Danish consonants may manifest itself in the following features:

- disaspiration of /p/, /t/, /k/;
- velarization of /l/;
- voicing of non-aspirated consonants /b/, /d/, /g/, also of the intervocalic [s].
- palatalization of /b/, /d/, /g/ especially before Danish front /i/, / $\emptyset$ /, / $\epsilon$ /, /y/ and /e/.
- dentalization of /d/, /s/, /t/ and /n/;
- /r/-assimilation to a thrilling /r/; "consonantization" of [y];
- [ts]-overtone in the Danish /t/.

### 3.4. Word stress in Danish and Russian

The previous sections of my contrastive overview were dedicated to the comparison of the two phonological systems on the segmental level. As was mentioned earlier, this study also addresses the issue of the Danish word stress assignment by late Russian learners. It will be a target of our second word list, along with the diphthongs.

My decision to study Russian foreign accent on the level of word stress was not accidental. On the one hand, "the acquisition of word stress assignment by late L2-learners has received limited attention in the L2 literature" (Archibald 1998: 177), and as for the language combination Russian (L1) - Danish (L2) no previous research has been done at all. However, this level of the second language phonology is promising in terms of the foreign accent improvement, since adults are capable of resetting their L1 metric parameters to the L2 setting (Archibald 1998: 177). Thus, the current accent study of the word stress assignment would have a practical significance for Russian learners of Danish. On the other hand, methodologically, the word stress assignment logically seems to be a next step in the current Russian accent study, because segments are organized hierarchically into syllables (Hall, 2006) and on the suprasegmnetal level the word stress or its absence function as major syllable properties (Hall, 2006). Since the scope of the current thesis does not admit a more extended research on the word stress assignment under various phonological conditions, I shall focus on the issue of word stress assignment when "the word is chosen as a focus word" (Gilbert 2008: 15). Moreover, this part of my thesis should lay the foundation for further research on the Russian foreign accent features in Danish on the level of sentence or prosodic stress assignment.

The word stress as a suprasegmental characteristic of the word phonological image, both in Danish and Russian has a dynamic nature. It means that a stressed syllable is produced, according to Avanesov (1956: 64), with a more tensed articulation of the syllable segments, especially vowels. In Russian, "more tensed" does not mean, however, that the stressed vowel is always long, and thus cannot be opposed to short vowels. It is rather a bit longer, compared to the unstressed vowels in the word.

Russian stress patterns can be hardly described from the point of view of regularities, since word stress in Russian cannot be attributed to a specific type of syllables or a type of a vowel (Avanesov, 1956). To know where the stress should be set, it is necessary to know the word (Avanesov, 1956). The word stress in Russian is not fixed and does not depend of the order number of a syllable in a word '*me.bel.ny* (furniture-related), '*kraj.ny* (extreme, adj.) – the stress is on the first syllable; *pri.be*'.*gat* (to run to a place), *pe.re.vo.*'*dit* (to translate) – the stress is on the last syllable; u.sta.'nov.ka (an installation) – on the third syllable; *ras.*'*smat.ri.vat* (to consider) – on the second syllable. Moreover, a word stress variation is a typical characteristic of Russian. Word stress can be movable depending on a word's grammatical form *smely* (brave) – *sme* '*leye* (more brave); *no* '*ga* (a leg) – '*nogi* (legs), as well as derivational processes, historical change, professional or dialectal use (Lagerberg, 2007).

In the field of Russian accentology, word stress patterns are usually studied (Ukiah, 2002; Sharapova, 2000; Lagerberg, 2005; Lagerberg, 2006) in relation to a particular class of morphologically similar classes within part of speech, since no "general" rules applicable to all the parts of speech can be formulated.

As for the word stress assignment in Danish, compared to that one in Russian, it can be more or less predicted, and general rules concerning the assignment of word stress can be formulated. According to Grønnum (2005: 245), general word stress patterns can be systematized according to the following rules, with exceptions, of course:

- Word stress can only fall on the syllable with the full vowel.
- If there are more than one full vowel in the word, and one of them is a long one, then the syllable with the long vowel will take the stress.
- If there are more than one long vowel in the word, the first syllable with the long vowel will take the stress.
- If there is no long vowels in the word, the last syllable with a short full vowel followed by a consonant will take the stress.
- Borrowings from French have stress on the last syllable.
- If ending in a nasal consonant, a word would have the word stress on the last but one syllable.
- Foreign words ending in /r/ never have the word stress on the last syllable (with a few exceptions).

• Prefixes, such as *be-*, *ge-*, *er-*, *for-*, are usually unstressed. Some of them, such as *u-*, *und-*, *mis-*, *van-* can be either stressed or not.

Above, I spoke about the primary stress. However, in both Russian and Danish a secondary stress may be distinguished. In Russian, the secondary stress is inherent to compounds and non-compound long words. It is important to note, that in a few-stem compounds the secondary stress is usually assigned to the first stem, and the main stress – to the second or following stems, e.g. *dalnevos'tochny* (Far-Eastern), *samo\_letostro'yeniye* (aircraft construction) (Avanesov, 1956).

In Danish the secondary stress in typical of some suffixes in non-compound words, for example, *-dom, -hed, -skab, -som, -vis, -bar, mæssig, -dømme* (Grønnum, 2005). However, the secondary stress generally is more typical of compounds (komposita) in Danish. The general rule about the assignment of the secondary stress in compound words, according to Heger (1992: 125) is that the first stem of the compound carries the main stress on *that* syllable, which carries the main stress in an isolated pronunciation of the stem. The secondary stress is assigned to the last stem of the compound on *that* syllable, which carries the main stress in an isolated pronunciation of this stem, as for instance, *engelsk lærer, 'morgen menneske* (Heger 1992: 125). However, there are numerous exceptions to this rule e.g. *med lidende, syd vest*, etc. (Fischer-Jørgensen 2001: 4-6). When one compound forms the final part of a new longer compound, the same rule is applied, e.g. *'voldgrav* – *'slots, vold grav* (Heger 1992: 125).

A third type of the word stress called a *weak stress* ("svagtryk" – usually marked as <sub>o</sub>) (Heger 1992), can be often distinguished in Danish. The latter is typical of cases, when a compound forms an initial part of a new longer compound, normally the last stem of the original (initial) compound would carry the weak stress, while the last component of the new compound would have the secondary stress, e.g. *'tryk luft bor* (Heger 1992: 125).

We can see that secondary stress principles are different in Danish. It is possible to assume that Russian native speakers would probably tend to set the secondary stress closer to the beginning of the Danish word instead of the main stress. Moreover, the distinction between stressed or non-stressed Danish prefixes and suffixes will be probably the most serious challenge, and thus a potential ground for a foreign accent in the word stress assignment. Another possible scenario is the ignorance of the secondary stress.

Another potential difficulty will be hidden in the Danish words carrying a double primary stress, such as compound numerals; words, consisting of a preposition and an adverb (*'bag'efter, 'der'for*)<sup>11</sup>, and some random words (*'abso'lut, 'di'rekte, 'alle'rede*)<sup>12</sup>, (Grønnum, 2005) or adjectives and verbs with the first stem which indicates a very high degree of the second one (Heger, 1992; Fischer-Jørgensen, 2001), e.g. *'is 'kold, 'lyn 'hurtig*, etc. The double main stress is absolutely non-typical of Russian word stress principles, and thus will be probably ignored or produced as if there were a secondary and main stress in the word. All these potential "stumble blocks" will be considered while preparing reading materials for the reading task.

## 4. Collecting empirical data

In section 2, I gave a short outline of the case study and its methodological principles. Section 4 will present the linguistic experiment carried out as a part of the current project in a more detailed way.

## 4.1. Preparing reading materials

#### 4.1.1. Word list 1

The first world list (further referred to as WL1) consists of words with vowel and consonant segments as target sounds for the error analysis (see **Appendix 2**). Words with short vowels as target sounds were chosen in such a way that those Danish vowel segments which hypothetically could be subject to the qualitative reduction in unstressed positions (according to degrees of vowel reduction in Russian analyzed in section 3.1.2.), would occur in the pre-stressed and post-tonic positions. That allowed me testing the hypothesis about the excessive reduction of unstressed Danish vowel qualities by Russian native speakers influenced by the reduction patterns of their mother tongue.

As far as the consonant targets are concerned, the words containing them were chosen according to the principle of having a target consonant sound in the most illustrative positions. The target words for the target consonant sounds, which were predicted in the theoretical comparative study to be pronounced non-native like, were selected in such a way that they would cover the target sounds' syllable-initial and syllable-final positions, as well as in the middle of a word.

<sup>&</sup>lt;sup>11</sup> However, some native speakers would pronounce such words with one main stress and one secondary stress.

<sup>&</sup>lt;sup>12</sup> See footnote 11.

It should be noted, that in my project I share Grønnum's (2001: 258) point of view in classifying Danish [ $\check{0}$ <sup>x</sup>] as an approximant rather than a semivowel. Therefore, [ $\check{0}$ <sup>x</sup>] was included in WL1. I shall further use the term *semivowel* in connection with the following non-syllabic qualities: [j/i], [w/u] and [g] as allophones of /j/, /v/, and /r/, respectively. Sounds [w] and [j/i] however are also relevant as allophones of /g/. Since I treat [j/i], as well as [w/u] and [g] as allophones of consonant phonemes I have included these semivowel qualities in WL1 as allophones of consonant phonemes and shall discuss the pronunciation of these qualities in the section about the results of the error analysis for consonants. These semivowels shall be also in the focus of our analysis as diphthong components in section 5.3.1.4.

Overall, the WL1 consists of 130 words with target vowels and consonants. The words were shuffled, so that the subjects could not find any regularity in the segments occurrence. (see Appendix 2). Appendix 3 shows the target vowel segments, and Appendix 4 – the consonant segments.

### 4.1.2. Word list 2

The second word list (further referred to as WL2) (see **Appendix 5**) for the reading task consists of target words for the word stress assignment and words with diphthongs as target segments. The word stress targets were chosen in such a way that they include words with two primary stresses and with one primary and one secondary stress as well as words with one primary stress and more than one secondary stress. For the word stress targets, see **Appendix 6**. Some of the words stress targets were chosen according to the word stress patterns described by Grønnum (2005: 245). See section 3.4.

The target words for the word stress assignment were shuffled with the diphthong targets. See **Appendix 7** for the diphthong targets. The latter ones were chosen in such a way that both falling and rising diphthongs would be covered, in order to find out if there is a correlation between the structure of the diphthong and the pronunciation of the latter by the Russian native speakers. Overall, the WL2 consists of 51 shuffled items: 26 word stress targets and 25 diphthongs targets. *Evnesvag* was a two-target word with two target diphthongs.

#### 4.1.3. Text sample

While the WL1 and the WL2 were aimed at revealing accent features on the segmental level (pronunciation of consonants and vowels) and one aspect of the suprasegmental level – word stress, the third task – reading a small text (further referred to as **T**) aloud – provided data for the global accent assessment, in order to find out whether SIPT can decrease the degree of the global foreign accent in Danish learners with Russian as their mother tongue, or not. The T was taken from a student book *"Skolegade 4"* (Sandal, 2005: 79), see section 6. The level of this book corresponds to that of upper intermediate and advanced students. I shall further give a more detailed description of the recording and rating procedures for the T-task separately in section 6.

## 4.2. Procedure

#### 4.2.1. D-vs.-R taxonomy

The recording of the reading samples took place in Russia at *Moscow State Linguistic University* (further MSLU) and in Denmark in language schools *Lærdansk Aarhus* and *Lærdansk Sønderborg*, as well as *AOF Sprogcenter Aabenraa*. Overall, 12 subjects with Russian as L1 took part in the recording in Denmark and 15 in Russia. This fact stands for the D-vs.-R-taxonomy.

The idea behind this D-vs.-R taxonomy is that all the students from MSLU had a SIPT when they started learning Danish, which anticipated the main language practice course, while in the case of students of the mentioned language schools, the teaching of the Danish language pronunciation has been integrated in day-to-day teaching activities and students did not have a special four-month introductory course on the Danish pronunciation.

This division of the subjects into two groups was mainly needed to test the hypothesis about the role of SIPT as an accent-mitigating factor, by means of the global accent assessment. However, the D-vs.-R taxonomy was also applied in the error analysis to see whether the D-group and the R-group would have similar or different typical accent features and in what way SIPT could contribute to the foreign accent reduction as opposed to living in the country and being exposed to the Danish native input.

#### **4.2.2. Recording Procedure**

The recording procedure was identical for all the participants. All the recorded samples are available as a digital supplement to the thesis. See **Appendix 25.** Each participant was offered to read the three types of materials in the following order: WL1 - T - WL2. The reading materials were not in any way introduced to the participants before the recording. Only instructions about reading words at a natural pace were given. Thus, 27 samples of WL1, WL2 and T were recorded. It should be noted that in the case of the T-readings, the title was not read by all the participants, but this fact is considered to be a minor issue, since this would affect the ratings.

Since the recordings were anonymous, each participant received his or her personal participant number (PPN). Personal numbers were later used in all the file names and the error analysis tables. The following two types<sup>13</sup> of taxonomy were used as interchangeable principles in notations:

1) R + PPN - for example, R3 stands for the participant from the R-group with SIPT, whose PPN was 3.

D + PPN - for example, D6 stands for the participant from the D-group without SIPT, whose PPN was 6.

2) 1 + PPN stands for participants from the R-group, e.g. 1.1., 1.6.

2 + PPN stands for participants from the D-group. e.g. 2.1., 2.6.

Thus, for instance 1.4.WL1 stands for the reading sample of the WL1 read by a participant from the R-group whose PPN is 4, while 2.4.T stands for the reading sample of the text by a participant from the D-group whose PPN is 4.

After having read the WL1, T and WL2, the participants were offered to fill out a questionnaire consisting of 20 questions (see **Appendix 1**). The questionnaire from the theoretical point of view was wholly and totally based on the overview of the factors said to determinate accent degree according to numerous previous studies - see section 2.2. The

<sup>&</sup>lt;sup>13</sup> I had to preserve the usage of these two types of taxonomy, since letters R and D had been applied in the announcement of the participants during the recording procedure even before the error analysis, while number codes were used afterwards in the file systematization. Moreover, the letter "codes" applied are more convenient for distinguishing the two groups of subjects in the text of our project, while it was more convenient to work with numbers while doing error analysis.

information reported in the questionnaires would help mainly in the analysis of the global accent ratings. The questionnaire made it also possible to sketch general linguistic portraits of the participants.

## 4.3. Subjects

I recorded 27 subjects, 18 females and 9 males. According to the questionnaires, they were aged between 19-60 years. All the subjects speak the modern Russian literary language (see section 2.4.) as their L1. It is important to note that the distinctive features of the Russian phonemic inventories, the mentioned weakening of the vowels and the peculiarities of the word stress assignment are inherent to the pronunciation of all the Russian subjects, recorded in Denmark and in Russian.

Although the percentage of Russian usage everyday (from 10% - 90% of all the languages in use including English, Lithuanian, Latvian, Ukrainian, Belarusian and Chechen) varies from subject to subject due to different study, work or family conditions, they all speak Russian native-like. **Figure 4** shows a more detailed overview of the subjects' linguistic portraits, as based on the questionnaires. See **Appendix 1** for the questionnaire.

All the subjects with PPNs 2.1.-2.8, 2.10-2.13 (on the light green field in **Figure 4**) have studied or are studying Danish as a second language in Denmark. D5 (2.5), however, started learning Danish as a foreign language in Russian as an undergraduate student. Nevertheless, I referred D5 to the D-group, because D5 have lived in Denmark and studied as a full-degree student for approximately a year, of which D5 had one semester of graduate studies in Danish. D5 have for certain time followed a course on Danish as a second language in a Danish language school, and never had any SIPT as intensive as the one that the R-subjects received when they started earning Danish.

All the R-subjects with PPNs from 1.11-1.15 (on the light orange field in **Figure 4**) have studied Danish as a foreign language in Russia. R11 stayed in Denmark for three weeks for a summer language course, and R9 spent 10 days in Denmark as a tourist. All the R-subjects have had Russian native teachers with Danish as a foreign language compared to the D-subjects, who had Danish native teachers (only D5 had three Russian native speakers and one Danish native speaker when D5 started learning Danish in Russian).

11.1.19 M2410030101.2.21 M24100school only101.3.19 F24100school only101.4.19 F24100school only101.4.19 F24100school only101.5.19 F24100school only101.6.21 F24100school only101.6.21 F241009101.7.19 M181009101.8.21 F181004101.9.20 M181003141.11.21 F451003141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5141.14.22 F301515762.4.22 F301515762.5.22 F483551,5102.6.30 F3680every day (w)every day52.7.60 F3070every day (w)every day52.7.60 F3070every day (m) <td< th=""><th>DDN</th><th>ogoloov</th><th>Length/Instr</th><th>Danich 9/</th><th>speak</th><th>listen/n+MEDIA/week (h)</th><th>less/week/</th></td<>	DDN	ogoloov	Length/Instr	Danich 9/	speak	listen/n+MEDIA/week (h)	less/week/
11.17.17.17.1005050101.2.21 M24100school only101.3.19 F24100school only101.4.19 F24100school only101.5.19 F24100school only101.6.21 F241001101.6.21 F241001101.6.21 F241004101.7.19 M181004101.8.21 F181003141.1.22 F451003141.1.21 F451003141.1.2.21 F451003141.1.3.22 M451003141.1.4.22 F451003.5141.1.5.20 F421003.5141.1.4.22 F301515762.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day5		10 M	24	10			<b>a 45 mm</b>
1.2.21 M24100school only101.3.19 F24100school only101.4.19 F24100school only101.5.19 F241001101.6.21 F241001101.7.19 M181009101.8.21 F181004101.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5141.14.22 F301515762.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (w)every day52.825 M2610114.52.1	1.1.	19 M	24	10	0	school only	10
1.3.       19 F       24       10       0       school only       10         1.4.       19 F       24       10       0       school only       10         1.5.       19 F       24       10       0       school only       10         1.6.       21 F       24       10       0       school only       10         1.6.       21 F       24       10       0       school only       10         1.7.       19 M       18       10       0       9       10         1.8.       21 F       18       10       0       3       14         1.10.       22 F       45       10       0       3       14         1.11.       21 F       45       10       0       3       14         1.12.       21 F       45       10       0       3.5       14         1.13.       22 M       45       10       0       3.5       14         1.14.       22 F       42       10       0       3.5       14         1.14.       22 F       30       160 (w)       10       6       2.4         2.1.       31 M	1.2.	21 M 10 E	24	10	0	school only	10
1.4.19 F24100school only101.5.19 F24100school only101.6.21 F241001101.7.19 M181009101.8.21 F181004101.9.20 M181003141.10.22 F451003141.11.21 F451003141.12.21 F451003141.13.22 M451003.5141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.6.30 F3680every day (w)every day52.7.60 F3070every day (w)every day52.7.60 F3070every day (m)11152.8.25 M2610114.52.1.34 M24204.53.54	1.3.	19 F	24	10	0	school only	10
1.3.1924100school only101.6.21 F241001101.7.19 M181009101.8.21 F181004101.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003.5141.14.22 F451003.5141.15.20 F421003.5141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M204.53.542.1130 F24<	1.4.	19 F	24	10	0	school only	10
1.0.2.1 F2.4100110101.7.19 M181009101.8.21 F181004101.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.6.30 F3680every day (w)every day52.7.60 F3070every day (h)11152.8.25 M2610114.52.1.24 M204.53.542.1.130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.5.	191 <sup>-</sup> 21 E	24	10	0	1	10
1.7.19 M18100910101.8.21 F181005101.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.0.	21 F 10 M	19	10	0	0	10
1.a. $21  \text{F}$ 181003101.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1./.	19 M 21 E	10	10	0	5	10
1.9.20 M181004101.10.22 F451003141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M204.53.542.1130 F241020school only202.1230 F1820school only20212.1333 F1410303513	1.0.	21 F	10	10	0	3	10
1.10.22 F451005141.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.5.22 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.1130 F24204.53.542.1130 F1820school only202.1230 F1820school only20	1.9.	20 M	18	10	0	4	10
1.11.21 F45205.56141.12.21 F451003141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.1130 F241020school only202.1230 F1820school only202.1333 F1410303513	1.10.	22 F	45	10	0	3	14
1.12.21 F451005141.13.22 M451003141.14.22 F451003.5141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.11.	21 F	45	20	5.5	6	14
1.13.22 M451005141.14.22 F451003141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.12.	21 F	45	10	0	3	14
1.14.22 F451003141.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.13.	22 M	45	10	0	3	14
1.15.20 F421003.5142.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.14.	22 F	45	10	0	3	14
2.1.31 M106025 (w)1042.2.31 M1753242.3.27 F2730160 (w)1062.4.22 F301515762.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	1.15.	20 F	42	10	0	3.5	14
2.2.       31 M       17       5       3       2       4         2.3.       27 F       27       30       160 (w)       10       6         2.4.       22 F       30       15       15       7       6         2.5       22 F       48       35       5       1,5       10         2.6       30 F       36       80       every day (w)       every day       5         2.7       60 F       30       70       every day (h)       11       15         2.8       25 M       26       10       1       1       4.5         2.1       24 M       24       20       4.5       3.5       4         2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.1.	31 M	10	60	25 (w)	10	4
2.3.       27 F       27       30       160 (w)       10       6         2.4.       22 F       30       15       15       7       6         2.5       22 F       48       35       5       1,5       10         2.6       30 F       36       80       every day (w)       every day       5         2.7       60 F       30       70       every day (h)       11       15         2.8       25 M       26       10       1       1       4.5         2.1       24 M       24       20       4.5       3.5       4         2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.2.	31 M	17	5	3	2	4
2.4.       22 F       30       15       15       7       6         2.5       22 F       48       35       5       1,5       10         2.6       30 F       36       80       every day (w)       every day       5         2.7       60 F       30       70       every day (h)       11       15         2.8       25 M       26       10       1       1       4.5         2.1       24 M       24       20       4.5       3.5       4         2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.3.	27 F	27	30	160 (w)	10	6
2.522 F483551,5102.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	2.4.	22 F	30	15	15	7	6
2.630 F3680every day (w)every day52.760 F3070every day (h)11152.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	2.5	22 F	48	35	5	1,5	10
2.7       60 F       30       70       every day (h)       11       15         2.8       25 M       26       10       1       1       4.5         2.1       24 M       24       20       4.5       3.5       4         2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.6	30 F	36	80	every day (w)	every day	5
2.825 M2610114.52.124 M24204.53.542.1130 F241020school only202.1230 F1820school onlyvery little outside school202.1333 F1410303513	2.7	60 F	30	70	every day (h)	11	15
2.1       24 M       24       20       4.5       3.5       4         2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.8	25 M	26	10	1	1	4.5
2.11       30 F       24       10       20       school only       20         2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.1	24 M	24	20	4.5	3.5	4
2.12       30 F       18       20       school only       very little outside school       20         2.13       33 F       14       10       30       35       13	2.11	30 F	24	10	20	school only	20
2.13 33 F 14 10 30 35 13	2.12	30 F	18	20	school only	very little outside school	20
	2.13	33 F	14	10	30	35	13

## Figure 4. Linguistic portraits of the subjects according questionnaires

### **Abbreviations:**

- *Length/Instr (mos)* total number of months of instruction;
- speak to n/mo(h) number of hours per months used to speaking to native speakers;
- *listen/n+MEDIA/week (h) outside school* number of hours per week used to listening activities (mass media, music, films, native speakers);
- *less/week/ à 45 min* number of lessons per week à 45 min;
- (w) at work.
- (h) at home
- Danish % percentage of Danish use on an everyday basis out of 100% (all languages spoken).

For the error analysis, I chose 36 samples – 18 WL1 and 18 WL2 samples read by correspondingly 9 subjects from the D-group and 9 from the R-group. The chosen samples were read by the same subjects, except for one WL2 sample when D13 was changed to D10 because of D13's very heavy accent in the WL2 reading, which would otherwise make the reading sound unnatural and with breaks preventing from doing any analysis on the word stress assignment. In general, both in the case of the WL1 and WL2, the samples were chosen randomly.

As far as the T-readings are concerned, all 27 samples were taken for accent degree and accent comprehensibility ratings and shuffled in a random order supplemented with 4 T-samples of Danish native controls. See section 6 for more details on the T-task recording procedure.

## 5. Data analysis

## 5.1. Transcribing speech samples & error systematization

Overall, 18 samples or 3258 words were transcribed: 18 WL1s and 18 WL2s. Tsamples were not transcribed, but shuffled and sent out directly to the raters. As was mentioned above, the IPA narrow phonetic transcription was applied for all 36 samples. The mispronounced sounds and word stress assignment different from the pronunciation recommended in *Den Store Danske Ordbog* were highlighted by hand in red. All the transcriptionsheets are presented in **Appendices 19, 20, 21** and **22**.

Since one of the two main goals of the current master thesis was to find out what the most typical features of the Russian accent are, I have decided to do the data systematization by focusing on the mispronounced elements. Firstly, all the final transcriptions were gathered in two spreadsheets, one for the D-samples, and the other one for the R-samples. Afterwards, each sample was copied into a separate column with PPNs as headlines of the column. Such a way of organizing all the transcribed samples allowed tracing all the varieties of the target elements across all the R- and D-subjects. Moreover, having two separate sheets for each group could give space for figuring out whether the typical features would differ in the two groups.

The error systematization was realized by means of counting the number of the correct readings of the target sounds and of all the variants which differed from the recommended. The latter ones were noted as \*. The error systematization was done separately for the vowel, consonant, diphthong and word stress targets. Below I shall discuss the results of the error

systematization and shall do an error analysis in order to find out what the typical features of the Russian accent are.

## 5.3. Error analysis

According to Flege's Speech Learning Model (SLM) (theoretical model of the second language learning introduced in section 2.2.) L2 learners tend to classify an L2 sound which is acoustically similar to an L1 sound into a pre-existing phonic category (Birdsong 2007: 100), i.e. process the L2 sounds through their well-established L1 phonological system, and as a result in the long run more similar sounds lack in native-likeness compared to more dissimilar sounds.

Another theoretical approach mentioned above in section 2.2. - Similarity Differential Rate Hypothesis (SDRH) – analogically to SLM addresses an accent issue from a dichotomy point of view (dissimilarity vs. similarity), but argues in favor of speaking about different rate of acquisition for similar and dissimilar sound rather than just for a better or worse performance at different stages of a L-acquisition. In our current research, I argue in favor of the former model (see section 2.2.), and below shall try to substantiate it with particular examples.

## 5.3.1. Vowel features

Proceeding from the principle of "equivalence classification" I earlier formulated the hypotheses about accent features in the case of Russian accent by means of a contrastive method, having comparing two phonological systems, and predicted the following eventual realizations of the "equivalence classification":

## For vowel segments:

- excessive and unnecessary qualitative reduction of unstressed vowels;
- shortening of the long Danish vowels and lengthening of the short ones;
- fewer quality distinctive properties of the back vowels and front /i/ and /e/. Let us now do the error analysis for the vowel targets.

## 5.3.1.1. Front and central full vowels

5	3	1	1	1	Ou	ali	tativ	ve	feat	tures
$\mathcal{I}$	•••	• •		т.	٧u	un	luli	vv	rou	iui ob

Target words	t/s <sup>14</sup>	÷	[a]*	[e]*	[ɛ]*	[ə]*	[ạ:]*
p <b>a</b> nde		6	1				2
p <b>a</b> nere	ą	4	3		2		
sof <b>a</b>	_	3			4	2	
kærlighed		9					
erkende	3	4		4		1	
kæresterere		3		6			

Table 6. Pronunciation of the target [a], [ɛ] by the R-group

Target words	t/s	+	[a]*	[e]*	[v]*	[9]*
p <b>a</b> nde		8	1			
p <b>a</b> nere	ą	2	1		3	3
sofa		1	1		6	1
kærlighed		8		1		
erkende	3	7				2
kæresterere		8		1		

Table 7. Pronunciation of the target [a], [ε] by the D-group

As is seen from the error systematization, Danish front vowels [a] and [ $\epsilon$ ] were most often pronounced correctly by both the D- and R-subjects in stressed positions (see **Table 6** for the R-group and **Table 7** for the D-group; further in the text all the error systematization tables for the R-group would come before those for the D-group).

As for the unstressed positions, the D-subjects produced more correct variants of the target unstressed [ $\epsilon$ ] in *erkende* and *kæresterere*. Even though the word stress in *erkende* in 5 out of 4 cases was wrong, only 2 [ $\vartheta$ ]\*s were produced instead of [ $\epsilon$ ]; while in the R-group 4 out of 5 incorrect variants were [e]\* instead of [ $\epsilon$ ]. Here it is possible to assume, that it is more typical of Russian to have [e] rather than [ $\epsilon$ ] at the beginning of a word. Words with an initial [ $\epsilon$ ] are not numerous in Russian, and often are of a foreign origin. The reduction of [ $\epsilon$ ] to [ $\vartheta$ ], even with a wrong words stress as well as an illustrative example of the post-tonic [ $\mathfrak{q}$ ] reduction to either [ $\mathfrak{v}$ ]\* or [ $\vartheta$ ]\* in *sofa*, especially in the D-group with only 1 correct pronunciation, lay ground for verifying the hypothesis about the post-tonic and pre-tonic vowel reduction of the Danish phonemes /a/ and / $\epsilon$ / to a more closed allophone according to the Russian weakening of vowels.

In my predictions about the front vowel reduction, I pointed out the phonemes /a/ and  $\epsilon$ / as major eventual targets for weakening in unstressed positions. The data analysis showed however, that this could have been extended over other vowel qualities. Probably, the most

<sup>&</sup>lt;sup>14</sup> t/s stands for the target segment

convincing examples of a qualitative reduction of front and central vowels in unstressed syllables according to Russian vowel reduction patterns are those of [e] in *beslægtet*, *forbillede* made by both the R- and D-subjects. See **Tables 8, 9.** 

Target words	t/s	<b>.</b>	[i]*	[i:]*	[I]*	[i <sup>.</sup> ]*
binde		2	6	1		
beslægtet	ę	2			7	
forbillede	_	2	5		1	1

Table 8. Pronunciation of the target [e] by the R-group

Target words	t/s	+	[i]*	[I]*	*[3]
b <b>i</b> nde		2	7		
beslægtet	ę	0		6	3
forb <b>i</b> llede		5	2	2	

Table 9. Pronunciation of the target [e] by the D-group

Similarly to the above-mentioned [a] and [ $\epsilon$ ], [e] was reduced both in post-tonic and pre-tonic positions: altogether 13 out of 18 reductions to [1]\* in the first pre-stressed syllable of *beslægtet*, however, only 3 reductions to [1]\* in the post-tonic syllable of *forbillede*. These cases of the reduction of [e] were probably even in favor of the subjects, [e] is even more similar to [1] than the Russian [i] is. However, the reductions to [1]\* were mainly due to a wrong primary word stress assigned by 13 participants to the second syllable of *forbillede*, thus [e] was stressed, and since stressed, due to the spelling influence it was produced by 7 subjects as [i]\*. In contrast to /a/ and / $\epsilon$ /, /i/ in *vikar* and *livlig* was produced correctly by the majority of the subjects (15 and 16 correct variants respectively for each word).

Not only front vowel segments were reduced in quality. The central [ä] in the pre-tonic position in *utaknemmelig* and <u>the second pre-tonic syllable</u> in *apparat* was reduced to [v] by 5 out of 9 R-subjects. See **Tables 10, 11.** 

Target words	t/s	+	[b]*	[ã]*
mangle		8		1
ut <b>a</b> knemmelig	ä	4	5	
<b>a</b> pparat		4	5	

Table 10. Pronunciation of the target [ä] by the R-group

Target words	t/s	÷	[a]*	[ß]*	[ə]*	[ạ]*	[ã̃]*
mangle		5	3			2	1
ut <b>a</b> knemmelig	ä	5	2	2			
<b>a</b> pparat		5	2	1	1		

Table 11. Pronunciation of the target [ä] by the D-group

The D-group showed a less frequent reduction of [ä] in *utaknemmelig* and *apparat*. This was very likely due to less natural, slower and constraint readings by the D-subjects, especially those of *utaknemmelig*. Such readings lead to the less natural and less fluent productions, where a qualitative reduction is less probable.

What was positive about these slightly slower readings of *utaknemmelig* and *apparat* by the D-group is that one could clearly trace a more retracted Russian [a], instead of the Danish central [ä]. It is assumedly right because of the above-mentioned *equivalence classification* in accordance with Flege's SLM.

Target words	t/s	+	[i]*	[i:]*	[I]*	[i <sup>-</sup> ]*	[e]*
binde		2	6	1			
beslægtet	ę	2			7		
forb <b>i</b> llede	_	2	5		1	1	
vilde		3					6
vikar	i	8			1		
l <b>i</b> vlig		9					

Table 12. Targets [e] and [i] by the R-group

Target words	t/s	+	[i]*	[I]*	*[3]	[e]*
binde		2	7			
beslægtet	ę	0		6	3	
forb <b>i</b> llede		5	2	2		
vilde		3				6
v <b>i</b> kar	i	7		2		
l <b>i</b> vlig		9				

Table 13. Targets [e] and [i] by the D-group

Other vivid evidence in support of Flege's SLM is the production of the Danish [e] and [i]. See **Tables 12, 13**. In the stressed position only 4 out 18 possible correct pronunciations of [e] were given in *binde*, while 13 subjects pronounced [i]\* instead. This could be easily heard as the sound [i]\* was purely Russian, and this could be substantiated by the fact that some participants gave a palatalized  $[b^{j}]^{*}$  which would be natural before a front vowel [i] in Russian. Such poor results for the sound [e] can be explained from the point of view of the phonological inventory of the Russian language. As was mentioned in section 3.1., Russian makes only distinction between vowel qualities [i], [i] in stressed positions and [i], [i] and [1] in unstressed ones. No sound equal to [e] can be found in Russian.

However, it would be a mistake to suppose, that the absence of [e] in Russian would according to SLM, on the contrary lead to a better performance for this sound by Russian natives. The absence of [e] gives an opposite effect due to a high degree of the similarity

between [i] and [e], since the distinction between [i] and [e] is very narrow in terms of height, the Russian learners of Danish tend to automatically fail to preserve it, and classify [e] as [i]\*.

Nonetheless, the latter does not automatically mean that Russian natives are not aware of this distinction. They *are* (no teacher of Danish would ignore explaining the difference between the two sounds), and this awareness accounts for the idea that the [i]-vs.-[e]- confusion is a two-sided phenomenon, and has an impact on the performance of both sounds. In other words, the Russian natives, on the one hand classify [e] as [i]\*, especially, when letter *i* stands for the former sound, and on the other hand, being aware that there *is* [i]-vs.-[e]-distinction in Danish may confuse [i] with [e]\*. The latter may lead to semantic mistakes in the case of minimal pairs, such as *ski* – *ske*.

The fact that the Russian [i] is similar to Danish [i] does not directly lead to the conclusion that [i] would always "get higher scores" compared to [e], and I have an empirical evidence for that: in *vilde* 12 out of 18 participants produced [e]\* instead of [i]. This happened assumedly, because Russian learners, being aware of the [i]-vs.-[e]-distinction tend to open [i] to [e] in a closed syllable, what is not an absolute rule in all cases, and secondly due to a high similarity between [e] and [i].

Thus, with the example of a lacking [i]-vs.-[e] distinction in Russian accent, it was shown that the equivalence classification may be regarded as a two-sided phenomenon. Therefore, it is possible to empirically extend Flege's idea by saying that not only *do* advanced the L2 learners demonstrate a worse performance for the L2 sound (A) more similar to the corresponding L1 sound (B), they may also have a tendency to produce A instead of B. Only large exposure to the native input can make Russian learners of Danish be able to produce these two distinctive sounds correctly, and this may happen even long after a learner have gained an advanced command of Danish, as the current experiment showed.

Whereas in the above mentioned short sounds [a], [ $\epsilon$ ], [a], [e] and [i] the realization of the equivalence classification effect was evident in stressed positions, other accent features characterize the pronunciation of the target stressed [e]. See **Tables 14, 15**.

Target words	t/s	Ŧ	[I]*	*[3]
pædagog pædagogisk	е	8 8	1 1	
hemmeligt		6		3

 Table 14. Target [e] by the R-group

Target words	t/s	<b>4</b>	[I]*	[-]*	*[3]	[i]*
pædagog pædagogisk	e	7 9	1	1		
hemmeligt		3			4	2
Table 15 Ta	maget [	al hr fl		MOIN		

 Table 15. Target [e] by the D-group

In the stressed position in *hemmeligt* it was produced by 7 subjects as  $[\epsilon]^*$  and 2 as  $[i]^*$  (out of 18). The former sound was produced assumedly due to the initial [h]-influence. In Russian [e]-production after [h] would always imply palatalization of [h]. Thus, a strive to preserve the Danish hard [h] resulted in 7 cases in a more open front vowel quality, more typical of the Russian "C+[ $\epsilon$ ]", where C is a hard consonant. What is interesting, is that in the unstressed positions in *pædagog* and *pædagogisk* only a couple of the target [e]-sounds were produced with a change to an unstressed [1]\*, probably due to the facts that if produced without reduction at a slow rate, corresponding Russian words *pedagog* (a teacher) and *pedagogichesky* (pedagogical) would have [e] in both words after the initial [p<sup>i</sup>].

A very vague for Russian natives, Danish front-mid vowel quality [a] (more front than the Russian /a/) which usually occurs in a post-/r/ position as in the target word *dreng* was incorrectly produced as [e]\* and [ $\epsilon$ ]\* (4 and 3 times respectively, out of 18). If we consider, that [ $\epsilon$ ]\*-variants would hardly cause any comprehensibility problems among native speakers, then the performance for the target [a] in *dreng* was better in the D-group. This was probably due to a larger exposure to the native input compared to the R-subject, whose mispronunciation in this case was assumedly the result of the *spelling interference* (Miglio & Fukazawa 2006; Ehri & Wilce 1980), mentioned in section 2.2.4, and lack of the native input.

Similarly to Miglio & Fukazawa's theoretical approach, after having done the error analysis, I can argue that in some cases the Russian natives' pronunciation in Danish is not the result of a direct interference from the Russian language or equivalence classification, but that it is rather a complex "phenomenon whereby the spelling of the word..." in the L2 "...triggers a correspondence between..." an L2 spelling symbol "...and the pronunciation of the same symbol in the native language..." (Miglio & Fukazawa 2006: 4145).

I argue that in Danish words where vowel allophones are represented in spelling by means of the letters also found in the Russian language (that *may* be or *are* used in the latter to represent a different vowel quality) a Russian-like pronunciation will have a greater probability. That accounts for the [e]\* and  $[\varepsilon]$ \* realizations in *dreng*, since in the Russian natives' letter-to-sound system, the letter *e* would automatically stand for either sound [e]

after  $C^{j}$  or [ $\epsilon$ ] after a C, whereas the /r/-influence on a vowel quality typical of Danish remains to be neglected, as the latter phenomenon is absolutely alien to Russian.

Flege's SLM focuses both on the similar and dissimilar sounds. Above, I discussed the front and central short Danish vowel segments which have roughly speaking at least *some* similarity with major Russian vowels qualities. Now let us discuss the results for the dissimilar front vowel segments, namely [y],  $[\phi]$ ,  $[\phi]$ ,  $[\phi]$ ,  $[\phi]$ , and all the long vowels [ $\varepsilon$ :],  $[\ddot{\alpha}$ :],  $[\phi:]$ ,  $[\phi:]$ ,  $[\psi:]$  and  $[\phi:]$ . As far as the long vowels [ $\varepsilon$ :],  $[\ddot{\alpha}$ :],  $[\phi:]$ ,  $[\phi:]$ ,  $[\psi:]$  and  $[\phi:]$  are concerned, I shall discuss the error analysis for them in section 5.3.1.1.2., in order to test the hypothesis about the shortening of long vowel separately.

A common foreign accent feature which characterizes four labialized front short Danish [y],  $[\phi]$ ,  $[\phi]$ ,  $[\phi]$ ,  $[\phi]$  is that the Russian accent manifested itself in a wider range of the mispronounced sounds produced compared to the front non-labialized vowels. See **Tables 18, 19, 20, 21.** 

Target words	t/s	+
f <b>y</b> sisk		9
fysiologi	У	9
Lyngb <b>y</b>		9
<b>T</b> 11 46 <b>T</b>		

 Table 16. Target [y] by the R-group

Target words	t/s	+	[Y]*	[I]*
f <b>y</b> sisk		8		1
fysiologi	у	6		3
Lyngb <b>y</b>		7	2	

Table 17. Target [y] by the D-group

The pronunciations of the Danish front open [y] (see **Tables 16, 17**) in target words *fysisk, fysiologi* and *Lyngby* proved to be convincing evidence in support of Flege's idea about a better performance for dissimilar sounds: 15 out of 18 subjects gave a correct variant in the three words. The R-group pronounced all the target [y] correctly in stressed and unstressed positions, while in the D-group the influence of the Russian corresponding terms *fizichesky* (physical) and *fiziologiya* (physiology) resulted in 3 cases of [1]\* in the second prestressed syllable of *fysiologi* and 1 [1]\* in *fysisk*. Two subjects gave a front-mid allophone [x]\* instead of the final [y] in *Lyngby*, but this minor feature could be also heard in a native variant.

Target words	t/s	+	[ø <sup>.</sup> ]*	[y]*	[3]*	[y <sup>.</sup> ]*
kysse		2	1	5		1
nødvendig	ø	9				
nervøsitet (one failed)		6			2	

Table 18. Target [ø] by the R-group

Target words	t/s	•	[ø·]*	[y]*	[0]*	[ø]*	[3]*	[œ]*	[8]*
kysse nødvendig nervøsitet	Ø	3 7 6	1	5	1	1	1	1	1

## Table 19. Target [ø] by the D-group

A worse performance was shown for the target  $[\phi]$ . See **Tables 18, 19**. In a stressed position in *kysse*, only 5 correct readings were given, while the top realization was  $[y]^* - 10$  subjects. What is remarkable is that in the unstressed positions in *nødvendig* and *nervøsitet* the number of correct variants was more than convincing in both groups. It turned out to be that the subjects could easier produce  $[\phi]$  in the unstressed position rather than in the stressed one. I argue however that more words need to be pronounced with this target sound in order to formulate any regularity concerning the "superiority" of the unstressed position. What is only evident from our data is that  $[\phi]$  is most often mispronounced as  $[y]^*$ .

Target words	t/s	+	[y]*	[ø] *	[6]*	[ə]*	[ʊ] *	[ <b>θ</b> ]*	[œ]*
tr <b>y</b> kke		0	9						
ømtålelig	œ	0		2	2	3	1	1	
tørklæde		7							2

 φ
 7
 1
 1

## Table 20. Targets [@] and [@] by the R-group

Target words	t/s	+	[y]*	[o]*	[ø] *	[œ]*	[u]*	[ʊ] *	[Y]*	a[0]*	[œː]*	[ʉ]*
trykke		0	6				1	1	1			
ømtålelig	œ	2		1	2		1	1		1		1
tørklæde		5				1			1	1	1	

		+	[œ]*	[0]*	[0.]*
smør	Ē	3	4	1	1

Table 21. Targets [a] and [a] by the D-group

As far as the sounds  $[\phi]$  and  $[\phi]$  are concerned, the error analysis showed that these segments are generically susceptible to *narrowing* in terms of height to such qualities as  $[\phi]^*$ ,

 $[\mathfrak{G}]^*$ ,  $[\mathfrak{Y}]^*$  and  $[\mathfrak{G}]^*$  as well as a tongue retraction. See **Tables 20, 21**. The latter tendency results in such accent qualities as  $[\mathfrak{G}]^*$ ,  $[\upsilon]^*$  and even  $[\mathfrak{O}]^*$ .

However, another feature that characterizes  $[\phi]$ ,  $[\phi]$  and  $[\phi]$  production by the Russian learners is that in *trykke* (15 out of 18 subjects gave  $[y]^*$ ) the letter symbol is still superior to sound, i.e. the Russian speakers' pronunciation decisions are influenced by the correlation between the letter *y* and the sound [y]. The reason for that can be assumedly ascribe to the Russian morphophonological writing system in which letter-to-sound correlations are much less ambiguous than those in Danish.

## 5.3.1.1.2. Quantitative features

In this section I shall discuss the results of the error analysis for the long [ $\epsilon$ :], [ $\ddot{\alpha}$ :], [ $\phi$ :], [ $\phi$ :], [ $\phi$ :], [ $\gamma$ :] and [ $\phi$ :]. What was typical of all the vowel quantitative features is that they have been in many cases accompanied by *qualitative* changes. However, I shall focus on the shortening of long vowels and secondly on lengthening of short vowels, which was predicted in my hypotheses.

Target words	t/s	÷	[a]*	[e]*	*[3]	[ä]*	[e:]*	[A:]*
b <b>a</b> deværelse	C.	2		1	6			
bag <b>a</b> ge	ъ.	4	1		2		1	1
f <b>a</b> re	ä	8	1					
rare	u.	5				4		

Table 22. Targets  $[\epsilon:]$  and  $[\ddot{a}:]$  by the R-group

Target words	t/s	÷	[a]*	[e]*	*[3]	[ạː]*	[ạ]*	[ä]*	[a:]*	[ạ <sup>.</sup> ]*
badeværelse	33	1	4	1	3		4		1	1
f <b>a</b> re		4	4		1	1	2		1	1
rare	a:	4	1				2	2		

 Table 23. Targets [ɛː] and [ɑː] by the D-group

Of the two long segments [ $\epsilon$ :] and [ $\ddot{\alpha}$ :] (see **Tables 22, 23**) the first one was shortened in *badeværelse* by the overwhelming majority of the subjects from both groups (15 out of 18), while in *baggage* – if not reduced, it was produced qualitatively wrong as [e]\* or [ $\Lambda$ :]\*, [ $\alpha$ :]\* or [ $\dot{\alpha}$ ·]\*. As for the long [ $\ddot{\alpha}$ :], it was shortened by more than half of the subjects mostly to [ $\ddot{\alpha}$ ]\* in the R-group, and to either [ $\alpha$ ]\* or [ $\dot{\alpha}$ ] in the D-group. The long segment [e:] saw major shortening in the D-group, where only 2 subjects gave correct long variants, while the rest 7 were distributed as [e]\* (2),  $[\epsilon]$ \* (2),  $[\epsilon']$ \* (2) and  $[\epsilon']$ \* (1). See **Tables 24, 25.** 

Target words	t/s		[e <sup>-</sup> ]*
læse	e:	8	1

			[i]*	[i:]*	[i]*	[i <sup>-</sup> ]*
alene	e'	2		6		1
sene	<b>.</b>	1	2	5		1
skrive Kina	i:	7 7	2		1	1

Table 24. Targets [e:], [e:] and [i:] by the R-group

Target words	t/s		[I]*	[e]*	*[3]	[e <sup>.</sup> ]*	<b>[</b> ٤ <sup>·</sup> ]*					
læse	e:	2		2	2	2	1					
								_				
Target words	t/s	Ŧ	[i]*	[i:]*	[I]*	[i]*	*[3]	[ɛ·]*	[e]*	[e]*	[i <sup>-</sup> ]*	[i:
alene sene	ę:	3 3	2 1	1 1	1		1	2	1	1	1	
skrive Kina	i:	1 2	6 6			1					1	1

Table 25. Targets [e:], [e:] and [i:] by the D-group

The target sound [e:] in correspondingly *alene* and *sene*, was shortened by 9 D-subjects and 4 R-subjects, while [i:] in *skrive* and *Kina* by 4 R-subjects and 13 D-subjects, both including half-long pronunciations, e.g. [i<sup>·</sup>]\* and [ $\epsilon$ <sup>·</sup>]\*. See **Tables 24, 25**. What is remarkable here is that the D-group outnumbered the R-group by giving much more shortened variants. This can be ascribed to the fact that the R-group had<sup>15</sup> SIPT with a special focus on the distinction between short and long vowels, thus giving evidence to give credit to it.

Target words	t/s	+	[u]*	[y·]*	[uː]*	[Y]*	
r <b>y</b> ge	y:	5	1	1	1	1	
		+	[ø·]*	[y]*	[ø]*	[y:]*	
købe	ØĽ	3	1	2	2	1	

 $<sup>^{\</sup>rm 15}$  according to the information reported by the teachers of the R-group

		+	[œ]*	[œ·]*	[Ē] *	[Yː]*	[ʉ]*	[øː]*
høne	œ:	2	2	1		1	1	2
gøre	Ē.	6			3			

Table 26. Targets [y:], [ø:], [œ:] and [œ:] by the R-group

Target words	t/s	+	[u]*	[y]*	[uː]*	[Y]*			
ryge	y:	1	1	4	1	2			
						_	-		
		+	[ø·]*	[y]*	[ø]*				
købe	Ø:	4	2		3				
						-			
		+	[œ]*	[œ.]*	[œ]*	[œ:]*	[œ]*	[θ]*	[Œ.]*
høne	œ:	3	2	2	1	1			
gøre	Œ	3			1		3	1	1

Table 27. Targets [y:], [ø:], [œ:] and [œ:] by the D-group

The same tendency is true of the long [y:] - only 3 shortened \*-variants in the R-group and 7 \*-variants in the D-group of which 4 were mispronounced as [y]\*. See **Tables 26, 27**. A long  $[\varphi:]$  in *købe* underwent primarily quantitative changes in both groups being shortened to either  $[\varphi]$ \* or  $[\varphi^{\cdot}]$ \*. The same is true of the R-group's realizations of  $[\varphi:]$  in *gøre*, while in the D-group the shortening was accompanied with minor qualitative changes, primarily narrowing. The long  $[\varphi:]$  in *høne* turned out to be a challenge for all the subjects and its readings gave only 5 out of 18 possible correct ones. The mispronounced variants were  $[\varphi:]$ \*, [Y:]\*,  $[\varphi]$ \*, [u]\*, and  $[\varphi:]$ \* in the R-group. The D-group gave more shortened variants.

## 5.3.1.2. Back vowels

In the theoretical contrastive study, my main idea concerning back Danish vowels was that since the Russian phonological inventory of back vowels is limited to very few vowel phonemes, mainly /o/, /u/ and /a/, the Russian foreign accent in back vowels would mainly lie in fewer back vowel distinctions. My hypotheses were that the most challengeable would be the distinctive articulation of [o], [o], [o] and [A], the distinction between vowel qualities [u] and [o], and that the articulation of [u] would be most native-like of all the back vowel qualities due to its phonological properties identical to those of the Russian stressed /u/.

Here we can hardly say that Flege's principle of equivalence classification could work for the phoneme /u/, since we should not speak in terms of "more or less" similar, but rather in terms of a total similarity. While in the case of [0], [2], [2] and [4] the equivalence classification would be right to the point since for any Russian native speaker these four distinct short vowel qualities would resemble the allophones Russian /o/.

## 5.3.1.1.1. Qualitative features

Target words	t/s	÷	[ʊ]*	[uː]*
r <b>u</b> tine ak <b>u</b> punktur k <b>u</b> lde	u	5 6 8	4 3	1

		+	[0]*	[u]*	[ɔ]*
g <b>o</b> d		5		4	
irokeser	<b>o</b>	3	5		1
b <b>o</b> liviansk	-	2	7		

Table 28. Targets [u] and [o] by the R-group

Target words	t/s	÷	[ʊ]*	[u <sup>.</sup> ]*
r <b>u</b> tine ak <b>u</b> punktur k <b>u</b> lde	u	8 8 8	1 1	1

		+	[o]*	[u]*	[၁]*
god		2	6	1	
ir <b>o</b> keser	Q	0	7		1
boliviansk		1	7		1

Table 29. Targets [u] and [o] by the D-group

Having analyzed the collected data, I can substantiate my hypothesis about the nativelike articulation of /u/ due to its correct variant as [u] in target words by the overwhelming majority of the subjects in a stressed position in *kulde* – in 16 out of 18 samples. See **Tables 28, 29**. On top of that, in the unstressed positions in *rutine* and *akupunktur*, it was reduced to  $[\upsilon]^*$  by a total of 5 and 4 subjects respectively for each word, which can be considered as foreign accent within a norm, because such a minor reduction towards a more closed and central allophone  $[\upsilon]$  could be traced even in native speakers of Danish.

As far as [0] is concerned it was, by and large, mispronounced in both groups of subjects as either  $[0]^*$ ,  $[u]^*$  or in few cases as  $[0]^*$  in *god*. See **Tables 28, 29.** What is interesting in the error systematization, is that for this stressed position the R-group demonstrated a clear tendency towards  $[u]^*$  articulation instead of [0] (4 out of 4 mispronounced), while the D-group gave more  $[0]^*$ -variants (4 out of 5 mispronounced).

From the point of view of equivalence classification both of these accent realizations are quite predictable, since [o] "lies between" [o] and [u] for a native Russian speaker. However, the fact that the R-samples were collected in two academic groups of students who had a similar instruction during their SIPT may lead to the conclusion that [u]\*-variants of the mid [o] was due to instruction reasons. Whereas in the D-group, one can evidently trace the spelling interference, mentioned above in section 5.3.1.1.

The effect of spelling interference also manifested itself in the unstressed syllables of *irokeser* and *boliviansk* with 5 and 7 variants [o]\* respectively in the R-group and 7 ones in the D-group. It should be noted, however, that even though [o]\* in the pre-tonic syllable of *irokeser* can be heard in native speakers, in the Russian samples [o]\* had a purely Russian quality (see section 3.1.). This influence of the spelling can be of two types – one coming from subconscious associations with one's mother tongue, as was the case with *dreng*, *boliviansk*, *irokeser*, and the other – from a direct correspondence between an L2 sound and its representation in spelling.

Target words	t/s	+	[u]*	[ų]*	[Y]*	[ʊ]*	[o]*
l <b>u</b> kke	•	0	7	1	1		
<b>u</b> ngdommelig	ş	0	2			6	1
	4 F		41 1				

 Table 30. Target [9] by the R-group

Target words	t/s	+	[u]*	[ų]*	[ø]*	[ʊ]*	[ɔ]*	[o]*	[Ŏ.]*	[ə]*	[ʉ]*
l <b>u</b> kke		2	3	2	1		1				
<b>u</b> ngdommelig	Ş	1	1			4		1	4	1	1

Table 31. Target [9] by the D-group

An evident and convincing prevalence of /u/-allophones in the readings of  $[\mathfrak{g}]$  (see **Tables 30, 31**) as the target sound in *lukke* (10 out of 18 target sounds were pronounced as [u], 1 as [u]) and *ungdommelig* (3 as  $[u]^*$ , 4 as  $[\upsilon]^*$ , 1 as  $[u]^*$ ) gives evidence in favor of the spelling interference, and makes us believe that the spelling plays a huge role in the Danish pronunciation of Russian native speakers. This leads to the conclusion that learning though the "written" language and written input still works as a main factor in forming a learner's L2 pronunciation habits in the current case study.

In my hypotheses about the features of the Russian foreign accent in back vowels, I did not predict any weakening of back vowels. However, similarly to the above-mentioned nonpredicted front [ $\varphi$ ]-weakening, [ $\varphi$ ] was exposed to [ $\upsilon$ ]\*-weakening – this degree of reduction is typical of the Russian /u/- in the pre-stressed position in *ungdommelig* by respectively 6 R-subjects and 4 D-subjects. Such an acting on analogy with the Russian /u/-weakening in the case of [q] supports the conclusion that replacing [q] with [u] can be regarded as a typical feature of the Russian accent in Danish, especially when in spelling the letter *u* stands for the sound [q], i.e. as the result of the above-mentioned spelling interference.

If one looks at the results of the error systematization for the back vowels from the point of view of equivalence classification, one will see certain regularity. Those Danish vowel segments which are open-mid and close-mid –  $[\rho]$  and the advanced  $[\rho]$  respectively – tend to be mispronounced by Russian native speakers as  $[u]^*$  or more rarely as the allophones of the Russian /o/, the latter is more typical of  $[\rho]$ .

Target words	t/s	÷	[၁]*	[ǫ]*	[0]*	[v]*
storme		6	3			
borgmester	•	6	1	2		
korrupt	Ş	0		1	7	1
korrektur		6	3			

		Ļ	[0]*	[ɔ]*	[ɔָ]*	[ə]*	[ɔː]*	[ß]*
ånder		2	2		4		1	
forskellig	Ą	1	5	1		2		
nærmere		5				3		1
						-		

 Table 32. Targets [ɔ] and [ʌ] by the R-group

		+	[ɔ]*	[ǫ]*	[0]*	[θ]*			
storm		5			4				
borgmester	•	6	3						
korrupt	Ş	1	1	1	7				
korrektur		2	3		3	1			
Target words	t/s	+	[o]*	[၁]*	[ɔ]*	[ə]*	[θ] *	[e]*	[ɐ]*
ånder		3	1	2	3		1		
forskellig	Ą	3	1	2		2			1
nærmere		4				4		1	

Table 33. Targets [2] and [A] by the D-group

As far as  $[\mathfrak{d}]$  and  $[\mathfrak{d}]$  are concerned, the results of the error analysis for these targets are quite varied. See **Tables 32, 33**. In the stressed positions, 6 R-subjects and 4 D-subjects gave correct pronunciations of  $[\mathfrak{d}]$  in *storm* (in this target word *stød* was neglected as was methodologically introduced in section 3). In the stressed position  $[\mathfrak{d}]$  was pronounced as a less lowered  $[\mathfrak{d}]^*$ , while 4 D-subjects demonstrated a vivid equivalence classification by giving 4 cases of  $[\mathfrak{d}]^*$ . The latter could be traced in the R-group only for the first pre-stressed

syllable. In the unstressed positions in *borgmester* altogether 12 correct variants were given first of all due to a wrong word stress assigned to the first syllable. Even though in both *korrupt* and *korrektur* the words stress was set correctly, no vivid evidence in support of any weakening could be traced for this vowel segment, except for the already-mentioned feature to produce it as either a more open  $[\mathfrak{d}]^*$  or like  $[\mathfrak{d}]^*$ .

A much more non-native like performance was shown for the words with  $[\Lambda]$  as a target sound with 5 correct varinats in the stressed position in *ånder*. The R-subjects mispronounced it in the stressed position as either  $[\mathfrak{q}]^*$  or  $[0]^*$ , while the D-subjects produced also 3 lowered  $[\mathfrak{q}]^*$  and 1 close-mid  $[\mathfrak{s}]^*$ . This vowel segment was exposed to a qualitative reduction in both groups: in *forskellig* the unstressed  $[\Lambda]$  was reduced to the neutral  $[\mathfrak{q}]^*$  all in all by four subjects but, not as a regular weakening pattern would require in Russian for the vowel  $[\mathfrak{q}]^*$ in the first pre-stressed syllable, and also in *nærmere* to neutral  $[\mathfrak{q}]^*$  by a total of 7 subjects.

Thus, we can see that qualitatively the most native-likely produced back vowel in Russian speakers of Danish is [u]. Vowel [ $\varphi$ ] is typically mispronounced as either [u]\* or [ $\phi$ ]\*, where the latter is assumedly caused by both spelling interference and equivalence classification. Vowel [ $\varphi$ ] is most often mispronounced as [u]\* in the stressed positions and can be reduced to [ $\sigma$ ]\* in the pre-stressed ones. The two back vowel segments [ $\phi$ ] and [ $\varphi$ ] are both subject to narrowing in the Russian foreign accent. The former one is most likely to be produced as either [ $\varphi$ ]\* or [ $\phi$ ]\*, whereas [ $\varphi$ ] beyond [ $\varphi$ ]\* or [ $\phi$ ]\*, would be also produced as [ $\sigma$ ]. On top of that, Russian natives, as the data showed tend to reduce [ $\phi$ ] to [ $\varphi$ ]\* in the pre-stressed and final post-tonic positions.

It is possible to conclude that in the case of the Danish back vowels, Russian learners tend to resort to the usage of the familiar sound inventory, namely back vowels [u] and [o], with a worse distinction of other back vowel qualities subject to narrowing.

5.3.1.1.2. Quantitative features

Target words	t/s	+	[u]*	[u?]*	[u <sup>.</sup> ]*
m <b>u</b> ligvis	11'	3	3	3	
uge	u.	8			1

		Ŧ	[Ĵ.]*	[ɔː]*	[0.]*	[o:]*	[ɔ៎.]*
m <b>å</b> le	ð:	4			2	2	1
storme	Ş:	6	1	2			
						_	

Table 34. Targets [u:],  $[\mathfrak{z}:]$  and  $[\mathfrak{z}:]$  by the R-group

		+	[ʊ]*	[u]*	[u <sup>.</sup> ] *
rutine		8	1		
ak <b>u</b> punktur	u	8	1		
kulde		8			1
m <b>u</b> ligvis		1		6	2
uge	u.	2		7	

		÷	[ɔ͡]*	[θ·]*	[J.]*	[ɔ]*	[ɔː]*	[oː]*	[ǫ]*	[ð.]*	[0:]*	[0]*	[œː]*	[ɔ͡]*	[ɔ៎.]*
m <b>å</b> le	ð:	0	1	1	1		1		1	1	1	1	1		
st <b>or</b> me	Ş:	3			1	3								1	1

Table 35. Targets [uː], [ɔ̣̣ː] and [ɔ̣̣ː] by the D-group

In this section I shall discuss the results of the error analysis for the long [9:], [u:], and [9:]. See **Tables 34, 35**. These long vowel segments are also covered by the hypothesis about shortening of long vowels, and the collected data again serves to verify it.

Compared to [u] which was produced correctly by most of the subjects, the long [u:] scored only 14 out of possible 36 readings of together *muligvis* and *uge*. It was primarily shortened to  $[u]^*$  or  $[u^{-}]^*$  - twice as often in *muligvis*, than in *uge*.

As for the sound  $[\mathfrak{g}:]$  (see **Tables 34, 35**), I have to admit that methodologically this vowel was set in a very favorable distribution compared to  $[\mathfrak{u}]$  in *muligvis* for instance. In the open syllable of *måle* it should have been less exposed to shortening conditioned that the subjects knew about the regularity of a long vowel in an open syllable. However, the error analysis showed the opposite. Two out of 9 R-subjects shortened  $[\mathfrak{g}:]$  to a half-long  $[\mathfrak{o}:]^*$  and 1 to  $[\mathfrak{g}:]^*$ . The D-subjects did not produce any correct pronunciations and gave seven shortened variants as  $[\mathfrak{g}]^*$ ,  $[\mathfrak{o}:]^*$ ,  $[\mathfrak{o}:]^*$ ,  $[\mathfrak{g}:]^*$ ,  $[\mathfrak{g}:]^*$  and  $[\mathfrak{o}:]^*$  - one of each respectively. Thus, quantitatively there *is* an evident feature of shortening a long vowel.

As was previously mentioned, the samples of the D-group are characterized by fewer correct productions of sounds in terms of the length of vowels. I argue for the fact that a weaker performance for the long vowels arises to the problem of a lacking pronunciation training focused on the opposition between long and short vowels. Therefore, [9:]-readings in target *storme* showed again, that the R-subjects (6 long variants) are better at preserving distinction between long and short vowel phonemes than the D-subjects (only 3 out of 9), because the training of the long vs. short opposition was one of the core aspects in SIPT.

#### 5.3.1.3. Shortening of long vowels

Having analyzed the quantitative characteristics of all the vowel segments, I can conclude that my hypothesis about the shortening of long vowels proved to be consistent and correct, while the hypothesis about the lengthening of short vowels could not been substantiated, at least in the current research, because according to the error systematization, only very few examples of the long vowel shortening were collected (see **Figure 5**).

- ✓ *pande* [*q*:]\*(2)
- ✓ binde [i:]\*(1)
- ✓ forbillede  $[i^{\cdot}]^*(1)$
- $\checkmark kysse [y]^*(l)$
- ✓ ånder [ɔ̯ː]\*(1)
- $\checkmark kulde [u]^*(1)$
- ✓ *kysse* [*ø*<sup>·</sup>]\*(1)
- ✓ tørklæde [œ:]\*(1)
- ✓ ungdommelig [œ<sup>·</sup>]\*(4), [o<sup>·</sup>]\*(1)

# Figure 5. Examples of the long vowel shortening as according to the error systematization

If we consider the number of all the lengthened variants in relation to all the 17 words with a long target vowel multiplied by 18 samples that would make 10 cases out of 306, i.e. 3.3%, which is methodologically very inconsiderable and cannot be used to verify the hypothesis.

## 5.3.1.3. Neutral [*ə*]

As was described in section 3.1.2., the Russian patterns of weakening vowels in the unstressed positions imply a Degree 2 reduction to [ə] of the phonemes /a/, /o/ and / $\epsilon$ / in second pre-stressed syllables and all the post-tonic syllables. In this connection, the Danish neutral [ə] is not a new vowel quality for Russian native speakers even though in terms of phonotactics and phonological distribution, [ə]-occurrence in Danish may differ from that one in Russian.

For the reading materials, according to the pronunciation guidelines in Den Store Danske Ordbog, the following target words for the neutral [ə] were chosen: *ganske* (the final schwa after a consonant phoneme), *bue* (final schwa after a vowel phoneme) and *doven* 

(schwa in the final closed syllable after a sonorant consonant). It is necessary to admit that the recommended pronunciation taken as a point of departure for the error analysis was only relevant for the target word *ganske*, while in *bue* and *doven* the presence of schwa is highly questionable in native speakers' pronunciations. Therefore, methodologically the choice of *bue* [bu.:] and *doven* [down] was not to the point, thus it is only possible to make conclusions about [o]-pronunciation in *Ganske, where* all the 18 subjects pronounced the target schwa correctly. See **Tables 36, 37**.

Target word	t/s	+					
gansk <b>e</b>	ə	9					
bu(e) dov(e)n	methodologically invalid: all the R-subjects pronounced [ə]						

Table 36. Target [ə] by the R-group

Target word	t/s		
ganske	ə	9	
bu( <b>e</b> ) dov( <b>e</b> )n	metl [ə]* [ə]*	nodol (4), [ (5), [	ogically invalid; <sup>°</sup> ]*(3), [i]* (1), [ɛ <sup>·</sup> ]* (1) in bue; -] (4) in doven.
			(1 D

 Table 37. Target []
 by the D-group

However, what was valuable in the readings of *doven*, is that 4 out of 9 D-subjects reduced it to zero, as natives would do. None of the R-subjects pronounced the final syllable without [ə].

Such results of the error systematization can lead to the following conclusions. Firstly, the two groups had different guidelines while dealing with the neutral central [ə]. I found out that SIPT included [ə]-articulation in the R-group, and thus the R-subjects were aware of the [ə]'s typical distribution and occurrences, but they are still very dominated by the word's spelling, and therefore, gave the neutral [ə] triggered by the letter *e*, even in *bue* and *doven*, where it would not be pronounced by native speakers. Whereas the D-group tend to be guided more by their exposure to native input, where [ə] can be exposed to different degrees of reduction depending on an individual speech rate or assimilation processes. This difference in the learning approach explains why the R-subjects pronounced [ə], where it would not be pronounced, while the D-subjects, who mostly learned it through imitation rather than explanation, tend to reduce it to zero as Danes do in a natural speech.

Thus, I can conclude that since [ə]-realizations are in real speech closely connected with the type of distribution and influenced by assimilation processes as well as individual properties of speech such as rate, for instance, foreign speakers of Danish may assimilate different strategies of [ə] production depending on the instruction input they receive.

Methodologically speaking, proceeding critically from the results of the data analysis I conclude that accent in [a] should be studied in the framework of either a spontaneous speech task or reading tasks with a text reading, where [a] would be in an environment close to that f a natural speech.

## 5.3.1.4. Diphthong features

The WL2 included words with the following target rising [jü], [ju], [jy], [ja] and falling [qew], [yw], [ew], [qw], [qw], [iw], [qw], [uj/i], [aj], [Aj],  $[\epsilon j]$ , [aw/au], [iv],  $[\epsilon v]$ , [ev], [ev], [yv], [qv], [

#### 5.3.1.4.1. Consonantization of the glide

The consonantization of a glide was typical of [w]-diphthongs and [y]-dithphongs. In the examples below from the samples read by the R- and D-subjects we can see that semivowels [w] and [y] are produced as if they were full consonants qualities of the phonemes /r/ and /v/.

• [w]-diphthongs

in *tyveri* as [yv]\* (4);
in *evnesvag* as [εv/ ε<sup>v</sup>]\* (6);
in *peber* with [b]\*- or [b]\*-glide;
in *øvre* as [øv]\* (2) or [œv]\* (1);
in *automatisk* as [äv]\* (1);

• [v] dithphongs:

in *kørsel* and *ørred* as  $[\Phi \&]^*$  (5),  $[\emptyset \&]^*$  (2),  $[\Theta \&]^*$  (1),  $[\Theta \&]^*$  (3); in *urbanisere* as  $[\upsilon \&]^*$  (2).

## 5.3.1.4.2. Nucleus replacement

The nucleus replacement as a feature can be defined as a change of the diphthong's prominent vowel's quality in either height or backness. Here are the most illustrative examples of this phenomenon:

- [jä] in *kajak* was produced as mostly as [ja]\* (5);
- [œw] in *søvnløs* was narrowed to [θə]\* (1), [θw]\* (3), [œw]\* (1), [œ<sup>u</sup>]\* (1), [YY]\* (1) or [œu]\*(2); or pronounced with a more open nucleus vowel back e.g. [bw]\*(1), [ow]\* (1);
- [ew] in <u>evnesvag</u> was either narrowed as [iw]\* (4) or opened to [εv/ε<sup>v</sup>]\* (6); in *jævnaldrende* was opened to [æw]\* (2), [aw]\* (3) or [ɐw]\* (1);
- [<code>;w]</code> in *lovgivning* was narrowed to [yv]\* (3);
- [ɛj] in *evnesvag* was opened to [aj]\* (1), or back [ɐj]\* (1);
- $[\Lambda j]$  in *fløjet* was narrowed to  $[\check{y}j]^*(1)$ ,  $[jj]^*(5)$  or centralized to  $[\mathfrak{u}j]^*(1)$ .

# 5.3.1.4.3. Monophthongization

The error systematization showed that a number of subjects tend to convert diphthongs into monophthongs. Here are examples of monophthongization as a feature of the Russina pronunciation in Danish:

- [ju] in *skjulte* as [v]\* (1);
- [uj/i] in *huje* as [u]\* (6);
- [yɐ̯] in *dyrke* as [ʊ]\* (1), [y]\* (4), [y:]\* (2);
- [ug] in *urbanisere* as [u]\* (4), [v]\* (10);
- [ɔ̥ɐ̯] in *bortfalde* as [o̯]\* (3), [o]\* (13);
- [iɐ̯] in *kirkelig* as [i]\* (3);

- [φ𝔅] in *hjørne* as [φ]\* (8), [θ]\* (2), [φΞ]\* (6);
- [εj] in *evnesvag* as [ε]\* (1), [ε]\* (2);
- [äu] in *automatisk* as [ɔ]\* (4);
- [øw] in *øvre* as [œ]\* (1);
- $[\Lambda j]$  in *fløjet* as  $[\Theta^j]^*(4)$ ,  $[\mathfrak{z}]^*(1)$ ,  $[\mathfrak{a}]^*(1)$ ;
- $[\emptyset \mathfrak{p}]$  in *kørsel*  $[\emptyset :]^*(4)$ ,  $[\mathfrak{p}]^*(1)$ ,  $[\emptyset]^*(3)$ ,  $[\mathfrak{p}]^*(1)$ .

What is typical of all the above-presented monophthongized diphthongs is that their nucleus remains more or less preserved while a less prominent semivowel component is omitted, or alternatively a whole diphthong is monophthongized into a new vowel quality as was the case in *automatisk* with  $[o]^*$  instead of [äu]. (The latter is assumedly the result of the third language transfer, namely from English into Danish, since all the four subjects who mispronounced [äu] as  $[o]^*$  are advanced English-speakers.) In some situations, this new vowel quality is lengthened, as was the case in *kørsel* with  $[\phi]^*$ .

Thus, I substantiated the hypotheses about the monophthongization of diphthongs and consonantization of the glide, but have to admit that these two are not the only features of the Russian accent, but should be considered as a property of the Russian foreign accent together with the qualitative replacement of the diphthong nucleus.

## **5.3.2.** Consonant features

Proceeding from SLM's principle of equivalence classification, I earlier in section 3.2. formulated the hypotheses about the features of the pronunciation of consonants in the case of the Russian accent by means of the contrastive method, having compared two phonological systems, and predicted the following eventual realizations of the Russian accent in consonant segments:

- disaspiration of /p/, /t/, /k/;
- velarization of /l/;
- voicing of non-aspirated consonants /b/, /d/, /g/, also of the intervocalic [s].
- palatalization of /b/, /d/, /g/ especially before Danish front /i/, / $\emptyset$ /, / $\epsilon$ /, /y/ and /e/.
- dentalization of /d/, /s/, /t/ and /n/;
- /r/-assimilation to a thrilling /r/; "consonantization" of [ŋ];
• [ts]-overtone in the Danish /t/<sup>16</sup>.

The results of the error systematization for segments [f] in *flame*, *gaffel*, *falsk*, *film*, [j] in *jod*, [w/u] in *kniv*, *koge*, *brev* and [i] *dej kaj* showed no evident foreign accent in these consonants in either groups of the participants. However,  $[\eta]$  in *banke*, *gange*, *bang* was opposite to my hypothesis realized 3 times as  $[\eta k]^*$  in *bang* and 5 times as  $[\eta g]^*$  in *gange*. The fact that  $[\eta]$  was mispronounced only by the D-subjects leads to the conclusion that this was most evidently due to lacking corrections or/and training of this sound in case of particular subjects, rather than due to any kind of spelling interference or equivalence classification.

The foreign accent in the pronunciation of all the rest target consonant segments had to a more or less considerable degree of regularities. I classified the typical features of the pronunciation of consonants in the following ones:

- disaspiration;
- voicing;
- assimilation to the Russian segment and dentalization;
- palatalization and velarization;
- other features.

## 5.3.2.1. Disaspiration

Target words	t/s	÷	[p]*
<b>pi</b> skefløde	Å <sub>p</sub>	6	3
pa <b>p</b> ir		6	3

		÷	[k]*	[kj]*	not
					read
kirsebær		5	3	1	
økologi	$\mathring{g}^{\mathrm{h}}$	8	1		
kone		5	3		1

Table 38. Targets [b<sup>h</sup>], [g<sup>h</sup>] by the R-group

<sup>&</sup>lt;sup>16</sup> See section 3.3.

Target words	t/s	+	[þ]*	[p]*
<b>p</b> iskefløde pa <b>p</b> ir	Å <sub>p</sub>	4 2	1	4 7

		+	[k]*
kirsebær		5	4
ø <b>k</b> ologi	$\mathring{g}^{\rm h}$	6	3
kone		8	1

Table 39. Targets [b<sup>h</sup>], [g<sup>h</sup>] by the D-group

Danish aspirated consonants  $[b^h]$  in *piskefløde* and *papir*, and  $[b^h]$  in *kirsebær*, *økologi* and *kone* were mispronounced by a series of subjects from both groups. See **Tables 38, 39**. In the whole,  $[b^h]$  was disaspirated in 50% of all the readings in the two target words (18 out of 36). It was disaspirated  $[p]^*$ -likely, i.e. pronounced as the Russian hard [p]. Only one D-subject produced  $[b]^*$  instead of  $[b^h]$ . The subjects from both groups showed a better performance for  $[b^h]$  in *kirsebær*, *økologi*, and *kone*, with 37 out of 54 possible correct variants: one target word *kone* was omitted by one subject, while in 13 cases  $[b^h]$  was read as  $[k]^*$  and in 1 case – as  $[k^j]^*$  in *kirsebær*, assumedly due to the front [i], which would require the palatalization of a preceding consonant in Russian.

Anyway, I can conclude that Russian native speakers tend not to aspirate  $[b^h]$  and  $[g^h]$ . The reason for that is twofold. On the one hand, the disapiration comes from the spelling interference – directly for the letter *k* and inter-language for the letter *p* (*n* in the Russian alphabet). As the result of spelling interference, *k* is realized as /k/ or /k<sup>j</sup>/ and *p* as /p/ or /p<sup>j</sup>/ (depending on the vowel distribution) due to the equivalence classification, since the Danish  $[b^h]$  and  $[g^h]$  are most similar to the Russian /k/, /k<sup>j</sup>/, /p/, /p<sup>j</sup>/.

One can draw a conclusion that from the very beginning of instruction Russian learners associate sounds  $[b^h]$  and  $[g^h]$  with the corresponding letters p and k, which is far from being true in many Danish words, where letters p and k are not syllable-initial and thus represent sounds [b] and [g] respectively. Even though learners may *be* aware of aspiration as a property of these two Danish sounds, they cannot automatize the aspiration even at an advance level, which may in its turn result in mistakes in minimal pairs.

t/s	+	[ģ]*	[t]*	[ <del>Î</del> s]*
	3			6
ds	5	1	1	2
	4			5
	<b>t∕s</b> ∲s	t/s         3           \$\hat{q}\$\$s         5           4	t/s         •         [\dots]*           \$\dots\$         3         1           \$\dots\$         5         1           4         4         1	t/s       I $[d]^*$ $[t]^*$ $\hat{qs}$ 3       1       1 $\hat{qs}$ 5       1       1         4       4       4       4

 Table 40. Target [ds] by the R-group

Target words	t/s	+	[t]*	[ <u>t]</u> *	[ <u>t]</u> *	[t]*
tøj		2	5	1	1	
de <b>t</b> al <b>j</b> e	ds	3	1	4		1
øm <b>t</b> ålelig		2	5	2		

Table 41. Target [ds] by the D-group

As far as the aspirated sound  $[\hat{qs}]$  is concerned (see **Tables 40, 41**), it was realized differently in the R- and D-groups. However, the features discovered in the samples of both gourps, verify the hypothesis about 1) the disaspiration of Danish /t/ and 2) a  $[\hat{ts}]^*$ -overtone, i.e. the replacement of  $[\hat{qs}]$  with the most similar Russian affricate  $[\hat{ts}]$ . The latter was typical of the R-group in *tøj* (6), *detalje* (2) and *ømtålelig* (5)<sup>17</sup>. The former feature manifested itself in the D-samples as either  $[t]^*$ ,  $[\underline{t}]^*$  or  $[t]^*$ .

## 5.3.2.2. Voicing, assimilation and dentalization

I predicted that the opposition voiced vs. devoiced as an inseparable characteristic of the consonant phonological inventory in Russian, can be transferred into Danish. This hypothesis proved to be true regarding a series of Danish consonant segments. What is characteristic of this transfer is that it turned out to be many-folded, and manifested itself differently for different positions of Danish consonant segments in the target words. See **Tables 42, 43**.

Target words	t/s	+	[b]*	[p]*
bestemme		6	3	
ho <b>pp</b> e	h	8		1
skar <b>p</b>	Ŷ	9		
ondska <b>b</b>		7		2

		÷	[k]*	[g]*	[-]*	[g <sup>j</sup> ]*	[ĝ <sup>h</sup> ]
gulv	ĝ	4	1	4			

<sup>&</sup>lt;sup>17</sup> Out of 9 possible.

begejstret	2		4	2	1	
skinne	2	5				
fræ <b>k</b>	9					
<b>g</b> emme	6		1			2

Table 42. Targets [b], [g] by the R-group

Target words	t/s	÷	[b]*	[p]*
bestemme		3	6	
hoppe	h	6		3
skar <b>p</b>	Ŷ	4		5
ondska <b>b</b>		5		4

		+	[k]*	[g]*	[-]*	[ɣ]*	[g <sup>j</sup> ]*
<b>g</b> ulv		4		5			
be <b>g</b> ejstret		0		8	1		
skinne	ĝ	2	7				
fræ <b>k</b>		5	3			1	
gemme		4		3			2

Table 43. Targets [b], [g] by the D-group

In the initial position of the unstressed syllable in *bestemme* [b] was mispronounced as the voiced [b]\* by 3 R-subjects and 6 D-subjects, assumedly as the result of equivalence classification since, [b] and [b] are both bilabial plosives. In *hoppe, skarp* and *ondskab* [b] was mispronounced as the most similar Russian bilabial voiceless [p]\* in both groups. In *hoppe* [b] was mispronounced as the result of letter-to-sound correlations between Danish *p* and Russian sound [p]. Whereas in *skarp* and *ondskab* [p]\* was predictable, because in Russian a final bilabial plosive would be always [p]. Considering that the Danish [b] is voiceless in the final position, this feature of the Russian pronunciation would hardly affect the comprehensibility. In *hoppe* the only possible origin of [p]\* goes to the alreadymentioned spelling interference. Thus, we can see that the mispronunciation of [b] can take different directions (voicing to [b] or assimilation (as the result of equivalence classification) to the Russian [p]) depending on both position of the sound and the corresponding letter representing this segment.

Another consonant segment (see **Tables 42, 43**) that was voiced in the syllable-initial position, and assimilated to the most similar Russian sound [k] after *s* and in the syllable-final position, is [g]. It was mispronounced as [g]\* (25 out of possible 54) in *gulv, begejstret* and *gemme;* as [k]\* (15 out of possible 36) in *skinne* and *fræk* respectively.

Complicated features characterize the pronunciation of [d]. See **Tables 44, 45**. Here again we deal with the equivalence classification and spelling interference depending on the

position. In *storm* and *sytten* [d] was in the most favorable position<sup>18</sup> from the point of view of the pronunciation hints (after *s* and double *t*), and thus none of the R-subjects mispronounced it, while the D-subjects produced it in these two words as voiced alveolar [t]\* (6) and dental [t]\* (3).

Target words	t/s	+	[ģs]*	[ď]*	[d]*	[t]*	[ts]*
dum storm sytten dyne tidligt	ģ	3 9 9 3 3	1	1	5	1 4	1

Table 44. Target [d] by the R-group

		÷	[d]*	[ds]*	[-]*	[d]*	[d]*	[t]*	[ <u>t]</u> *	[t]*
dum		3	1			3	2			
storm		5						2	2	
sy <i>tt</i> en	d	4						4	1	
dyne		2				5	2			
tidlig <b>t</b>		2		1	2			2	1	1

Table 45. Target [d] by the D-group

Dissimilar phenomena characterized the pronunciation of [d] in *dum* and *dyne*, with the initial *d*. Being aware that [d] is a voiceless sound in Danish, one R-subject pronounced it as  $[t]^*$  in *dum* as the result of hypercorrection, mentioned above. Whereas the majority of the participants voiced it to the alveolar  $[d]^*(8)$ , dental  $[d]^*(14)$ ,  $[d^j]^*(1)$ , or  $[d]^*(1)$  – out of 36 possible native-like pronunciations. Here we see a clear feature of voicing before a vowel and dentalization analogously to the Russian [d].

In the final position in *tidligt* [d] was realized as  $[t]^*$ ,  $[ds]^*$ ,  $[ts]^*$ ,  $[t]^*$ ,  $[t]^*$  or omitted. Thus, no clear regularity can be found so far for the final position except for saying that [d] in the final position would be most likely classified by Russians as a sound similar to /t/.

Target words	t/s	+	[ <u>s</u> ]*
presse		9	
savne		5	4
fy <b>s</b> isk	s	9	
læse		3	

<sup>&</sup>lt;sup>18</sup> Here I mean that learners of Danish are usually aware that p, t, k after after s or double p, t, or k are never pronounced with aspiration. This is one of the basic rules about aspiration traditionally explained to learners of Danish.

		÷	[ŋ]*	[n <sup>j</sup> ]*
norsk		8	1	
ku <b>nd</b> e	n	9		
næste		8		1

Table 46. Targets [s], [n] by the R-group

Target words	t/s	+	[s]*	[Z]*
presse		6	3	
savne		5	4	
fy <b>s</b> isk	s	5	1	4
læse			2	2

		÷	[ <u>n</u> ]*
norsk		5	4
ku <b>nd</b> e	n	6	3
næste		5	4

 Table 47. Targets [s], [n] by the D-group

The voicing as a foreign accent feature is typical not only of the Russian natives' pronunciation of Danish segments [b], [d], [g] in a syllable-initial position. As the data analysis showed, the alveolar [s] in an intervocalic position can be also exposed to voicing as  $[z]^*$  (see **Tables 46 47**), which does not exist as a phoneme in Danish and may only seldom occur as an alophone as the result of assimilation. Relying on the data, I argue however, that this feature of the Russian natives' pronunciation may vary from an individual to individual, and may be determined by a particular type of phonetic instruction. As is seen from the error systematization this feature was only typical of the D-subjects in *læse* and *fysisk* (2 and 4 [z]\* respectively for each word). The voicing however was not however a major accent feature of [s]-articulation.

While the dentalization was an additional feature of [d]-articulation by the participants, it was a major one in the articulation of the Danish alveolar segments [s] and [n] (see **Tables 46, 47**). This accent feature fits into SLM, since these sounds have a certain degree of similarity with the Russian [s] and [n] and differ only in their place of articulation as was mentioned in the comparative study. In the R-group the dentalization was more typical of the [s]-targets in *savne* rather than of the [n]-targets - only 1 variant as [n]\* in *norsk*. The

dentalization of [s] and [n] was a more vivid accent feature in the D-group: presse (3), savne (4), fysisk (1), lase (2); norsk (4), kunde (3), naste (4)<sup>19</sup>.

#### 5.3.2.3. Palatalization and velarization

As was mentioned in section 3, the phonological opposition palatalized vs. nonpalatalized is a crucial characteristic of the Russian consonant inventory. It would be hardly possible to imagine the Russian accent without any transfer of this consonant feature. The analysis of the reading samples verified that the palatalization affects Danish consonants in the position *before* front vowels, as it was predicted in the contrastive study, but additionally to the hypothesis I also have evidence in support of the palatalization *after* a front vowel and before the foreign accent induced  $[ə]^*$ . However, I cannot conclude that the palatalization manifested itself as a primary property of the Russian accent.

In both groups [h] in *hemmelig* was palatalized by an equal number of participants: in each group one subject mispronounced [h] as  $[x^j]^*$  and two as  $[h^j]^*$ . The sonorant segments [m], [n], [1] were also exposed to the palatalization, though the distribution of palatalization was uneven. Two R-subjects produced [m] as  $[m^j]^*$  in *menneske* and one R-subject realized [n] as  $[n^j]^*$  in *næste*. The liquid sonorant [1] was palatalized by some subjects from both groups: as  $[l^j]^*$  in *længe* (3), *bopæl* (9), *kulde* (4)<sup>20</sup>, while in *læmmekød* it was velarazied as [ $l^j$ \* analogously as it would be in Russian after /a/. Other not numerous cases of palatalization were realized in *vække*  $[v^j]^*$  (4), *gemme*  $[g^j]^*$  (2) and *begejstret*  $[g^j]^*$  (2).<sup>21</sup>

In my hypothesis I predicted that the Russian accent would have the palatalization of /b/, /d/, /g/ especially before /i/,  $/\epsilon/$ , /y/, /e/ in Danish. The data analysis showed that this assumption was right for /g/, but I should admit that also sonorant segments, as well as the voiced [v] and voiceless [h] may be palatalized.

<sup>&</sup>lt;sup>19</sup> In this line, the results are given for 9 D-participants.

 $<sup>^{20}</sup>$  Out of 18.

 $<sup>^{21}</sup>$  [g<sup>j</sup>]\* in *begejstret* was due to the mispronunciation of [aj] as [ej]\* which made the nucleus of the diphthong a front vowel.

## 5.3.2.4. Other consonant features

Target words	t/s	+	[1]*	[R]*	[γ]*	$[\hat{R}_{\ell}]_*$	[3]*	[R]*
<b>r</b> igdom		6				1	1	1
ris		7					1	1
beredskab	Ŕ	8			1			
irokeser		8		1				

		+	[Ŕ]*	[1]*	[-]*
bær		9			
ki <b>r</b> ke	<b>n</b>	9			
mo <b>r</b>	ĕ	8			1
gerne		9			

		+	[s̪]*	[ç]*	[§]*	[ʃ]*
sjov	0	5		3	1	
<b>ch</b> okolade	C	3		1	2	3

		+	[-]*
stri <b>d</b> e		8	1
ma <b>d</b>	ăv	1	
he <b>dd</b> et	Ō,	8	1
vasket		9	

# Table 48. Targets [ʁ], [ɐ], [ɛ], [ðv] by the R-group

Target words	t/s	+	[1]*	[R]*	[r]*	*[1]	[γ]*	[-]*
<b>r</b> igdom		4	2	1	1	1		
ris		6			2	1		
be <b>r</b> edskab	Ŕ	4			1	1	1	2
i <b>r</b> okeser		5			2	1		1

		+	[Ŕ]*	[1]*	[-]*	[ə]*
bær		8				
ki <b>r</b> ke	<b>n</b>	5	1	1	2	
mo <b>r</b>	Ĕ	5			4	
gerne		5			3	

		÷	[s]*	[Z]*	[s̪ <sup>j</sup> ]*	[s <sup>j</sup> ]*	[s <sup>j</sup> ]*	[ç]*	[§]*	[t∫]*	[ʃ]*
<b>sj</b> ov <b>ch</b> okolade	e	2 6			1		2	3	1 1	1	1

			[ŗ]*	[l <sup>j</sup> ]*	[j]*	[ð]*	[-]*	[1]*	[ð]*	[t]*	[d]*
stride		5	1		1	1	1				
ma <b>d</b>	ăv	6		1			1	1			
he <b>dd</b> et	Ōż	7				1			1		
vasket		3						1		4	1

 Table 49. Targets [µ], [v], [c], [ðv] by the D-group

In the error systematization, I met a difficulty of systematizing accent features typical of the collected reading samples for the targets [v], [v],  $[\delta v]$  and [c], since their pronunciation (see **Tables 48, 49**) by the subjects differed considerably from the phonological phenomena I discussed above in connection with other consonant segments. If we analyze accent features for these sounds from the point of view of Flege's SLM model, we can see that they all fall into the category of dissimilar sounds except for the alveolar-palatal [c], which is very similar to the Russian palatal [c:].

In accordance with this model, the Russian foreign accent would be less noticeable for [
[
[
[
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]], [
[
]],

What is more interesting about [v], [v],  $[\delta v]$  in the D-group is that opposite to my prediction about the pronuonciation of the lateral [1]\* instead of  $[\delta v]$ , only 3 variants of  $[\delta v]$  as either  $[1^j]^*$  or [1]\* were given by the D-subjects out of a total of 15 wrong variants of  $[\delta v]$  in *stride, mad, heddet, vasket.* No clear foreign accent regularity can be drawn from the data analysis except for a dialectal pronunciation of the final  $[\delta v]$  in *vasket* as either [t]\*or [d]\*.

My prediction about a more consonant-like pronunciation of [v] could neither be verified (see **Tables 48, 49**), but is sooner falsified since the main foreign accent feature for [v] was its omission rather than [v]-like production in *kirke, mor* and *fersken*. On the contrary a less probable replacement by the hardly similar Russian thrilling  $[r]^*$  (6) or  $[R]^*$  (1) proved to be a characteristic of [v]-pronunciaiton in the D-group in *rigdom, ris, beredskab, irokeser*.

The pronunciation of [c] compared to those of [v], [v] and  $[\Delta v]$  was not native-like in both groups. Since this segment is very similar to the Russian /c:/, it was mispronounced as

<sup>&</sup>lt;sup>22</sup> If we consider that  $[\mu]$  (fricative) and  $[\mu]$  (approximant) are both uvular segments this would decrease the number of mispronounced variants to 5 out of 36.

 $[c]^*(7), [s]^*(5), [f]^*(4), [sj]^*(2)$  or  $[tf]^*(1)$ . The latter two were assumedly produced as the result of spelling interference in *sjov* and *chokolade*, while  $[c]^*, [s]^*[f]^*$  were given according to the equivalence classification.

Thus, due to the error systematizations done separately for the D- and R-groups I could trace different stages of the foreign accent in Danish learners for sounds  $[
[y], [
[y]] and [
[<math>\check{Q}^{v}$ ]. It is possible to draw a conclusion that more frequent mispronunciations of dissimilar [[
[y]], [[
[y]] and [ $\check{Q}^{v}$ ] by the D-participants advocate for their less advanced level of the Danish pronunciation for these three sounds, as Fledge's SLM model would account for that. However, the latter is not claimed but only suggested and needs to be tested in other tasks with more [[y]], [[y]] and [ $\check{Q}^{v}$ ]-targets set in a more natural environment than isolated words.

#### **5.3.4.** Word stress features

#### 5.3.4.1. Double primary word stress

From the error systematization below, we can easily see that the ignorance of the double primary stress in Danish was a major characteristic in both groups. See **Tables 50, 51**. In *direkte* and *allerede* the double primary stress was one of the options recommended as a standard variant along with the the combination of the main stress on the first syllable and the secondary stress (I included these words as targets for the double primary stress). However, the overwhelming majority of subjects produced none of the options correctly. Target words *femogtredive* and *julefest* were pronounced with the double primary stress by a total of 4 subjects and 6 subjects respectively.

direkte	[ˈdɨˈʁağd̥ə] or	1	di'rekte* (8)
	[ˌˈdiˈka͡ɕdə]	0	
allerede	[ˌåjəˌřeː፬̯ʎ ə] ou	0	alle'rede* (7)
	[ˌå]əˈƙe:Q̃^ ə]	2	
femogtredive	[ˈfemʌ̯ˈd͡strað̪ˠvə]	4	femog'tredive* (4), 'femog tredive* (1)
julefest	[ˈjuləˈfesd]	3	'jule fest* (4), 'julefest* (1), jule fest* (1)
bagefter	[ˈbɛ²jˈefd̥ʌ] or	2	bag'efter* (3)
	[ˈbɛ²jˌefd̯ʌ]	4	

Table 50. Double primary word stress in the R-group

direkte	[ˈd̪iˈkagd̥ə] or	0	di'rekte* (8), _di'rekte* (1),
	[ˈdiˈƙagd͡ə]	0	
allerede	[ˌåjəˌře:ỹ̯ł ə] ou	1	
	[,åjə'ře:Q <sub>Å</sub> 9]	0	'allerede* (3), alle'rede* (5)
femogtredive	[ˈfemʌᢩˈd͡strað̪ᢩvə]	0	femog'tredive* (8), femogtre'dive* (1)
julefest	['julə' fesd]	3	jule fest* (6)
-			
bagefter	['bɛ²j'efdʌ] or	2	,bag'efter* (5)
	[ˈbɛ²jˌefd̥ʌ]	2	

Table 51. Double primary secondary stress in the D-group

Such a poor performance can be ascribed to the absence in Russian of the double primary stress. A more Russian-like stress pattern ta- ta-(ta) for non-compound words was a major guideline in the realizations of *direkte* and *allerede* resulting in di'rekte\* (16) alle'rede<sup>\*23</sup> (12) with only one primary stress or with a correct primary stress on the first syllable of 'allerede\* (3), but omission of the secondary/second primary stress on the second syllable. The same was true of *bagefter*, which was produced as bag'efter\* (8) according to the Russian pattern *ta-* '*ta-*(*ta*).

Femogtredive and julefest were exposed to a different non-native like word stressassignment, namely the combination of a secondary and a primary stress or vice versa, thus the most typical variants were femog' tredive\* (12), and 'iule fest\*  $(10)^{24}$ . The subjects processed these words as compounds having one primary and one secondary stress, which was quite predictable because of their morphological properties.

### 5.3.4.2. Secondary stress in compound and non-compound words

Even though both Russian and Danish are stress-timed languages their stress patterns does not always work in the same way. As was mentioned in the comparative study the occurrence of secondary stress is one of the major distinctions between the two languages. The secondary stress in Russian is typical of long words, having the main stress more than three syllables away from the beginning of the word, not necessarily compounds. Thus, in

<sup>&</sup>lt;sup>23</sup> The neglected secondary stress here can be aslo heard sometimes in Danish native speakers. It was however classified as \*, since in Russian native speakers this would not sound native-like because of the syllable dynamics, which was not in the focus of the current study. <sup>24</sup> See footnote 23.

Russian the secondary stress will be inherent to the beginning of the word, i.e. the first or the second stem/syllable (Avanesov, 1956). In Danish, especially in compound words the secondary stress is usually assigned to the stem/syllable(s) following the first syllable (Heger 1992), as is it the case in the target words *ytringsfrihed, andetsprogspædagogik, barnevogn,* and non-compound *sårbar* and *barndom*.

The data analysis and error systematization showed (see **Tables 52, 53** below) that in the compounds with more than two stems Russian native speakers tend to "save" the primary stress for the last stem in the word as was the case with *ytringsfrihed* and *andetsprogspædagogik*, while setting the secondary stress on the first stem(s). Another typical feature of the Russian accent that should be highlighted in connection with the secondary word stress is a mere ignoring of the latter, as was the case with *barndom* and *eventyret*.

eventyret	['eːvənˌd͡syɐ̯ˀəð̪ˠ]	1	even'tyret (8)
ytringsfrihed	[ˈyd̥keŋsˈtkipəð̪ʎ]	3	yt'rings fri hed* (2), ytrings frihed* (3)
			yt rings fri hed* (1)
andetsprogs-	['ạnəð̆ <sup>v</sup> ,sþ¥osb̥ʰed̥ḁġo,ˈġiġ]	1	'andet'sprogspæda'gogisk* (2),
pædagogik			andet sprogspæda gogik* (3)
			'andet sprogspæda gogik* (1),
			andet sprogspædago gik* (1),
barnevogn	[ˈb̥äːnəˌvə̯wn]		'andet sprogspæda go gik* (1)
		9	
sårbar	[ˈsɔ̣ːˌb̥ä]	9	
barndom	['b៉äːn'd̈́v̈́w,	4	'barndom* (5) <sup>25</sup>

Table 52. Secondary stress in compound and non-compound words in the R-group

eventyret	[ˈeːvənˌd͡syɐ̯ˀəð̪ˠ]	1	even'tyret*(7), eventyr'et* (1)
ytringsfrihed	[ˌʌɡ̊keulsˈtkipəg̃ʌ]	3	yt'rings fri hed* (2), ytrings frihed* (3)
			yt rings fri hed* (1)
andetsprogs-	['ạnəð̆ <sup>v</sup> ,sþ¥osb̥ʰed̥ḁġo,ˈġiġ]	1	'andet'sprogspæda'gogisk* (2),
pædagogik			andet sprogspæda gogik* (3)
			'andet sprogspæda gogik* (1),
			andet sprogspædago gik* (1),
	[ˈb̥äːnəˌvə̯wn]		'andet sprogspæda 'go gik* (1)
barnevogn		9	
sårbar	[ˈsəːˌb̥ä]	9	
barndom	[ˈb̥äːnˌd̥ʌ̃m²]	3	'barndom* (6)

## Table 53. Secondary stress in compound and non-compound words in the D-group

<sup>&</sup>lt;sup>25</sup> See footnote 23.

## 5.3.4.3. Broken word stress in prefixed words

In the contrastive study, I put forward the idea that Danish prefixed words would be a particular challenge for Russian speakers. However, it was methodologically impossible to predict what particular features would characterize word stress in prefixed words.

According to the data and error systematization (see **Tables 54, 55**), a general tendency in the word stress assignment in prefixed words can be defined as *a broken word stress*. This phenomenon can be obviously traced as the result of

- re-distribution of the stress from the root to the prefix or vice versa, or the assignment:
  e.g., *u'held\**, *u'held\**, *u'kendt\**, *u'kendt\**, *'ukendt\**, *'umuligt\**;
- setting a secondary or a primary word stress on an unstressed prefix: e.g.
  'be arbejde\*, 'be 'arbejde\*, 'bearbejde\*, 'bear bejde\*, 'mistænksom\*, 'u muligt\*, 'u 'heldig\*, 'u heldig\*, 'u heldig\*.

uheldig	[u'hel'di]	7	u'heldig* (1), 'u heldig* (1)
uheld	['u hel <sup>?</sup> ]	4	u'held* (2), u'held* (2), 'uheld* (1)
mistanke	[ˈmisd͡säŋġə]	2	mis'tanke* (7)
mistænksom	[mis'dsen sim?]	9	
bearbejde	[b̥eʾāˈb̥äjd̥ə]	0	be arbejde* (2), be arbejde* (1), be arbejde* (4),
			'bearbejde* (1), 'bear bejde* (1)
ukendt	['u,ghend]	5	u'kendt* (3), 'u'kendt* (1),
umuligt	[uˈmu²lid]	9	

Table 54. Word stress in prefixed words in the R-group

uheldig	[u'hel <sup>?</sup> di]	5	u'heldig* (1), 'u heldig* (1), 'uheldig (2)
uheld	['u hel <sup>?</sup> ]	4	u'held* (2), 'uheld* (2), u'held* (1)
mistanke	[ˈmisd͡säŋg̊ə]	3	mis'tanke* (6)
mistænksom	[mis'dsen sym?]	6	'mistænksom* (2), 'mis tænksom* (1)
bearbejde	[þęäˈþäjd̥ə]	0	<i>1 not valid,</i> be arbejde* (2), 'be arbejde* (1) bear bejde*
			(2) bear bejde* (1), bearbejde* (2)
ukendt	['u ghend]	3	u'kendt* (3), u'kendt* (2), 'ukendt* (1)
umuligt	[uˈmu²lid]	5	'umuligt* (2), 'u muligt* (2)
-	-		

Table 55.	Word	stress in	ı prefixed	words in	the D-group
-----------	------	-----------	------------	----------	-------------

Thus, I verified the hypothesis about the double primary stress, which is typically either ignored or set in a word with two primary stresses, as if there were a secondary and main stress in this word. I also examined whether the secondary word stress is preserved in non-compound Danish words and may conclude that it was ignored by more than 50% of subjects

<sup>&</sup>lt;sup>26</sup> See footnote 23.

in the study. As for the secondary stress in compound words, it is very likely to be moved to the last stem of a compound as the data showed. Finally, I narrowed my predictions about the word stress assinement in prefixed words to the phenomenon of *a broken word stress*. It should be noted, that I studied the features of word stress assignment in Danish by Russian native speakers only typical of a word pronunciation in an isolated position or a focus word in an utterance. As the error systematization showed, the described features were traceable in both groups of subjects. Further analysis can address the word stress assignment in word combinations and sentences, and the obtained results may serve as a basis for other studies of the Russian accent on suprasegmental levels.

## 6. Global accent rating

For the global accent assessment, for the methodological reasons described in section 2.3, I also chose a reading task. I applied the so-called paragraph-reading technique, according to Piske et al. (2001: 193) widely used in other accent rating experiments (e.g., Oyama, 1976; Neufeld, 1979, 1980; Tahta et al., 1981; Piper & Cansin, 1988; Thompson, 1991; Bongaerts et al., 1995; Moyer, 1999).

As was mentioned earlier in section 4.1.3, all 27 subjects were offered to read aloud a small passage from *"Skolegade 4"* (Sandal, 2005: 79) – see **Figure 6** – after they have read the WL1 and before reading the WL2. To be exact, I recorded 12 reading samples by subjects from the D-group and 15 samples in the R-group.

The above-resented D-vs.-R-taxonomy was an underlying characteristic of the whole experiment, which was aimed at investigating whether SIPT plays any accent-mitigating role in the case of Russian learners of Danish as either a second or a foreign language.

It should be noted that in the case of T-readings, the title was omitted by some participants, but this fact is considered to be a minor issue, since this would not affect the accent degree ratings.

#### Regler

"Da vi havde været i Japan i omkring et halv år, mødte jeg en hollandsk ingeniør der havde boet i Japan i mange år. Jeg fortalte ham om vores middagsinvitation. Han lyttede og kunne fortælle mig følgende: i Japan spiser man op. Hvis man har taget noget op på sin tallerken må man spise det.

Man kan ikke give gaver der ikke er pakket ind. En cd er ikke en passende gave. Det er f. eks. blomster eller chokolade.

Man går ikke ud i folks køkkener. Det er privat område, lige som soveværelser ofte er det i Danmark. Hvis det er sent og man bliver tilbudt en drink, siger man pænt nej tak."

#### Figure 6. Text for the global accent assessment

## 6.1. Rating procedure and rating method

All 27 samples of the text were exposed to the global accent rating by native speakers of Danish. Methodologically, the global accent rating experiment was based on the assessment of the subjects' accent in accordance with a 5-point rating scale:

- 1- heavy accent;
- 2- considerable accent;
- *3- slight accent;*
- 4- almost native-like;
- 5- native-like.

Since from the methodological point of view it was impossible to predict prior to the rating results that any of the subjects would sound native-like and score the highest point for the accent degree, I had to record four native controls on an anonymous basis to make sure that native-like samples were included in the rating procedure. Native controls were three females and one male aged approximately 20-50.

The ratings were carried out by native raters for each T-sample. Technically speaking the accent assessment was carried out by means of filling in a special accent rating template (see **Appendix 16**). All the participants' personal numbers (PPNs) in the rating template were changed to numbers from 1 to 31 (27 samples and 4 controls), without using the previously introduced PPN-taxonomy (neither R vs. D, nor 2 vs. I (see section 4.2.2). The names of the audio files were also changed in the same way. Then all the "new" samples were shuffled in a way that would ensure an even representation of the D- and R-subjects at the beginning of the rating list as well as at the end of the latter. Thus, for instance, D5 was numbered as 10, D13 as 17, R4 as 19, etc. For the correspondences between the rating numbers and PPNs, see **Appendix 17**.

#### 6.2. Raters

Eight raters aged 17-65 years took part in the experiment: 4 expert raters (with a linguistic background) and 4 non-expert raters (without special linguistic background). I have previously explained in section 2.2., why I decided to choose volunteer raters from two groups. Five females and three males volunteered to be raters. Three of eight raters live currently in Aarhus area, while other 5 raters live in Greater Copenhagen.

#### 6.3. Results

**Figure 7** shows rating results for the global accent degree assessments of samples from the R-group (*PPNs 1.1. - 1.15 on the orange field*) and the D-group (*PPNs 2.1.-2.8., 2.10.-2.13 on the green field*). I used an averaging technique to find out which group received higher scores. The R-group as experiment showed demonstrated a better performance in the reading task with an average of 2.17, while the D-group's average was slightly lower, namely 1.8. on the above-mentioned 5-point scale.

Initially, I considered the groups to be methodologically equal, mainly because in the case of the R-participants a longer instruction and SIPT would compensate for the D-group's advantage of living in a language environment where Danish is a dominating language. Thus, by default the D-subjects would assumedly have more chances to put their language skills in language practice with native speakers, mitigate their foreign accent by a larger native input exposure and an active use of Danish outside language schools.

The analysis of the questionnaires filled in by all the participants showed that there is a very solid ground to conclude that the R-group's higher scores should be first and foremost ascribed to foreign accent mitigating role of SIPT, and these results are consistent with those obtained previously in the field of foreign accent studies. I have a series of arguments in favor of the fact that other factors, such as the length of instruction and the age of L2 acquisition did not play any significant role, at least under the conditions of this experiment.

Firstly, the percentage of a regular L1 use (which is considered to be a significant factor determining the degree of a foreign accent (Piske et al. 2001) and which has been recently set into focus by, for instance, Flege et al. (1997), (1999b), should have lead us to opposite results, since all but one R-subjects almost never use Danish outside the classroom and do not have conversations with native speakers on a regular basis - as a consequence their average foreign global accent degree ratings should have been lower compared to that of the D-subjects, but they are *not*. Thus, for example subject D6 with a stated percentage of Danish in everyday use at work equal to 80% received an average of 1.5., while R3 with a zero percentage of Danish use outside classroom and 90% of Russian use, had an average score of 2.6.

		Number in									
PPN	Age & Sex	rating									
11	10 M	sheet		ER2	ER3	ER4	N-ER 1	N-ER2	N-ER3	N-ER4	Average
1.1.	19 IVI 21 M	1		1	2	1	2 2	1	ے ا	1	1.5
1.2.	21 M 10 E	0		2	3 2	3	2	2	4	3	2.5
1.3.	19 F	3		3	3	2	4	2	3	2	2.6
1.4.	19 Г 10 Г	19		2	2	4	5	1	2	2	2.5
1.J.	19 F	J 12		2	4	3	4	1	2 2	2	2.5
1.0.	21 F 10 M	15		2	2	4	с С	2	3	1	2.5
1.7.	19 IVI 21 E	20		2 1	2 2	2 2	ے 1	3 1	3 1	3 1	2.3
1.0.	21 F	20		1	2	2	1	1	1	1	1.4
1.9.	20 M 22 F	23 A		1	2	נ ר	2	1	2	1	2.0
1.10.	22 F	+		1	2 1		3 2	2	3	2	2.0
1.11.	21 F	12		2	- -	- 1	2	2	2	3	2.0
1.12.	21 I 22 M	21	3	2 1	2	- 3	2	1	2	1	2.1 1.8
1.13.	22 F	8	1	2	23	3	2 4	2	2	2	2.4
1.15.	20 F	23	$\frac{1}{2}$	2	3	3 4	2	2	2	2	2.4
2.1.	31 M	31	1	1	2	1	3	1	2	1	1.5
2.2.	31 M	6	1	2	- 3	3	2	2	4	3	2.5
2.3.	27 F	24		- 1	2	3	- 1	- 1	2	1	1.5
2.4.	22 F	15	2	2	2	3	2	1	2	2	2.0
2.5	22 F	10	2	2	3	4	4	2	3	3	2.9
2.6	30 F	26	1	1	2	3	1	1	1	2	1.5
2.7	60 F	11	1	1	1	3	2	1	1	1	1.4
2.8	25 M	27	1	1	1	3	1	1	1	1	1.3
2.10	24 M	16	1	1	2	4	2	1	1	1	1.6
2.11	30 F	28	1	2	3	3	2	1	2	3	2.1
2.12	30 F	29	1	2	3	3	2	1	2	2	2.0
2.13	33 F	17	1	1	1	4	1	1	1	1	1.4
native con	trol 1	9	5	5	5	5	5	5	5	5	5.0
native con	trol 2	18	5	4	4	5	5	4	4	5	4.5
native con	trol 3	30	5	5	5	5	5	5	5	5	5.0
native con	trol 4	14	5	5	5	5	5	5	5	5	5.0

Figure 7. Results of the global accent degree ratings across all the 8 raters.

Abbreviations: PPN – participant's personal number; ER – expert rater; N-ER – non-expert rater.

Secondly, according to Piske et al. (2001) the length of instruction as one of the instructional "variables", was found by Flege & Fletcher (1992) to be a significant predictor

of the degree of L2 foreign accent, but accounted for only 5% of the variance in the foreign accent ratings obtained in experiment (Flege & Fletcher, 1992) with Spanish learners of English. Moreover, in some studies according to Piske et al. 2001: 200) "...a total amount of formal classroom training, was found to be inversely related to the L2 pronunciation accuracy". This means that a longer instruction in the case of the R-subjects (average of 31 months) compared to that of the D-subjects (average of 25 months) would hardly account for a higher average score in the accent degree rating. Vivid examples in support of a very inconsiderable role of the length of instruction – 17 months); and R13 (average score – 2.5.; length of instruction – 17 months); and R13 (average score – 2.0.; length of instruction – 30 months); and R7 (average score – 2.5.; length of instruction – 18 months).

Thirdly, according to the questionnaires all the subjects started learning Danish as adults after the age of 17-18 or even later as D7, who started learning Danish at the age of 58, (her scores however are none the worse than those of D13 who started at the age of 32), and consequently can be all regarded as late learners, thus minimizing to zero an early exposure to Danish as a decisive and advantageous factor.

Finally, the results obtained in the case-study on the global accent degree are consistent with other studies supporting a crucial role of an intensive phonetic training for late foreign and second language learners as an accent-mitigating factor. Thus, in Bongaerts et al. (1997)'s study the ratings received by late learners of English were comparable to those obtained for native speakers of English and according to Bongaerts et al. (1997), the latter was due to an intensive phonetic training in the perception and production of English sounds.

Another study carried out by Moyer (1999) provides evidence supporting a positive effect of the suprasegmental and segmental training for native English learners of German. "She found that those subjects who had received both, obtained ratings that were closer to the range of ratings obtained for native speakers of German" (Piske et al. 2001: 200).

In my case study, the R-subjects received, according to the information reported by professors of the Institute of Scandinavian Languages, Dutch and Finnish at Moscow State Linguistic University a special introductory phonetic training in both segmental and prosodic aspects of the Danish pronunciation prior to the main course of the Danish language studies. According to the questionnaires, this course lasted for 4 months with 10 academic hours of instruction per week, including phonetic training in the language laboratories of

approximately 3 academic hours per week. In other words, SIPT was of a quite intensive character. Relying on Missaglia (1999)'s study, a prosody-oriented phonetic training is more effective in improving pronunciation than a segment-oriented training. Therefore, I should admit that the R-subjects, who received a special training in both aspects, should have probably received higher scores.

Interestingly, according to the few comments given by the volunteer raters (only expert raters gave comments) on the R-subjects' elicitations in most cases the foreign accent manifested itself most vividly through either prosodic characteristics, such as intonation and phrasal stress, or segments, and only in fewer cases through both. For example:

- for R10: "Lydene er rigtig gode, men trykfordelingen og intonationen driller".
- for R8: "Lydene er meget skæve, men trykfordelingen er faktisk OK".
- for R13: "Rigtig god intonation, men lydene forstyrrer lidt.".
- for R1: "Udtalen er god, men selve intonationen er særligt afslørende".'
- for R11: "Intonationen er næsten perfekt. Accenten er tydeligst i udtalen af "o" og "cd". God intonation."

Such a tendency can be probably explained by the fact that even in advanced learners of Danish, automatization of pronunciation simultaneously on both segmental and susprasegmnetal levels, comes into force much later, if it does at all.

The comments on the D-subjects' reading samples addressed mainly the accent on the segmental level. Only two subjects received comments concerning their intonation.

- for D1: "Lydene er simpelthen for uklare."
- for D3: "Accenten afsløres særligt i udtalen af "havde"."
- for D4: "*Trykfordeling og intonation er gode og ville have givet 5 hvis ikke nogle af lydene forstyrrede. Endelserne bliver "slugt" ved få af ordene.*"
- for D6: "Intonation og trykfordeling forstyrrer. Derimod gør det faktisk ikke noget at hendes r'er er triller."
- for D7: "Svært ved at sige 'h' og 'ø'."
- for D8: "H'erne er ret forstyrrende. Speciel udtale af 'h'."
- for D10: "Trykfordeling og intonation er gode, men lydene er svære at kende. Speciel udtale af 'lyttede og 'følgende'."
- for D13: "Mange forstyrrende lyde."

The latter can be assumedly ascribed to the D-subjects' larger exposure to native input and therefore a better performance in terms of prosody. The latter however is only an assumption and further research is needed to test this idea.

Among the most common remarks, concerning segments in both R-and D-samples were those related to the articulation of /h/. This tendency goes at first sight somewhat against my results of the study on the typical segmental features of Russian accent. I argue that a better performance of subjects in the production of target [h]-sounds in the reading task with the WL1 was due to the target word's isolated position when subjects did not have to be so concentrated on the suprasegmnetal level features of their pronunciation compared to the text reading.

All I all, I believe that in general neither of groups demonstrated an outstanding performance in the T-task, according to the scores assigned to the participants. However, two participants – R11 and D5 – have been assigned scores almost equal to 3 points, which correspond to "a slight accent" on the five-point scale. I argue that in the case of R11 this was probably due to R11's three-week stay in Denmark for the purpose of summer language course, when R11 had more exposure to a native input than other subjects from the R-group as well as due to the fact that R11, compared to other R-subjects have had approximately 5.5. hours of conversation with native speakers per month. As far as D5 is concerned, her higher than others' average scores can be assigned to the fact that, according to her questionnaire, she has had Danish as a language of instruction on a graduate-level for the last 6 months.

I have thus empirically, verified my hypothesis that SIPT plays an accent-mitigating role and argue that it would considerably improve an accent in late learners of Danish as a second language, if it is introduced before the first modules of the Danish language course (see **Appendix 24** for the module system applied in the field of the Danish language course for adult and young learners of Danish as a second language according to the Common European Language Framework). The latter is relevant at least in the case of later learners with Russian as L1. Further research is needed to find out SIPT's role in the global accent improvement in late learners of Danish with other L1s.

## **Final conclusions and discussions**

The goal of this thesis was to answer two questions. The first question was the following:

# 1) What are the most typical foreign accent features in Russian native speakers with Danish as a foreign and second language on the segmental level and in the word stress assignment?

To answer the first question I used one of the methodological principles of accent studies most widely applied during the recent decades (Flege, 2002; Best et al. 2001; Flege et al., 1995; Flege 1981a; Ingram & Park, 1998; McAllister et al., 2002; Missaglia, 1999). I proceeded from more abstract predictions about how dissimilarities and similarities between Danish and Russian phonemic inventories can be reflected in the Russian accent. These predictions were formulated as the result of the contrastive analysis of the Danish and Russian phonemic inventories according to the theory of the distinctive phonological features, as well as the comparison of the distinctive aspects of word stress assignment in two languages.

All the hypotheses about the features of the Russian natives' pronunciation in Danish were tested empirically by means of the case-study. The case study included: 1) data collection in the form of recordings of the two word lists with target segments and word stress patterns read by 18 subjects; 2) error systematization and error analysis.

The error analysis was mainly based on two major theoretical assumptions. The first one was, according to Flege's theory of equivalence classification that Russian native speakers would tend to classify Danish sounds, which are similar to the Russian ones in terms of articulation, according to the categories and articulation properties of the corresponding Russian similar sounds. The second assumption was that the typical errors in pronunciation would be also the result of the spelling interference, "whereby the spelling of the word…" in L2 "…triggers a correspondence between…" an L2 spelling symbol "…and the pronunciation of the same symbol in the native language…" (Miglio & Fukazawa 2006: 4145). Both of these assumptions were empirically substantiated and the following conclusions can be made on the typical features of the Russian natives' foreign accent in Danish on the segmental level and on the level of word stress assignment.

The Russian natives tend to qualitatively reduce Danish vowel qualities [a] to  $[v]^*$  or  $[a]^*$ ;  $[\varepsilon]$  to  $[a]^*$ ;  $[\ddot{a}]$  to  $[v]^*$ . All these Danish vowels tend to be qualitatively reduced 93

according to the principles of the Russian weakening of vowels in the unstressed position, because the Russian natives tend to classify them as similar Russian vowel qualities. Remarkably, as it may seem, in some dissimilar vowels, the reduction was realized in compliance with the Danish principles of vowel weakening, as for instance, in the case of [y] to a front-mid allophone  $[y]^*$ .

The Russian natives' pronunciation in Danish is characterized by fewer quality distinctive properties of the back vowels. Those Danish vowel segments which are open-mid and close-mid – [q] and the advanced [q] respectively – tend to be mispronounced by the Russian native speakers as  $[u]^*$  or more rarely as allophones of the Russian /o/, the latter is more typical of [q]. Thus, Russian learners tend to resort to the usage of the familiar sound inventory, namely [u] and [o] (according to the equivalence classification or as the result of spelling interference of the letters u and o), with a worse distinction of other back vowel qualities subject to narrowing. Danish words where vowel allophones are represented in spelling, by means of the letters also found in the Russian language (that *may* be or *are* used in the latter to represent a different vowel quality) mispronunciation will have a greater probability.

The Russian natives tend to shorten the Danish long vowels. However, the latter depends on the type of the instruction the learners receive, in the current study those subjects who had a special phonetic training with the focus on the distinction between long and short vowels performed better in the reading task for the long targets.

The non-syllabic elements [w] and [y] in Danish diphthongs can be consonantized as  $[v]^*$  and  $[r]^*/[y]^*$  respectively. Danish diphthongs can be mispronounced as monophthongs, especially [y]-diphthongs.

The Russian native speakers tend not to aspirate segments  $[b^h]$  and  $[b^h]$ . The aspirated [ds] is either disaspirated as  $[t]^*$ ,  $[t]^*$ ,  $[t]^*$  or  $[t]^*$ , or is replaced by the similar Russian affricate  $[ts]^*$ . The voiceless segments [b], [d] and [b] are voiced after a vowel; aslo [s] in an intervocalic position. The Russian natives tend to dentalize /d/, /s/, /t/ /n/.

The double primary stress is typically either ignored by the Russian natives or set in a word with two primary stresses, as if there were a secondary and main stress in this word. The secondary word stress is often ignored in non-compound words, and in compounds with more than two stems, the Russian native speakers tend to "save" the primary stress for the last stem in the word.

Methodologically speaking, these are the features, which I put forward as the result of the contrastive study. Thus, in general, the case study provided evidence in support of most of the predictions. The case study gave also grounds to figure out other typical features, which had not been predicted in the contrastive study. The latter gives credit to the error analysis made, in terms of the methodological value. Therefore, I have to *supplement* the features presented above with the following ones.

Russian advanced late learners of Danish in general tend to preserve the Danish vowel quality better in the stressed positions. However, the equivalence classification is also typical of the stressed vowels. The current study provided evidence, that the equivalence classification may be regarded as a two-sided phenomenon and therefore, can empirically extend Flege's idea by saying that not only *do* advanced L2 learners demonstrate a worse performance for the L2 sound (A), more similar to the corresponding L1 sound (B), they may also have a tendency to produce A instead of B, as is the case with the Danish [i] vs. [e] distinction in Russian learners. Additionally, the front labialized [ $\emptyset$ ] is often mispronounced as [y], especially under the influence of the spelling interference. Sounds [ $\emptyset$ ] and [ $\emptyset$ ] are generically susceptible to narrowing in terms of height to such qualities as [ $\emptyset$ ]\*, [e]\*, [Y]\* and [ $\emptyset$ ]\* as well as a tongue retraction. The nucleus of the diphthong may be exposed to the same qualitative errors as the corresponding vowel quality. The error analysis also narrowed my predictions about the word stress assignment in prefixed words to the phenomenon of *a* broken word stress.

The error analysis, as the last step of the case study, also provided evidence that *falsifies* some of my hypotheses, or extends some of them. Thus, one of my hypotheses was that the Russians would palatalize /b/, /d/, /g/ especially before Danish front /i/, / $\epsilon$ /, /y/, /e/. The error analysis showed that this assumption was right for /g/, but I should admit that also sonorant segments, voiced [v] and voiceless [h] may be exposed to palatalization. I also have evidence in support of the palatalization *after* a front vowel and before [ $\rho$ ]\*. However, I cannot conclude that the palatalization manifested itself as a primary property of the Russian accent.

In my contrastive study on the vowel inventories, I pointed out the phonemes /a/ and / $\epsilon$ / and their allophones as major eventual targets for weakening in unstressed positions. The error analysis showed however, that this hypothesis could have been extended over other

vowel front and back qualities. Thus, front [e] in the pre-tonic syllable was usually reduced to [I]\*, just as the Russian [i] would be in the same position; [p] was exposed to [v]\*-weakening – typical of the Russian /u/ in the pre-tonic position.

The results of the error analysis also falsified my hypothesis about lengthening of the short vowels and about the velarization of /l/. Moreover, my prediction about a more consonant-like pronunciation of segment [v] could neither be verified, but is sooner falsified since the main accent feature for [v] was its omission rather than [v]-like production.

As far as the reading task (the word lists readings) is concerned, it is possible to conclude that methodologically it met the goals of the thesis in terms of the segmental aspects of the Russian natives' pronunciation in Danish as well as the word stress assignment. However, it is not possible to exclude the fact that a spontaneous speech task or, for example an elicitation of the target sentences or words after a native speaker would highlight other features of the pronunciation in the target groups. Moreover, the reading task, as was already mentioned was not illustrative for the segment [ə], which should be rather studied in the framework of either a spontaneous speech production or reading tasks other than with isolated words.

It should be noted that the whole thesis was not aimed at doing a quantitative research on the statistical significance of the mentioned features and was of a qualitative, descriptive and phonetic character. Nor should it have assessed the consequences of these or those typical segmental and word stress features from the point of the language use. However, I believe that the set of the features typical of the Russian accent described and systematized in this thesis have a significant practical importance for the speakers of Danish as a foreign or second language with Russian as L1 and teachers of Danish. It may serve a basis or a guideline for the development of study materials for the Danish pronunciation on the segmental level, as well as help the Russian natives to tackle their weak points in pronunciation.

Even though I have argued in the thesis that the D-group (without SIPT) and the Rgroup (with SIPT) sometimes performed differently for the same target sounds, mainly because of the different phonetic instruction, the described features can be applied to both groups: learners who have studied Danish as a foreign language and a second language.

And finally, on the basis of the results of the thesis I can supplement the answer to the first question by saying that the discovered features provide evidence in support of Flege's

Speech Learning Model, because the number of cases of mispronunciation of the Danish segments which are mostly, but not totally similar to the Russian sounds, were more considerable than those of the dissimilar sounds.

The second question this thesis was to give an answer to was the following:

# 2) Can a special introductory phonetic training anticipating the main language course mitigate the degree of a global foreign accent in late native Russian learners of Danish compared to the role of the phonetic training integrated in the main course of studies?

As was stated in section 6.3., the results of the global accent ratings showed that the SIPT anticipating the main language course *does* mitigate the degree of a global foreign accent in late native Russian learners of Danish. The difference in the mean (the D-group average of 1.80, and the R-group average of 2.17) ratings in the two target groups is statistically significant. Figure 8 shows the distribution of rating scores for the D-group (the red shape) and the R-group (the blue shape). The higher the overlap between the shapes is, the higher is the probability that the difference in the score distributions was random. This graphical representation was substantiated by a T-test, which assesses whether the means of the two target groups are statistically different from each other. See the calculation results in Appendix 23. The probability of the null hypothesis that SIPT does not play a foreign-accent mitigating role was estimated at 0.053 (Figure 9). This means, that the probability that the difference in mean scores obtained by the D- and R-groups (considering the conditions of the case study and the linguistic portraits of the subjects, and the fact that such factors as learning strategies and level of education were not taken into account) was random is 5.3%, which is statistically very low. In other words, the probability that SIPT plays an accent-mitigating role is 94.7% (considering the conditions of the case study), which is statistically very significant.



Figure 8. The distributions of scores for the D-group (in red) and the R-group (in blue).

#### 2/14/12

Student's t-test: Results

# Student's t-Test: Results

The results of an unpaired t-test performed at 11:34 on 14-FEB-2012

t= 2.04 sdev= 0.463 degrees of freedom = 25

The probability of this result, assuming the null hypothesis, is 0.053

## Figure 9. The results of the t-Test

Generally speaking, I can conclude that SIPT as an initially advantageous factor in the R-group should have been supplemented with more practice of the Danish language use in conversations with native speakers as well as more exposure to native input through mass media, for instance, and other listening activities outside the classroom. The latter two, according to the questionnaires were R-group's weak points, presumably, as the consequence of studying Danish outside Denmark, and as the result or having fewer chances of having conversations with native speakers in a non-Danish language environment. The D-subjects who on the contrary have a more advantageous situation in terms of exposure to the native input, and have received pronunciation teaching as a part of everyday classroom activities,

but have not had SIPT could have demonstrated a better performance than the R-subjects if they had SIPT prior to the main language practice course.

No research is available today on whether a special phonetic training should precede the main language course or be integrated into it parallel to the main language course, but I argue that the best results would be obtained if SIPT took place before the main language course of Danish as a second language, since in that case the learning of new pronunciation patterns and the development of new articulation habits would anticipate the interference of the Russian language in the most effective way, and learners would be made aware and conscious of the differences between two phonological systems. The latter would create a solid basis for the automatization of new pronunciation habits and articulation in various language activities in the classroom and outside it.

## References

Anderson-Hsieh, Janet. (1988). The Effect of Foreign Accent and Speaking Rate on Native Speaker Comprehension. *Language Learning*, 38 (4), 561-613.

Anderson-Hsieh, Janet; Koehler, Kenneth; Johnson, Ruth. (1992). The Relationship Between Native Speaker Judgments of Nonnative Pronunciation and Deviance in Segmentais, Prosody, and Syllable Structure. *Language Learning*, 42 (4), 529-555.

Aoyama, K., Guion, Flege, J. E., Guion, S.G., Akahane-Yamada R., Yamada, T. (2004). Perceived phonetic dissimilarity and L2 speech learning: the case of Japanese /r/ and English /l/ and /r/. *Journal of Phonetics*, 32, 233–250.

Archibald J. (1998). *Second language phonology*. University of Calgary. Amsterdam/Phladelphia: John Benjamins Publishing Company.

Avanesov, R. I. (1956). Fonetika sovremennogo russkogo literaturnogo jazyka. Moskva.

Baker, W., Trofimovich, P., Mack, M., & Flege, J. (2002). "The effect of perceived phonetic similarity on non-native sound learning by children and adults". In Skarabela, B., Fish, S. & Do, A. (eds.), *Proceedings of the 26th Annual Boston Univ. Conference on Lang. Development*. Somerville, MA: Cascadilla Press, 36-47.

Basbøll, Hans. (2005). The phonology of Danish. Oxford: Oxford University Press.

Best, C. T. (1995). "A direct realist view of cross-language speech perception". In W. Strange (eds.), *Speech perception and linguistic experience: Issues in cross-language research*. Timonium, MD: York Press. 171-204.

Best, C. T., McRoberts, G. W., & Goodell, E. (2001). Discrimination of non-native consonant contrasts varying in perceptual assimilation to the listener's native phonological system. *Journal of the Acoustical Society of America*, 109 (2), 775-794.

Bohn, Ocke-Schwen Munro, Murray J. (2007). *Language Experience in Second Language Speech Learning: In Honor of James Emil Flege*. Amsterdam: John Benjamins Publishing Company.

Bondarko L.V. (1977). Zvukovoi stroi sovremennogo russkogo iazyka. Moskva: Prosveshchenie

Bongaerts, T., Planken B. & Schils, E. (1995). "Can late learners attain a native accent in a foreign language? A test of the critical period hypothesis". In D. Singleton & Z. Lengyel (eds.), *The age factor in second language acquisition*. Clevedon, U.K.: Multilingual Matters. 30-50.

Bongaerts, T., van Summeren, C., Planken, B., & Schils, E. (1997). Age and ultimate attainment in the pronunciation of a foreign language. *Studies in Second Language Acquisition*, 19, 447-465.

Chomsky, Noam & Halle, Morris. (1968). *The sound pattern of English*. New York: Harper and Row. Reprinted 1991, Boston: MIT Press.

Crystal, David. 2003. A dictionary of linguistics & phonetics. Oxford: Basil Blackwell.

Cubberley, P. (2002). *Russian: A Linguistic Introduction* (1st ed.). Cambridge: Cambridge University Press.

Ehri, Linnea C. & Wilce, Lee S. (1980). The influence of orthography on readers' conceptualization of the phonemic structure of words. *Applied Psycholinguistics* 1, 371-385. Elliott, R. E. (1995) Field independence/dependence, hemispheric specialization, and attitude in relation to pronunciation accuracy in Spanish as a foreign language. *The Modern Language Journal*, 79, 356-371.

Encyclopedia of Language & Linguistics (Second Edition). (2006). Editor-in-Chief: Keith Brown. Elsevier Ltd.

Fischer, I. S. (1992). Fonetik og udtaleundervisning. Hinnerup: Fischers Forlag.

Fischer-Jørgensen, Eli. (2001). *Tryk i ældre dansk: sammensætninger og afledninger*. Det Kongelige Danske Videnskabernes Selskab: (Viborg).

Flege J. E. & MacKay. (2011). "What accounts for "age" effects on overall degree of foreign accent?" In M. Wrembel, M. Kul and Dziubalska-Kołaczyk, K. (eds.), *Achievements and perspectives in the acquisition of second language speech: New Sounds*. Vol. 2, Bern, Switzerland: Peter Lang. 65-82.

Flege, J. & MacKay, I. (2004). Perceiving vowels in a second language. *Studies in Second Language Acquisition*, 26, 1-34.

Flege, J. E. (1988). Factors affecting degree of perceived foreign accent in English sentences. *Journal of the Acoustical Society of America*, 84, 70-79.

Flege, J. E. (1995). "Second language speech learning: Theory, findings, and problems". In Strange, W. (eds.), *Speech perception and linguistic experience: Issues in cross-language research*. Timonium, MD: York Press. 233-277.

Flege, J. E. (2002). "Interactions between the native and second-language phonetic systems". In P. Burmeister, T. Piske, & A. Rohde (eds.), *An integrated view of language development: Papers in honor of Henning Wode*. Trier. 217-244.

Flege, J. E., Yeni-Komshian, G. & Liu, H. (1999b). Age constraints on second language acquisition. *Journal of Memory & Language*, 41, 78-104.

Flege, J., E. (1981a). Cross-Language Phonetic Interference: Arabic to English. *Language* and Speech, 24 (2), 125-146.

Flege, J., E. (1981b). The Phonological Basis of Foreign Accent: A Hypothesis. *TESOL*, 4 (15), 443-455.

Flege, J., E. (1987). A Critical Period for Learning to Pronounce Foreign Languages? *Applied Linguistics*, 8 (2), 162-177.

Flege, J., Schirru, C. & MacKay, I. (2003). Interaction between the native and second language phonetic subsystems. *Speech Comm.*, 40, 467-491.

Flege, J., Takagi, N., & Mann, V. (1995). Japanese adults can learn to produce English /r/ and /l/ accurately. *Language & Speech*, 38, 25-55.

Grønnum, Nina. (2001). Fonetik og fonologi: almen og dansk. Kbh.: Akademisk (Nørhaven, Viborg).

Grønnum, Nina. (2005). Fonetik og fonologi: almen og dansk. Kbh.: Akademisk (Nørhaven, Viborg).

Grosjean, F. (1992). "Another view of biblingualism". In Harris, R. (eds.). Cognitive Precessing in Bilinguals. Amsterdam: North-Holland. 51-62.

Hall, T. Alan. (Editor). (2001). *Distinctive Feature Theory* (Phonology and Phonetics). Berlin/New York: Mouton de Gruyeter.

Hall, T.A. (2006). Syllable: Phonology. In Keith Brown (eds.) *Encyclopedia of Language & Linguistics* (Second Edition). 329-333.

Handbook of the International Phonetic Association. A guide to use the International Phonetic Alphabet. (1999). Cambridge: Cambridge University Press.

Henrichsen, P. J. (2009). The CBS Text-to-Speech Workbench Copenhagen Business School. Hickey, R. (1986). "The status of diphthongs in Irish and Russian". *Zeitschrift für. Phonetik, Sprachwissenschaft und Kommunikationsforschung* (39), 97-105.

Ingram, J. C., & Park, S.G. (1998). Language, context, and speaker effects in the identification and discrimination of English /r/ and /l/ by Japanese and Korean listeners. *Journal of the Acoustical Society of America*, 103, 1161-1174.

Jakobson, R., Gunnar C., Fant, M., Halle, M. (1954). *Preliminaries to speech analysis*. MIT Press, Cambridge, MA .

Jakobson, R."The concept of phoneme". Reprinted in L.R. Waugh, M. Monville-Burston (eds.), *On language* (1990), Harvard University Press, Cambridge, MA (1942). 218-241.

Jones D., Ward D. (1969). *The Phonetics of Russian*. Cambridge: Cambridge University Press.

Kirk, K. (2008). *Dansk udtale: en undervisningsvejledning* /redaktion: Gitte Østergaard Nielsen ... [et al.]; Kbh.: Ministeriet for Flygtninge, Indvandrere og Integration, Albertslund.

Ladefoged, P. (1971). *Preliminaries to Linguistic Phonetics*. Chicago: Chicago University Press.

Ladefoged, Peter (2006). A Course in Phonetics (Fifth Edition), Boston, MA: Thomson Wadsworth.

Lado, R. (1957). Linguistics across cultures. Ann Arbor: University of Michigan Press. *Lang.*& *Speech* 24, 25-146.

Lagerberg, R. (2007). Variation and Frequency in Russian Word Stress. *Australian Slavonic and East European Studies*, 21 (1–2), 165-176.

Lagerberg, R. J. (2005). Towards a comprehensive account of the stress of Russian verbs containing the suffix –ировать: a survey of Russian speakers', *Russian Linguistics*, 29, 39-47.

Lagerberg, R. J.: 2006, 'The stress of Russian adjectives containing the suffix -чатый,' Russian Linguistics, 30, 227–234.

Lenneberg, E.H. (1967). Biological Foundations of Language. ISBN 0898747007.

Lund, K. (2003). Alder og accent. Sprogforum, 26, 9-16.

Major, Roy Coleman. (2001). *Foreign accent: the ontogeny and phylogeny of second language phonology*. Mahwah, N.J.: Lawrence Erlbaum Associates.

Malmkjær, K. (2001). *The linguistics encyclopedia* /edited by Kirsten Malmkjær. London: Routledge.

McAllister, R., Flege, J., & Piske, T. (2002). The influence of the L1 on the acquisition of Swedish vowel quantity by native speakers of Spanish, English and Estonian. *Journal of Phonetics*, 30, 229-258.

Mielke J. & Hume, E. (2006). "Distinctive Features". In Keith Brown (eds.) *Encyclopedia of Language & Linguistics* (Second Edition). 723-731.

Miglio, V. & Fukazawa, H. (2006). "Spelling Influence on Pronunciation: Output-Output Correspondence and Interlanguage Phonology". *Proceedings from the 4<sup>th</sup> Hawaii International Conference on the Arts and Humanities*. 4143-4152.

Missaglia, F. (1999). Contrastive prosody in SLA - an empirical study with adult Italian learners of German, *Proceedings of the 14th International Congress of Phonetic Sciences*, Vol. 1. 551-554.

Moyer, A. (1999). Ultimate attainment in L2 phonology. *Studies in Second Language Acquisition*, 21, 81-108.

Munro, M., & Derwing, T. (1995). Foreign accent, comprehensibility, and intelligibility in the speech of second language learners. *Language Learning*, 4, 73-97.

Munro, M., Flege, J. & MacKay, I. (1996). The effect of age of second-language learning on the production of English vowels. *Appl. Psycholing*, 17, 313-334.

Neufeld, G. G. (1979). Towards a theory of language learning ability. *Language Learning*, 29, 227-241.

Oyama, S. (1976). A sensitive period for the acquisition of a nonnative phonological system. *Journal of Psycholinguistic Research*, 5, 261-283.

Patkowski, Mark. S. (1990). Age and Accent in a Second Language: A Reply to James Emil Flege. *Applied Linguistics*, 11 (1), 73-89.

Piper, T. & Cansin, D. (1988). Factors influencing the foreign accent. *The Canadian Modern Language Review*, 44, 334-342.

Piske, T., MacKay, I. R. A., & Flege, J. E. (2001). Factors affecting degree of foreign accent in an L2: a review. *Journal of Phonetics*, 29, 191–215.

Purcell, E. T. & Suter, R. W. (1980). Predictors of pronunciation accuracy: a reexamination, *Language Learning*, 30, 271-287.

Riney, T. J. & Flege, J. E. (1998) Changes over time in global foreign accent and liquid identifiability and accuracy. *Studies in Second Language Acquisition*, 20, 213-244.

Selinker, L. (1972). Interlanguage. *International Review of Applied Linguistics*, 10 (3), 209-231.

Selinker, L. (1996). *Fossilization: What We Think We Know*. London: Longman Group UK Limited.

Selinker, L. 1992. Rediscovering interlanguage. Harlow: Longman.

Sharapova, Marklund, E. (2000). *Implicit and Explicit Norm in Contemporary Russian Verbal Stress*, Uppsala: Uppsala University (Acta Universitatis Upsaliensis. Studia Slavica Upsaliensia).

Søgaard, H. (1999). *Min udtale*. [Bind] 1: Enkeltlyd, tryk, stød m.m. Kbh.: Gyldendal.

Tahta, S., Wood, M. & Loewenthal, K. (1981). Foreign accents: factors relating to transfer of accent from the first language to a second language. *Language & Speech*, 24, 26-272.

Thompson, I. (1991). Foreign accents revisited: the English pronunciation of Russian immigrants, *Language Learning*, 41, 177-204.

Ukiah, N. (2002). The stress of Russian nouns in -a and -я of Zaliznjak's pattern f (губа́ type)'. Australian Slavonic and East European Studies, 16/1–2, 1-39.

Whitman, R. (1970). Contrastive analysis: problems and procedures. *Language Learning*, 20, 29-41.

Wiese, R. Phonology: Overview. In Keith Brown (eds.) *Encyclopedia of Language & Linguistics* (Second Edition). 562-564.

Yermolovich, D.I. (2005). *Proper names: theory and practice of translation*. Moscow: R. Valent.

http://openarchive.cbs.dk/bitstream/handle/10398/7763/2009-1.pdf?sequence=1

http://	weston.ruter	.net/proi	ects/ipa-	chart/view/	kevboard/

http://ordnet.dk/ddo/

http://ordnet.dk/ddo/artiklernes-opbygning/udtale?set\_language=da

http://schwa.dk/filer/daniategnvaelger/

http://schwa.dk/lydskrift/dansk-lydskrift/

http://gyros.dk/usenet/dks/lydskrift\_oversigt.pdf

http://www.dansklektor.dk/arkiv/Lydskriftsystemer%20Volhardt.pdf http://en.wikipedia.org/wiki/International\_Phonetic\_Alphabet

# Appendices

## Appendix 1. Questionnaire for subjects

Participant's personal number ...

Age: Sex: F M Email:

1. When did you start learning Danish?

2. Did you learn/Are you learning (please, underline) Danish as a foreign (in Russia) /second (in Denmark) language?

As a for	reign language	As a sec	cond language
3.	How long have you been learning/did you	3.	How long have you been learning/did you
	learn Danish?		learn Danish?
4.	Have you ever lived/stayed in Denmark	4.	Did you use/Do you use Danish outside the
	during your studies of Danish?		language school?
5.	If yes, how long?		
6.	What was the purpose of your stay?	5.	If yes, how often?
7.	Where do you use Danish now?	6.	For what purposes?

- 7. How many hours per week of class teaching did you have/do you have on average?
- 8. Is/(Are) your teacher(s) a Danish native speaker (native speakers)?
- 9. How much time on average per month do you speak to native speakers?
- 10. Do you have any Danish next of kin?
- 11. How many hours per week do listen to native speakers, including TV, radio, music, online broadcastings, watching films? Please, specify the source!
- 12. Did you have a special phonetic training before you started the main course of the Danish language curriculum?
- 13. If yes, was it focused on: articulation of sounds/intonation/both?
- 14. If yes, how long did it take?
- 15. If yes, how often was it per week?
- 16. Was your Danish pronunciation training an integrated part of your curriculum when you started to learn Danish?
- 17. Are you motivated in learning to speak Danish native-like? Yes / No (underline, please)
- 18. Have you used Danish outside courses for the last year?
- 19. Do you use Danish more, than Russian? Please, indicate average relation in %.
- 20. Do you agree to take part in the presented study conditioned that your recordings will be used solely in the interest of and for the purpose of the current research project, and will not be disclosed to the research-unrelated persons?

Signature

Date
#### Appendix 2. Word List (WL1). Vowel and consonant segments

1		1	47	'1
]	I.	pande	47.	vikar
4	2.	dyne	48.	livlig
	3.	kærlighed	49.	skrive
2	4.	utaknemmelig	50.	dej
-	5.	badeværelse	51.	jod
6	5.	mangle	52.	økologi
7	7.	fare	53.	detalje
8	8.	ondskab	54.	kirsebær
ç	Э.	sofa	55.	længe
1	10.	rare	56.	bopæl
]	11.	bestemme	57.	måske
1	12.	hoppe	58.	kulde
1	13.	dum	59.	lammekød
1	14.	skarp	60.	banke
1	15.	kæresterere	61.	smadre
1	16.	storm	62.	kunde
1	17.	erkende	63.	næste
1	18.	sytten	64.	gange
1	19.	stride	65.	god
2	20.	gaffel	66.	gøre
2	21.	Kina	67.	menneske
2	22.	heddet	68.	bang
2	23.	vasket	69.	irokeser
2	24.	binde	70.	boliviansk
2	25.	apparat	71.	skole
2	26.	bagage	72.	kone
2	27.	panere	73.	piskefløde
2	28.	beslægtet	74.	korrektur
2	29.	forbillede	75.	norsk
3	30.	alene	76.	papir
2	31.	mad	77.	rigdom
2	32.	sene	78.	fræk
2	33.	falsk	79.	ris
3	34.	bue	80.	bær
3	35.	doven	81.	kirke
3	36.	gemme	82.	mor
2	37.	ganske	83.	presse
2	38.	flamme	84.	Lyngby
2	39.	gulv	85.	savne
4	40.	begejstret	86.	ryge
4	41.	skinne	87.	chokolade
4	42.	hemmeligt	88.	rutine
4	43.	film	89.	barsk
4	44.	vilde	90.	tøj
2	45.	kaj	91.	sjov
2	46.	hummus	92.	ømtålelig

93. muligvis 94. gerne 95. måle 96. akupunktur 97. vagt 98. kvota 99. havne 100.høne 101.kniv 102.koge 103.uge 104.korrupt 105.brev 106.tørklæde 107.munde 108.tidligt 109.fysiologi 110.vække 111.forskellig 112.trykke 113.beredskab 114.pædagog 115.nervøsitet 116.læse 117.fysisk 118.ungdommelig 119.kysse 120.købe 121.dreng 122.nødvendig 123.pædagogisk 124.smør 125.lukke 126.dromedar 127.borgmester 128.nærmere 129.storme

#### **Appendix 3. Target vowel segments in WL1**

Target words	Target segments
pande, panere, sofa	ą
kærlighed, erkende, kæresterere,	3
b <b>a</b> deværelse, bagg <b>a</b> ge,	3
b <b>a</b> nke, ut <b>a</b> knemmelig, <b>a</b> pparat <sup>27</sup>	ä
fare, rare	ä:
pædagog, pædagogisk, hemmeligt	e
læse	e:
dreng	a
binde, beslægtet, forbillede	ę
alene, sene	ę:
$gansk(\mathbf{e}), bu(\mathbf{e}), dov(\mathbf{e})n^{28}$	Э
vilde, vikar, livlig	i
skrive, Kina	i:
god <sup>29</sup> , irokeser, boliviansk	Q
skole, kone	Q:
rutine, akupunktur <sup>30</sup> , kulde	u
m <b>u</b> ligvis, <b>u</b> ge	u:
f <b>y</b> sisk, f <b>y</b> siologi, lyngb <b>y</b>	у
r <b>y</b> ge	y:
kysse, nødvendig, nervøsitet	Q
købe	Ø:
høne	œ:
trykke, ømtålelig, tørklæde	œ
smør	Œ
gøre	Œ:
lukke, ungdommelig	ç
måle	ð:
borgmester, korrupt, korrektur	ą
storme	ŷ:
ånder, forskellig, nærmere	$\Lambda$

<sup>30</sup> Here I took the second vowel [u] as a target segment

http://ordnet.dk/ddo/ordbog?query=akupunktur&search=S%C3%B8g

 $<sup>^{27}</sup>$  Most native speakers would reduce [ $\ddot{a}$ ] in the first pre-tonic syllable to either [ $\mathfrak{P}$ ] or [ $\mathfrak{I}$ ]. The targer segment in my analysis was the first  $[\ddot{a}]$  – in the second pre-stressed syllable. <sup>28</sup> While choosing the target words for  $[\exists]$  I took the pronunciation recommended by *Den Store Danske Ordbog*.

However, in a natural speech [ə] would be only pronounced in ganske. I return to this methodological issue in section 5.3.1.3. <sup>29</sup> Here I do not take *stød* into consideration for methodological reasons mentioned in section 3.1.

## Appendix 4. Target consonant segments in WL1

target words	target segment
bestemme, hoppe, skarp, ondskab	b b
<b>p</b> iskefløde, pa <b>p</b> ir	₿ <sup>h</sup>
havne, kniv, koge, brev	w/u
flame, gaffel, falsk, film	f
vække, vagt	V
dum, storm, sytten, dyne, tidligt	ģ
tøj, detalje, ømtålelig	<u> </u>
hemmeligt, hummus	h
jod	j
kaj, dej	j/į
længe, lammekød, bopæl, kulde	1
stride, mad, heddet, vasket <sup>31</sup>	₫ <sup>v</sup> ,
måske, smadre, menneske	m
kirsebær, økologi, kone	k
gulv, begejstret, skinne, fræk, gemme	ĝ
ba <b>nk</b> e, ga <b>ng</b> e, ba <b>ng</b>	ŋ
norsk, kunde, næste	n
presse, savne, fysisk, læse	S
sjov, chokolade	e
rigdom, ris, beredskab, irokeser	Ŕ

\_\_\_\_\_

 $<sup>^{31}</sup>$  In some regions of Denmark the final consonant would be [d], I took the Standard Copenhagen variant, as was mention in section 2.4. However, the regional variants of the pronunciation of this target word are discussed in section 5.3.2.4.

#### Appendix 5. Word list 2 (WL2). Words stress and diphthongs

- 1. billigst
- 2. eventyret
- 3. kvindelig
- 4. uheld
- 5. kajak
- 6. mistanke
- 7. sårbar
- 8. mistænksom
- 9. bearbejde
- 10. gebyr
- 11. ukendt
- 12. skjulte
- 13. grafik
- 14. søvnløs
- 15. violin
- 16. kritisere
- 17. jysk
- 18. uheldig 19. bilist
- 20. økonomisk
- 21. barndom
- 22. julefest
- 23. tyveri
- 24. umuligt
- 25. bagefter
- 26. jævnaldrende

- 27. femogtredive
- 28. ytringsfrihed
- 29. direkte
- 30. allerede
- 31. barnevogn
- 32. andetsprogspædagogik
- 33. lovgivning
- 34. evnesvag
- 35. peber
- 36. automatisk
- 37. drivhus
- 38. øvre
- 39. fløjet
- 40. dejlig
- 41. huje
- 42. kirkelig
- 43. Per
- 44. fersken 45. dyrke
- 46. kørsel
- 47. ørred
- 48. urbanisere 49. bortfalde
- 50. januar
- 51. hjørne

#### Appendix 6. Target words for the analysis of word stress assignment in WL2

Target word	Word stress
1. billigst	['bilisd]
2. bilist	[biˈlisd]
3. kvindelig	['ghven(ə)li]
4. gebyr	[ģeˈbyɐ̯]
5. eventyret	[ˈeːvənˌd͡syɐ̯ˀəð̪ˠ]
6. violin	[vioʻli <sup>2</sup> n]
7. kritisere	[ĝ <sub>p</sub> řiģi , sé <sub>s</sub> <sup>*</sup> ]
8. økonomisk	[øġoˌˈno²misġ]
9. grafik	[ģķäˈfiģ]
10. uheldig	[u'hel²di]
11. uheld	['u,hel <sup>?</sup> ]
12. mistanke	[ˈmisd͡säŋġə]
13. mistænksom	[mis'dsen, s,m?]
14. bearbejde	[b̥ẹäˈb̥äjd̥ə]
15. bagefter	$[b\epsilon^{2}(j)efd_{1}]$ or $[b\epsilon^{2}(j)efd_{1}]$
16. ukendt	['u,ghend]
17. umuligt	[uˈmu²lid]

18. femogtredive	[ˈfemʌ̯ˈd͡strað̪v̪və]
19. ytringsfrihed	[ˈyd̥ʁeŋsˌfʁihəð̪ˠʲ]
20. andetsprogs-pædagogik	['ạn(ə)ð̆', sþ¤osbhedago, gig]
21. barnevogn	['b̥äːn(ə)ˌvə̯wn]
22. julefest	['jul(ə)'fesd]
23. direkte	[ˌɑ̊iˌkagdə] or [ˌdɨˈkagdə]
24. allerede	$[, \dot{a}l(\vartheta), \dot{k}e; \tilde{Q}_{\Lambda}^{\Lambda}\vartheta]$ or $[, \dot{a}l(\vartheta), \dot{k}e; \tilde{Q}_{\Lambda}^{\Lambda}\vartheta]$
25. sårbar	[ˈsɔ̣ːˌb̥ä]
26. barndom	['b̥äːnˌd̥ʌᢩm²]

# Appendix 7. Target diphthongs in WL2

Target words	Target diphthong
1. ka <b>ja</b> k	[jä]
2. sk <b>ju</b> lte	[ju]
3. søvnløs	[œw]
4. <b>jy</b> sk	[jy]
5. tyveri	[yw]
6. jævnaldrende	[ew]
7. <b>ev</b> nesvag	
8. l <b>ov</b> givning	[ə̯w]
9. evnesv <b>ag</b>	[ɛj]
10. p <b>eb</b> er	[ęw]
11. <b>au</b> tomatisk	[äu]
12. dr <b>iv</b> hus	[iw]
13. øvre	[øw]
14. fløjet	[,i]
15. d <b>ej</b> lig	[äj]
16. h <b>uj</b> e	[uj/i]
17. k <b>ir</b> kelig	[iɐ̯]
18. Per	[ỹ3]
19. f <b>er</b> sken	[ỹ3]
20. dyrke	[yɐ̯]
21. kørsel	[øੲ]
22. ørred	[Ġš]
23. urbanisere	[uɐ̯]
24. bortfalde	[ắč]
25. <b>ja</b> nuar	[ją]
26. hjørne	[dīk]

Appendix 8. Transcription rea	ults for the vowe	l targets in the R-group
-------------------------------	-------------------	--------------------------

Target words		÷	[a]*	[e]*	*[3]	[8]*	[ə]*	[ạ:]*	[ạ]*	[8]*	[ä]*	[e:]*	[A:]*	[ã]*
p <b>a</b> nde		6	1					2						
panere	ą	4	3			2	2							
sofa		3				4	2							
erkende	ę	9 4		4			1							
kæresterere	C	3		6			1							
badeværelse		2		1	6									
bag <b>a</b> ge	33	4	1		2							1	1	
mangle		8												1
ut <b>a</b> knemmelig	ä	4				5								
apparat		4	1			3								
rare	ä:	0 5	1								4			
laic		5												
		Ļ	[I]*	[e]*	[-]*	*[3]	[e <sup>.</sup> ]*							
pædagog,		8	1											
pædagogisk	e	8	1											
hemmeligt		6				3								
læse	e:	8		2		1	1							
dreng	а	5		3		1								
		I	[i]*	[i:]*	[eː]*	[I]*	a[i]*	[i <sup>.</sup> ]*	[e]*					
1. 1		•		1			b[i:]*							
binde	0	2	6	1		7								
forbillede	ç	$\frac{2}{2}$	5			1		1						
alene		2	5	6		-		1						
sene	ę:	1	2	5				1						
ganske		9												
	ð													
bue	not	valid	method	lologic	allv• all	the R-	subject	s nrono	unced	invalid	target [	<b>`</b> _]		
doven	not	vanu		lologica	any. an	uic R-	subject	s prono	unceu	inivanu		9]		
vilde		3							6					
vikar	1	8				1								
living		9												
skrive	:.	7					1a	1						
Kina	1.	7	2											
		+	[0]*	[u]*	[u:]*	[ɔ]*	[o]*	[o:]*						
god		5		4										
irokeser	Q	3	5			1								
boliviansk		2	- 7	1	5		1							
kone	QĽ	2		$\frac{1}{2}$	5		1	1						
		1		2	5			1						

		+	[ʊ]*	[u]*	[u <sup>?</sup> ]*	[y <sup>.</sup> ]*	[uː]*	[Y]*	[I]*	[u·]*				
rutine		5	4											
ak <b>u</b> punktur	u	6	3											
kulde		8					1							
muligvis		3		3	3									
uge	u.	8								1				
fysisk		9												
fysiologi	y	9												
Lyngby	-	9												
rvge	V:	5		1		1	1	1						
		+	[ø.]*	[y]*	[ø]*	[y:]*	[ø] *	[3]*	[0]*	[y·]*	[ə]*	[ʊ] *	[θ]*	[œ]*
kysse		2	1	5						1				
nødvendig	a	9												
nervøsitet one	Ŷ	6						2						
failed														
købe	Ø:	3	1	2	2	1								
tr <b>y</b> kke		0		9										
ømtålelig	œ	0					2		2		3	1	1	
tørklæde		7												2
		+	[œ]*	[œ.]*	[œ]*	[œ:]	[Ē] *	[8]*	[θ]*	[Œ.]*	[Y:]*	[ʉ]*	[ø:]*	
høne	œ.	2	2	1							1	1	2	
smør	<u>ф</u> . Б	7		-	1				1		-	-		
gøre	ф. Ф.	6			1		3		-					
5010	<b>ч</b> .	0					5							
			[]*	[]*	[]*	[0:]*	[1*	[.]*	[_]*	[o:1*	[_]*	[11]*	[]*	[~]*
11-1		+	[u]**	[ų]* 1	[Y]**	[0.]	[0]*	[0]+	[Q]*	ΓŲ Ι.	[9]+	[ʉ]	[Q.]*	[ð].
	ç	0		1	1		6		1					
ungdommelig	I	0	2				6		1					
			F.1*	E July	F.1%	[]¥	F 3.4	F.1*	F .1*	[]¥	F 1*	E1.¥	Г <b>Т</b> ₩	Г 14
		+	[3]	[Ĵ.]*	[၁]*	[0:]*	[5:]*	[Q]*	[0.]*	[0:]*	[0]*	[6]*	[ɔ̃].	a[ɔ̥ː]* ⊾[₊]*
														D [θ]*
måle	<b>0</b> ]	4							2	2				1a
storme	- Ţ.	6			3									
borgmester		6			1			2						
korrupt	ç	Õ			_			1			7	1		
korrektur		6			3			-			,	-		
storme	<u>.</u>	6		1		2								
	ų.	5		1		-								
			[0]*	[0]*	[2]*	[ə]*	[œ:]*	[e] *	[0:]*	[8]*				
åndor		- 2	رما ۲	r.~1	LYJ ۸	r~1	r. 2. 1	r~1	1	L~J				
forskalling		2	2 5	1	4	n			1					
Torskeing	A ţ	1	5	1		2				1				
					1	1 1				1				

## Appendix 9. Transcription results for the vowel targets in the D-group

Target words	t/s	+	[a]*	[e]*	*[3]	[8]*	[ə]*	[ạ:]*	[ạ]*	[8]*	[ä]*	[a:]*	[ạ·]*	[ã̃]*
p <b>a</b> nde p <b>a</b> nere	ą	8 2	1 1			3	3							
sof <b>a</b>		1	1			6	1							
kærlighed	0	8		1										
erkende	υ	7					2							

kæresterere		8		1										
hadeværelse		1		1	3				4					
bagage	:3	1	4	1	1				1			1	1	
manala		5	+		1				1			1	1	1
		5	5			2			2					1
utaknemmelig	a	5	2			2								
<b>a</b> pparat		5	2			1	1							
f <b>a</b> re	ä.	4	1		1			1	2					
rare	u.	4	1						2		2			
		+												
			[I]*	[e]*	[-]*	*[3]	[e·]*	*[.3]	[i]*					
nædagog		7	1		1									
pædagogisk	P	ģ	-		1									
hommolist	C	2				4			2					
nemmengi		3				4	-	1	Ζ.					
læse	e:	2		2		2	2	1						
dreng	а	6		1		2								
	u													
		+	[i]*	[i:]*	[eː]*	[I]*	a[i]*	*[3]	[ɛ·]*	[e]*	[e]*	[i <sup>.</sup> ]*	[°]*	[-]*
							b[i:]							
							*							
binde		2	7											
beslægtet	ę	0				6		3						
forb <b>i</b> llede		5	2			2								
alene		3	2	1					2	1				
sene	ę:	3	1	1		1		1			1	1		
		-												
ganske	2	9												
gallske	9													
		met	nonoro	ucanv	invana									
1		E 1.4				, 5 34 (1)								
bue		[ə]*	(4), [ <sup>ə</sup> ]	*(3), [i	]* (1),	, [ε <sup>.</sup> ]* (1)	) in bue	;						
bu <b>e</b> dov <b>e</b> n		[ə]* [ə]*	(4), [ <sup>ə</sup> ] (5), [-]	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1)	) in bue	;						
bu <b>e</b> dov <b>e</b> n		[ə]* [ə]*	(4), [ <sup>ə</sup> ] (5), [-]	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1)	) in bue	;						
bu <b>e</b> dov <b>e</b> n v <b>i</b> lde		[ə]* [ə]* 3	(4), [ <sup>°</sup> ] (5), [-]	*(3), [i (4) in	[]* (1), doven.	, [ε <sup>.</sup> ]* (1)	) in bue	;			6			
bu <b>e</b> dov <b>e</b> n v <b>i</b> lde v <b>i</b> kar	i	[ə]* [ə]* 3 7	(4), [ <sup>°</sup> ] (5), [-]	*(3), [i   (4) in	[]* (1), doven.	, [ε <sup>.</sup> ]* (1) 2	) in bue	;			6			
bu <b>e</b> dov <b>e</b> n vilde vikar livlig	i	[ə]* [ə]* 3 7 9	(4), [ <sup>2</sup> ] (5), [-]	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1) 2	) in bue	;			6			
bu <b>e</b> dov <b>e</b> n v <b>i</b> lde v <b>i</b> kar livlig	i	[ə]* [ə]* 3 7 9	(4), [ <sup>2</sup> ] (5), [-]	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1) 2	) in bue	;			6			
bue doven vilde vikar livlig skrive	i	[ə]* [ə]* 3 7 9	(4), [ <sup>3</sup> ] (5), [-]	*(3), [i (4) in	]* (1), doven.	[ε <sup>·</sup> ]* (1)	) in bue	;			6			
bu <b>e</b> dov <b>e</b> n vilde vikar livlig skrive Kina	i i:	[ə]* [ə]* 3 7 9 1 2	(4), [ <sup>3</sup> ] (5), [-] 6 6	*(3), [i (4) in	]* (1), doven.	, [ε <sup>·</sup> ]* (1) 2	) in bue 1b	;			6	1		
bu <b>e</b> dov <b>e</b> n vilde vikar livlig skrive Kina	i i:	[ə]* [ə]* 3 7 9 1 2	(4), [ <sup>3</sup> ] (5), [-] 6 6	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1) 2	) in bue	;			6	1		
bu <b>e</b> dov <b>e</b> n vilde vikar livlig skrive Kina	i i:	[ə]* [ə]* 3 7 9 1 2	(4), [ <sup>3</sup> ] (5), [-] 6 6	*(3), [i (4) in	]* (1), doven.	, [ε <sup>.</sup> ]* (1) 2	) in bue	;			6	1		
bue doven vilde vikar livlig skrive Kina	i i:	[ə]* [ə]* [ə]* 3 7 9 1 2	(4), [ <sup>3</sup> ] (5), [-] 6 6	[u]*	[]* (1), doven.	[ε <sup>·</sup> ]* (1) 2	) in bue	;;	[œ·]*	[θ]*	6	1		
bue doven vilde vikar livlig skrive Kina	i i:	[ə]* [ə]* [ə]* 3 7 9 1 2 1 2	(4), [ <sup>a</sup> ] (5), [-] 6 6 6	(4) in	[]* (1), doven.	[ε <sup>-</sup> ]* (1) 2 [ο <sup>·</sup> ]*	) in bue 1b [ɔָ]*	;;	[œ·]*	[θ]*	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser	i i:	[ə]* [ə]* [ə]* 3 7 9 1 2 2 0	(4), [ <sup>a</sup> ] (5), [-] 6 6 6 7	(3), [i (4) in [u]* 1	[]* (1), doven.	, [ε <sup>-</sup> ]* (1) 2 [ο <sup>·</sup> ]*	) in bue 1b [ɔָ]*	;;	[œ·]*	[θ]*	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser bolivionsk	і і: о	[ə]* [ə]* [ə]* 9 1 2 4 2 0 1	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (7) (7) (7)	[u]* [u]*	[]* (1), doven. [0]*	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]*	) in bue 1b [?]*	;;	[œ·]*	[θ]*	6 [0`]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk	i i: 9	[ə]* [ə]* [ə]* 9 1 2 0 1	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (7), [-] (	[u]*	[]* (1), doven. [0]*	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]*	) in bue 1b [0]*	;	[œ·]*	[θ]*	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole	i i: o	[ə]* [ə]* [ə]* 9 1 2 0 1 0 1 0	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	[u]*	[]* (1), doven. [0]* 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3	) in bue 1b [0]*	;; [0:]*	[œ·]*	[θ]*	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone	i i: <u>o</u> <u>o</u> :	[ə]* [ə]* [ə]* [ə]*	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	[u]* [u]*	[]* (1), doven. [0]* 1 1 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3	) in bue 1b [ɔ]* 1 1 1	;; [0:]* 1 1	[œ·]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone	i i: <u> </u>	[ə]* [ə]* [ə]* [ə]*	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	[u]* [u]*	[]* (1), doven. [0]* 1 1 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3	) in bue 1b [ɔ]* 1 1 1	; [0:]* 1 1	[œ·]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone	i i: <u> </u> <u> </u>	[ə]* [ə]* [ə]* [ə]*	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (0)*	(4) in [u]*	[]* (1), doven. [0]* 1 1 1 1 1 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [0]* 1 1 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ`]* 1 [I]*	[θ]* 1	6 [0']* 1	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine	i i: <u> </u> <u> </u>	[ə]* [ə]* [ə]*	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (5), [-] (7), [3] (7), [3] (	(4) in (4) in [u]* 1	[]* (1), doven. [0]* 1 1 1 1 1 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [ɔ]* 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]* 1 [I]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur	i i:	[ə]* [ə]* [ə]* [ə]*	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (0)* (0)* (0)* (0)* (0)* (0)* (1) (0)* (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	(4) in (4) in [u]* 1	[]* (1), doven. [0]* 1 1 1 1 1 1 [u <sup>.</sup> ]*	, [ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [2]* 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde	i i: o o c : u	[ə]* [ə]* [ə]* [ə]* [ə]* [] 2 0 1 2 0 1 0 1 0 1 0 1 0 1 0 1	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (0)* 6 7 7 3 2 [0]* 1 1	(4) in (4) in [u]* 1	[]* (1), doven. [0]* 1 1 1 1 1 1 1 1 1	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [2]* 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]* 1 [I]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis	i i: o o c : u	[ə]* [ə]* [ə]* [ə]* [] 9 1 2 0 1 2 0 1 0 1 0 1 0 1 0 1 0 1 0 1	(4), [ <sup>3</sup> ] (5), [-] (5), [-] (5), [-] (0)* 6 7 7 3 2 [0]* 1 1	(4) in (4) in [u]* 1 [u]*	[]* (1), doven. [0]* 1 1 1 1 1 1 1 1 1 2	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [2]* 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]* 1 [I]*	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis uge	i i: o o c : u u u	[ə]* [ə]* [ə]* [ə]* [ə]* [] 2 0 1 2 0 1 0 1 0 1 0 1 0 1 1 0 1 2 0 1 2 0 1 2 0 1 2 0 1 2 0 1 2 0 1 2 1 2	(4), [°] (5), [-] 6 6 7 7 3 2 [v]* 1 1	*(3), [i (4) in [u]* 1 [u]* 6 7	[]* (1), doven. [0]* 1 1 1 1 1 1 1 2	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [0]* 1 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]* 1 [I]*	[θ]* 1	6 [0`]* 1	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis uge fysick	i i:	[ə]* [ə]* [ə]* [ə]* [ə]* [] 2 0 1 2 0 1 0 1 0 1 0 1 0 1 0 1 2 8 8 8 8 8 8 8 1 2 2 8	(4), [°] (5), [-] 6 6 7 7 3 2 [ $\upsilon$ ]* 1 1	<pre>*(3), [i (4) in [u]* [u]* [u]* 6 7</pre>	[]* (1), doven. [0]* 1 1 1 1 1 1 1 2	2 [φ <sup>·</sup> ]* 3 [y]*	) in bue 1b [0]* 1 1 1 [u:]*	; [0:]* 1 1 [Y]*	[œ·]* 1	[θ]* 1	6 [0`]* 1			
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis uge fysisk funiologi	i i: o o c: u u u u	[ə]* [ə]* [ə]* [ə]*	(4), [°] (5), [-] 6 6 7 7 3 2 [v]* 1 1	<pre>*(3), [i (4) in [u]* [u]* [u]* 6 7</pre>	[]* (1), doven. [0]* 1 1 1 1 1 1 2	2 [ο·]* (1) 2 [ο·]* 3 [y]*	) in bue 1b [0]* 1 1 [u:]*	; [0:]* 1 1 [Y]*	[œ·]*	[θ]* 1	6 [0 <sup>.</sup> ]* 1			
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis uge fysisk fysiologi	i i: o o c : u u u : y	[ə]* [ə]* [ə]* [ə]* [ə]* [] 2 0 1 2 0 1 0 1 0 1 0 1 0 1 0 1 0 1 2 8 8 6 7	(4), [°] (5), [-] 6 6 7 7 3 2 [v]* 1 1	(4) in (4) in [u]* 1 [u]* 6 7	[]* (1), doven. [0]* 1 1 1 1 1 1 2	[ε <sup>-</sup> ]* (1) 2 [ο <sup>-</sup> ]* 3 [y]*	) in bue 1b [0]* 1 1 [u:]*	;; [0:]* 1 1 [Y]*	[œ·]* 1 1 3	[θ]* 1	6 [0 <sup>.</sup> ]*	1		
bue doven vilde vikar livlig skrive Kina god irokeser boliviansk skole kone rutine akupunktur kulde muligvis uge fysisk fysiologi Lyngby	i i:	[ə]* [ə]* [ə]* [ə]* [ə]* [] 2 0 1 2 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1	(4), [°] (5), [-] 6 6 7 7 3 2 [v]* 1 1	<pre>*(3), [i (4) in [u]* [u]* [u]* 6 7 </pre>	[]* (1), doven. [0]* 1 1 1 1 1 1 2	2 [ο·]* (1) 2 [ο·]* 3 [y]*	) in bue 1b [0]* 1 [u:]*	;; [0:]* 1 1 [Y]* 2 2	[œ·]* 1 1 3	[θ]* 1	6 [0 <sup>.</sup> ]*			

		÷	[ø.]*	[y]*	[ø]*	a[o]* b[œ:] *	[ø] *	[3]*	[œ]*	[8]*	[u]*	[ʊ] *	[Y]*	a[θ]* b[ʉ]*
kysse nødvendig nervøsitet	Ø	3 7 6	1	5		1a	1	1	1	1				
købe	Ø:	4	2		3									
tr <b>y</b> kke ømtålelig tørklæde	œ	0 2 5		6		1a 1b	2		1		1	1	1 1	1a, 1b 1a
			5 34	5 3.4	5 J.4	5 3.6	5 3.4	5 3 de	5 . Data	5 3.4	5.34	5 34		
1		+	[œ]*	[œ.]*	[œ]*	[œ:]*	[Ē]*	[8]*	[θ]*	[Œ.]*	[0]*	[0.]*		
høne	ϕ	3	2	2	1	I					1	1		
smør	Ē	3			4		2		1	1	1	1		
gøre	Ē.	3			1		3		1	1				
					5.3.	5 3 di				r 34		5.34		
		+	[u]*	[ų]*	[ø]*	[0:]*	[ʊ]*	[၁]*	[o]*	[ŏ.]*	[ə]*	[ʉ]*		
l <b>u</b> kke ungdommelig	Ç I	2 1	3 1	2	1		4	1	1	4	1	1		
		<b>.</b>	[ <u>;</u> ]*	[0.]*	[0]*	[0:]*	[q:]*	[o]*	[ð.]*	[0:]*	[0]*	[œː]*	[ŷ]*	a[ə <sup>.</sup> ]* b[ə] *
måle	₽.	0	1	1	1	1	1	1	1	1	1	1		
storm b <b>or</b> gmester korrupt korrektur	Ş	5 6 1 2			3 1 3			1			4 7 3			1b
storme	Ş:	3		1	3								1	1a
		+	[0]*	[ɔ]*	[ɔ̯]*	[ə]*	[œː]*	[θ] *	[e]*	[b]*				
<b>å</b> nd <b>er</b> forskellig nærmer <b>e</b>	Λ <u> </u> <u> </u>	3 3 4	1 1	2 2	3	2 4		1	1	1				

## Appendix 10. Transcription results for the diphthong targets in the R-group

Target words	t/d	÷	[ja]*	[ʊ]*	[θə]*	[θw]*	[œw]*	[œ៉ <sub>ñ</sub> ]*	[yv]*
ka <b>ja</b> k	[jä]	8	1						
sk <b>ju</b> lte	[ju]			1					
s <b>øv</b> nløs	[œw]	3			1	3	1	1	
<b>jy</b> sk	[jy]	9							
t <b>yv</b> eri	[yw]	8							1
		+	[iw]*	[ɛv]*	[jạ]*				
jævnaldrende	[ow]	9							
evnesvag	[ew]	3	3	2	1				
lovgivning	[ə̯w]	9							

		+	[aj]*	[ɛj]*					
evnesvag	[ɛj]	7	1	1					
		+	[eːb]*	[ib]*	[iw]*	[ęb]*	[၁]*		
p <b>eb</b> er	[ew]	3	1	3	1	1			
automatisk	[äu]	6					3		
		+							
dr <b>iv</b> hus	[iw]	9							
		+	[øv]*						
øvre	[øw]	7	2						
		+	[ʉj]*	[ɔj]*	[ə̯əj]*				
fløjet	[ʌj]	2	1	5	1				
		+	[u]*						
d <b>ej</b> lig	[äj]	9							
h <b>uj</b> e	[uj/i]	7	2						
		+	[i]*	[eɐ̯]*	[eɐ̯]*	[fjɐ̯]*			
kirkelig	[iɐ̯]	8	1	2					
Per	[¥3]	/		2					
f <b>er</b> sken	[¥3]	2			6	1			
		+	[ʊ]*	[y]*	[y:]*				
d <b>yr</b> ke	[yɐ̯]	6	1	1	1				
		+	[ø̈́ɛ]*	[œ]*	[ø]*	[ĒŔ]*	[ <b>\omega F</b> ]*	[œŔ]*	[œ5]*
kørsel	[øɐ̯]	2	5	1	1				
ørred	[Ġš]	0				3	2	3	1
		+	[u]*	[ʊ]*	[ΩŘ]*	[o]*	[0]*		
urbanisere	[uɐ̯]	0	2	5	2				
bortfalde	[ðð]	1				2	6		
<b>ja</b> nuar	[jạ]	9							
		+	[Ē]*	[θ]*	[Ē:] <b>*</b>				
h <b>jø</b> rne	[ĠĂ]	2	5	1	1				

Target words	t/d	+	[ja]*	[jə]	[ɒw]*	[ow]*	[YY]*	[œ̪ <sup>ŭ</sup> ]*	[jʉ]*	[y <sup>w</sup> ]*	[yv]*
ka <b>ja</b> k	[jä]	4	4	1							
sk <b>ju</b> lte	[ju]	9									
søvnløs	[œw]	4			1	1	1	2			
<b>jy</b> sk	[jy]	9							1		
t <b>vv</b> eri	[yw]	5								1	3
		+	[aw]*	[æw]*	[e <sup>ə</sup> ]*	[vw]*	[ɛv]*	[ɛ <sup>v</sup> ]*	[ow]*	[iw]	[yv]*
jævnaldrende evnesvag	[ew]	2 4	3	2	1	1	3	1		1	
lovoivnino	[ow]	6					-			_	3
lovgrvning	L ¥ ··· ]	~									
			[vi]*	*[3]	[ɛ <sup>j</sup> ]*	[ɛ <sup>i</sup> ə]*	[v]*				
evnesv <b>a</b> g	[ɛi]	2	1	2	1	2	1				
concorug	[9]										
			[e:b]*	[ɛb]*	[ib]*	[eb]*	[ɔ]*	[äv]*	[gu]*		
peber	[ew]	3	1	3	1	1					
automatisk	[äw]		-	0	-	-	1	1	1		
uutomutisk	[]										
			[iu]	[ɛw]*							
dr <b>iv</b> hus	[iw]	7		2							
	[1,1]	,		_							
			[œv]*	[ŒW]*	[ø:w]*	[œ]*	[œw]*				
øvre	[øw]	4	1	1	1	1	1				
	L, J										
			[e <sup>j</sup> ]*	[0]*	[vi]*	[^j]*	[œ]*				
fløiet	[4]	+ 1	4	1	1	1	1				
	1401	-									
			[u]*	[ai]*	[ʉi]*						
deilig	[äj]	8		1	L 30 1						
huie	[ui/i]	4	4		1						
	1										
			[i]*	[iə]*	[iə]*	[eɐ]*	*[93]				
k <b>ir</b> kelig	[iv]	1	2	5	1		L- 41				
Per	[83]	7	<u> </u>			2					
Complex a	[~~]	8					1				
Tersken	႞ၓၟႄ႞	0					1				
			p Date	E 34	ل ا						
		+	[y*]*	[y]*	[y:]*						
d <b>yr</b> ke	[yĕ]	1	4	3	1						
		+	[øɐ̯]*	[œ]*	[ø]*	[ĠŔ]*	[œə]*	[œř]*	a[œɣ]* b[œ̥ə]*	[ø:]*	[œğ]*
kørsel	[øĕ]	1	1	1	2					4	
ørred	[ĖŠ]	0				2	3	1	1a, 1b		1

## Appendix 11. Transcription results for the diphthong targets in the D-group

		+	[u]*	[ʊ]*	[ʉ]*	[o]*	[o]*	[ja]*	[jɐ]*	
<b>ur</b> banisere	[uɐ̯]	1	2	5	1					
bortfalde	[àċ]	0				1	8			
<b>ja</b> nuar	[jạ]	3						4	2	
		+	[Ē]*	[θ]*	[Ē:] <b>*</b>					
hjø <b>r</b> ne	[¢š]	0	3	1	5					

# Appendix 12. Transcription results for the consonant targets in the R-group

Target words	t/s	+	[þ]*	[þʰ]	[b]*	[w]*	[ŭ]*	[p]*	[v <sup>j</sup> ]*	[u̯j]*	[-]*	[u̯]*	[v <sup>f</sup> ]	
bestemme hoppe skarp ondskab	þ	6 8 9 7			3			1 2						
<b>pi</b> skefløde pa <b>p</b> ir	<u>þ</u> հ	6 6						3 3						
havne kniv koge brev	w/ ų	9 9 9 9												
flame gaffel falsk film	f	9 9 9 9												
vække vagt	v	5 9							4					
		Ŧ	[d]*	[ģ]*	[ds]*	[ď],	[d]*	[d]*	[t]*	[ <u>t]</u> *	[ <u>t]</u> *	[ts]*		
dum storm sytten dyne tidligt	ģ	3 9 9 3 3			1	1		5	1			1		
tøj detalje ømtålelig	ds	3 5		1					1			6 2		
		4										5		
		4										5		
		4	[h]*	[X]*	[x <sup>j</sup> ]*	[h <sup>j</sup> ]*						5		

		Ļ											
jod	j	9											
kaj	j/į	9 9											
uej													
			r11.4	rii w									
		+	[1]*	[ŀ]*									
længe lammekød bopæl kulde	1	6 8 1 7	1	3 8 2									
		Ŧ	[ŗ]	[ł]*	[l <sup>j</sup> ]*	[j]*	[ð]*	[-]*	[1]*	[t]	[d]		
stri <b>d</b> e ma <b>d</b> he <b>dd</b> et vaske <b>t</b>	₫Ÿ,	8 1 8 9						1					
			[m <sup>j</sup> ]*										
måske		• 9											
smadre		9											
menneske		7	2										
		Ļ	[k]*	[g]*	[-]*	[Y]*	[g <sup>j</sup> ]*	[ģ <sup>h</sup> ]	[k <sup>j</sup> ]*	0			
kirsebær		5	3						1				
økologi kone	$\mathring{g}^{h}$	8 5	1 3							1			
gulv		4	1	4									
be <b>g</b> ejstret		2		4	2		1						
skinne	ĝ	2	5										
fræ <b>k</b>		9											
gemme		6		1				2					
			[m]+	[mc]*	[m]-1*	[m]1*							
1		+	[IJ↓	[ŋg]*	[IJК]≁	[II']*							
ba <b>nk</b> e	n	9											
gange	IJ	9											
norsk		8	1										
ku <b>nd</b> e	n	9											
næste		8				1							
			[ <u>s</u> ]*	[z]*	[§ <sup>j</sup> ]*	[s <sup>j</sup> ]*	[sj]*	[ç]*	[\$]*	[ţʃ]*	*[]]		
		and the second			- <b>1</b>		- 94	L 5 J	r Ch	- 0 -	L) J	 	
presse	ł – –	9											
presse savne	s	9 5	4										
presse savne fysisk	s	9 5 9	4										

sjov	0	5						3	1				
<b>ch</b> okolade	6	3						1	2		3		
		+	[1]*	[R]*	[r]*	*[1]	[γ]*	[-]*	$[\hat{\mathbf{R}}_{\boldsymbol{\lambda}}]_{*}$	[3]*	$[R]_*$		
<b>r</b> igdom		6							1	1	1		
ris		7								1	1		
be <b>r</b> edskab	Ŕ	8					1						
i <b>r</b> okeser		8		1									
		+	[Ŕ]*	*[I]	[-]*								
bær		9											
ki <b>r</b> ke	p	9											
mor	×	8			1								
gerne		9											

# Appendix 13. Transcription results for the consonant targets in the D-group

Target words	t/s	Ŧ	[þ]*	[þʰ]	[b]*	[w]*	[u̯]*	[p]*	[v]*	[u̯j]*	[-]*	[u̯]*	[v <sup>f</sup> ]	
bestemme hoppe skarp ondskab	þ	3 6 4 5			6			3 5 4						
<b>p</b> iskefløde, pa <b>p</b> ir	Å,	4 2	1					4 7						
havne		9												
kniv	w/	8							1					
koge	u	0								1	8			
brev		9												
flama		0												
name		9												
falsk	f	9												
film		9												
vække		9												
vagt	v	8											1	
		+	[d]*	[ģ]*	[d͡s]*	[-]*	[d]*	[d]*	[t]*	[ <u>t</u> ]*	[ <u>t]</u> *	[t]*		
dum		3	1				3	2						
storm		5							2	2				
sytten	ģ	4							4	1				
dyne		2					5	2						
tidligt		2			1	2			2	1		1		
tøj	÷	2							5	1	1			
detalje	<b>d</b> s	3							1	4		1		
ømtålelig		2							5	2				

		÷		[x]*									
1 1 4		7		2									
nemmeligt	h			2									
hummus		6		3									
			[œ]*										
		0	1										
jod	j	8	1										
koj		9											
кај	j/i	9											
dej	•												
			[ <u>+</u> ]*	[] <sup>j</sup> ]*									
		+	2	r. 1									
længe		0	3										
lammekød	1	6	3										
bopæl	1	8		1									
kulde		7		2									
										à			
			[ţ]*	[1]*	[4]*	[]*	[ð]*	[-]*	[1]*	[ <sup>2</sup> ]*	[t]*	[ģ]*	
stri <b>d</b> e		5	1			1	1	1					
ma <b>d</b>		6			1			1	1				
haddat	ðŸ,	7					1			1			
neuuet							1			1			
vasket		3							1		4	1	
		+											
måske		9											
amadra		Ó											
smaare		9											
menneske		9											
		+	[k]*	[g]*	[-]*	[Y]*	[g <sup>j</sup> ]*						
kirsebær		5	4										
delaci	åh	6	2										
økologi	g	0	3										
kone		8	1										
gulv		4		5									
begeistret		0		8	1								
skinno	å	2	7	-	-								
SKIIIIC C I	8	~	2			1							
fræk		5	3			1							
gemme		4		3			2						
			[n]*	[ŋg]*	[ŋk]*								
banke		<b>†</b>	L m J	2304									
Janke		9		_									
ga <b>ng</b> e	ŋ	4		5									
ba <b>ng</b>		6			3								
norsk		5	4										
ku <b>nd</b> e	n	6	3										
Runue	11	5											
næste		5	4										

		+	[ <u>s</u> ]*	[z]*	[s̪ <sup>j</sup> ]*	[s <sup>j</sup> ]*	[s <sup>j</sup> ]*	[ç]*	[§]*	[tʃ]*	[ʃ]*		
presse		6	3										
savne	s	5	4	4									
fy <b>s</b> isk	3	5	1	4 2									
læse			2	2									
sjov	C	2			1		2	3	1				
<b>ch</b> okolade	ø	6							1	1	1		
		+	[1]*	[R]*	[r]*	*[1]	[ɣ]*	[-]*					
rigdom		4	2	1	1	1							
ris		6			2	1							
be <b>r</b> edskab	Ŕ	4			1	1	1	2					
i <b>r</b> okeser		5			2	1		1					
		+	[Ŕ]*	[1]*	[-]*	[ə]*							
bær		8											
ki <b>r</b> ke	n	5	1	1	2								
mo <b>r</b>	ĕ	5			4								
gerne		5			3								

#### Appendix 14. Transcription results for the word stress assignment in the Rgroup

Target word	Word stress	<b>_</b>	Word stress*	
billigst	['bilisd]	8	bi'list* (1)	
bilist	[bi'lisd]	2	'bilist* (7)	
kvindelig	[ˈgʰve̯nəli]	9		
gebyr	[ģeˈbyɐ̯]	9	even'tyret (8)	
eventyret	[ˈeːvən d͡syɐ̯ˀəð̪ˠ]	1		
violin	[vioˈli²n]	9	1 failed	
kritisere	[ĜŗŔiģi zé,v]	8		
økonomisk	[øåo'no <sup>?</sup> miså]	9	'grafik* (1), 'grafisk* (1)	
grafik	[ģķäˈfiģ]	7		
uheldig	[u'hel'di]	7	u'heldig* (1), 'u heldig* (1)	
uheld	['u hel <sup>?</sup> ]	4	u'held* (2), u'held* (2), 'uheld* (1)	
mistanke	[ˈmisd͡säŋġə]	2	mis'tanke* (7)	
mistænksom	[mis'dsen sim?]	9		
bearbejde	[beäˈbäjdə]	0	be'arbejde* (2), 'be arbejde* (1), be'arbejde* (4),	
			'bearbejde* (1), 'bear bejde* (1)	
bagefter	[ˈbɛ²jˈefdʌ] or	2	bag'efter* (3)	
	[ˈbɛ²j efdʌ]	4		
ukendt	['u ghend]	5	u'kendt* (3), 'u'kendt* (1),	
umuligt	[u'mu'lid]	9		
femogtredive	['femʌˈdstrað̪vvə]	4	femog'tredive* (4), 'femog tredive* (1)	
ytringsfrihed	[, Å Å Å Å Å Å Å Å Å Å Å Å Å Å Å Å Å Å Å	3	yt'rings fri hed* (2), ytrings frihed* (3)	
			yt rings fri hed* (1)	
andetsprogs-	['anəðv_sb¥osbhe	1	'andet'sprogspæda'gogisk*	(2),
pædagogik	daģo ģiĝ]		andet sprogspæda gogik* (3)	
			'andet sprogspæda gogik* (1),	
			andet sprogspædago gik* (1),	
			andet sprogspæda go gik* (1)	

barnevogn	[ˈb̥äːnəˌvə̯wn]	9	
julefest	['julə'fesd]	3	'jule fest* (4), 'julefest* (1), 'jule'fest* (1)
direkte	[ˌˈdiˌˈƙa͡ɕd̥ə] or	1	di'rekte* (8)
	[ˌɑ̊iˈƙagd͡ə]	0	
allerede	[ˌåjəˌřeːỹ̯ʎ ə] ou	0	allerede* (7)
	[ˌåjəˈƙeː፬̯ʎ ə]	2	
sårbar	[ˈsɔːˌbɑ]	9	
barndom	['b៉äːnˌdʌ̯m <sup>?</sup> ]	4	'barndom* (5)

#### Appendix 15. Transcription results for the word stress assignment in the Dgroup

Target word Word stress		+	Word stress*	
billigst	['bilisd]	8	bi'list* (1)	
bilist	[biˈlisd]	5	'bilist* (4)	
kvindelig	[ˈģʰvenəli]	9		
gebyr	[ģeˈbyɐ]	9		
eventyret	['eːvən d͡syɐ̯ˀəð̪ˠ]	1	even 'tyret*(7), eventyr 'et* (1),	
violin	[vioˈli²n]	9		
kritisere	[Ĝ <sub>p</sub> Řiďi, sé <sub>3</sub> Ť]	9		
økonomisk	[ø̥ġo̯ˈno̯²misǧ]	9		
grafik	[ģ̃kɑ̈ˌlīġ]	6	'grafik* (3),	
uheldig	[u'hel <sup>?</sup> di]	5	u'heldig* (1), 'u heldig* (1), 'uheldig (2)	
uheld	['u,hel <sup>?</sup> ]	4	u'held* (2), 'uheld* (2), u'held* (1)	
mistanke	[ˈmisd͡säŋġə]	3	mis'tanke* (6)	
mistænksom	[mis'dsen sym?]	6	mistænksom* (2), mis tænksom* (1)	
bearbejde	[b̥ẹ̈̈ɑ̈̈́ḍə]	0	1 not valid, be'arbejde* (2), 'be arbejde* (1)	
			bear bejde* (2) bear bejde* (1), bearbejde* (2)	
bagefter	[ˈbɛ²jˈefdʌ] or	2	bag'efter* (5)	
-	['bɛ²j efdʌ]	2		
ukendt	['u ghend]	3	u'kendt* (3), u'kendt* (2), 'ukendt* (1)	
umuligt	[uˈmu²lid]	5	'umuligt* (2), 'u muligt* (2),	
femogtredive	['femʌ̯ˈd͡strað̪ˠvə]	0	femog'tredive* (8), femogtre'dive* (1)	
ytringsfrihed	[, Åďřeů2 'třipáy]	4	yt'ringsfri hed* (2), ytrings frihed* (1),	
			'ytringsfri hed* (1), y'trings fri hed* (1),	
andetsprogs-	['anəðv, sþrosbhedago, gig]	0	'andetsprogspædago'gik*	О,
pædagogik			andet sprogspæda gogik* (1),	
			andet sprogspæda gogik* (1),	
			andet sprogspædago gik* (5),	
			andet sprogspæda gogik* (1)	
barnevogn	[ˈb̥äːnəˌvə̯wn]	8	'barnevogn* (1)	
julefest	['julə'fesd]	3	jule fest* (6)	
direkte	[ˈd̪iˈk̄agdə] or	0	di'rekte* (8), _di'rekte* (1),	
	[ˈdiˈkağdə]	0		
allerede	[, ˈajə, ke: Ṣ̃ <sup>,</sup> ə] o.	1		
	[,ˈájəˈře:Q̃ <sup>,</sup> ə]	0	'allerede* (3), alle 'rede* (5)	
sårbar	[ˈsɔːˌbä]	9		
barndom	['bäːn dʌm <sup>?</sup> ]	3	'barndom* (6)	

### Appendix 16. Global accent rating sheet

Subject's No	Accent degree rating	Comments
please, note, that a subject's number	please, give a point from the scale below	
corresponds to the audio	<ol> <li>heavy accent</li> <li>considerable</li> </ol>	
file	accent 3 - slight accent 4 - <i>almost</i> native- like	
1	<b>5</b> - native-like	
31		

# Appendix 17. Correspondence between PPNs and numbers in the global accent rating sheet

1.1.	1
1.2.	6
1.3.	3
1.4.	19
1.5.	5
1.6.	13
1.7.	7
1.8.	20
1.9.	25
1.10.	4
1.11.	22
1.12.	12
1.13.	21
1.14.	8
1.15	23
2.1.	31
2.2.	2
2.3.	24
2.4.	15
2.5.	10
2.6.	26
2.7.	11
2.8.	27
2.9. did not participate	
2.10.	16

2.11.	28
2.12.	29
2.13.	17
88	9
89	18
91	30
92	14

# Appendix 18. Correspondence between transcription symbols in *IPA*, *Den Danske Ordborg* and *Dania*

IPA	Den Danske Ordbog	Dania
	http://ordnet.dk/ddo/	
ą	a	a
ε	æ	à
3	æ:	à.
ä	a	α
ä:	a:	α.
þ	b	b
ģ	d	d
ð۲.	ð	ð
ę	e	e
ę:	e:	e·
ə	ə	Э
f	f	f
0¢∘	g	g
h	h	h
i	i	i
i	i:	i <sup>.</sup>
j/į	j	j
j	j	j
β <sup>h</sup>	k	k
1	1	1
m	m	m
n	n	n
ŋ	ŋ	ŋ
Q	0	0
Q:	0:	0.
<u>þ</u> h	p	р
Ŕ	R	r
Ŕ	ğ	I
S	S	S

G	G	ş
ds	t	t
u	u	u
u:	u:	u <sup>.</sup>
V	V	V
w/ų	W	Y
у	у	y:
y:	y:	у.
	0	
e	ъ 	æ
e:	3	æ
a	a	ä
Ø	ø	ø
Ø:	ØĽ	ø.
œ	œ	ö
œ:	œ:	ö <sup>.</sup>
œ:		
		<u>?</u> .
œ		2
Ē	Œ	ö
Œ.	ŒĽ	ÿ.
5 <sup>†</sup>	Э	å
Ĵ.	o:	å.
Ş	D	å
ò;	DI	å·
Ŷ	Λ	0

#### Appendix 19. Transcriptions of the vowel and consonant targets in the Rgroup

R 1	R 4	R 5	R 6	R 7
[ <mark>p</mark> ạnə]	[ <mark>p</mark> ạnə]	[ <mark>p</mark> ạnə]	[ <mark>pa</mark> ːnə]	[b̥ʰạ̯nə]
[ˈdɣːnə]	[ˈ <mark>d</mark> ʲynə]	[ˈ <mark>d</mark> ynə]	[ˈ <mark>d</mark> yːnə]	[ˈdɣːnə]
[ˈ <mark>k</mark> ɛɐ̯liˌhəð̪ˠ]	['ġʰɛɐ̯l <sup>j</sup> i hɪð̪ˠ]	[ˈġʰɛɐ̯li <b>ˌh</b> ẹð̪ˠ]	[ˈġʰɛɐ̯liˌx <sup>j</sup> ęðᢩˠ]	['ĝ <sup>h</sup> ɛɐ̯li hẹð̪ˠ]
['utsäg'nemeli]	[ʊˈd͡sɐɡ̊nəmeli]	[uˈd͡sägˈknemeli]	[udsg <mark>vk</mark> 'nemeli]	[udsäg neməli]
['b̥ɛð̃ˠ ə v <sup>j</sup> eɐ̯lsə]	['b̥ɛð̪ˠ ə veɐ̯lsə]	['b̥ <mark>e</mark> ð̃ <sup>v</sup> ə veɐ̯lsə]	[ˈb̥ɛð̃ˠ əˈv <sup>j</sup> elsə]	[ˈb͡sði əˈʌskilsə]
[ˈm <mark>ɑ</mark> ̈̈ŋlə]	[ˈmäŋlə]	[ˈmäŋlə]	[ˈmäŋlə]	[ˈmäŋlə]
[ˈfäːä]	[ˈfäːä]	['fä:]	[ˈfäːə]	[ˈfäː <mark>ʁ</mark> ̞ə]
['onsˈgɛb]	['ọnˌ <mark>s</mark> gɛp]	['o²nˌsġɛb̥]	['onˌsgɛb]	[ˈo̯nsg̊əb̥]

[ˈsoºːfə]	[ˈ <mark>s</mark> oːfɐ]	[soˈfa̯]	[ˈs <mark>o</mark> fa̯]	[ˈsoːfə]
[ˌҟɑː <mark>1</mark> 9]	[ˌř <mark>ɑ</mark> ]	[ˌkɑːɑ]	[ˌŔ <mark>ɑ</mark> 9]	[ˌŔ <u>ɑ</u> ː <mark>Ŕ</mark> 9]
[b̥ <mark>ı</mark> ˈsd̥emə]	[ <mark>bɪˈs̪t</mark> ɛmə]	[b <mark>ı</mark> ˈsd̥emə]	[b̥ <mark>ı</mark> ˈsd̥emə]	[b̥ <mark>ı</mark> ˈsd̥ɛmə]
[ˈh <mark>op</mark> ə]	[ˈhəb̥ə]	[ˈhʌᢩbə]	[ˈhob̥e]	[ˈhʌᢩbə]
[tu <sup>ʔ</sup> m]	[ <mark>d</mark> um]	[dum]	[ <mark>d</mark> um]	[d <mark>̥o</mark> m]
[s <mark>ġä</mark> þ]	[ <mark>sg<sup>h</sup>ä</mark> b]	[sġäːþ]	[sģäːþ]	[sġäːþ]
[keɐ̯sˈd <mark>eɐ</mark> ̯]	[ˈ <mark>k<sup>j</sup>eɐ̯s</mark> d̥ɪə]	[keʌsˈdiʌʌ]	[¦ĝ <sub>p</sub> ešeďvv]	[ˌୡ <mark>ə</mark> ekəsdəkekə]
[sdɔ͡,m]	[ <mark>s</mark> d̥ɔm]	[ <mark>s</mark> d̥əm]	[sdɔm]	[sd̥ɔ̥ˀm]
[eɐ̯ˈ <mark>k</mark> entə]	[ˈ <mark>eɐ</mark> ̯g͡ənə]	[ˈ <mark>e</mark> ɐ̯g̊ənə]	[ɛɐ̯ˈgʰenə]	[ˈɛɐ <mark>̯ġə</mark> nə]
[ˈsød̥n]	[ˈ <mark>s</mark> ʲydn]	[ˈ <mark>s</mark> yd̥n]	[ˈ <mark>s</mark> yd̥ən]	[ˈs <mark>y</mark> d̥n]
[ˌsď祚iː <b>;</b> 9]	[ˈ <mark>s</mark> d͡͡�ið̪ˠ ə]	[ˌzģᠷ <mark>i</mark> ݡᠯ,ə]	[ˌ <mark>s</mark> ƙiਊʎə]	[ˌsďři.ŷř ə]
[ˈ <mark>k</mark> äfəl]	[ˈģ <mark>a</mark> fəl]	[ˈģäfəl]	[ˈ <mark>g</mark> äfəl]	[ˈģäfəl]
[ˈ <mark>k</mark> iːna̯]	[ˈġʰiːnə]	[ˈĝʰiːnə]	[ˈĝʰiːnə]	[ˈĝʰiːnɐ]
[ˈh <mark>eu</mark> ]	[ˈh <mark>i</mark> ðᢩvəd]	[ˈhẹð̪ˠəd]	[ˈ <mark>x<sup>j</sup>i</mark> ðᢩvəd]	[ˈhẹð̪ˠᢩə <b>t</b> ]
[ˈvạsg̊ə ð̪ˠ]	[ˈvạ <mark>s</mark> gəð̪ˠ]	[ˈvạsġəð̪ˠ]	[ˈvạsĝəð̪ˠ]	[ˈvạsġəð̪ˠ]
[ˈb̥inə]	[ˈ <mark>bi</mark> nə]	[ˈb <mark>iːn</mark> ə]	[ˈ <mark>bi</mark> nə]	[ˈb̥enə]
[ <mark>s</mark> þɑˌネɑ٫ð]	[ <mark>s</mark> βsˌkɑð]	[ɑɡʰɑˌkɑɡ]	[ɑɡʰɑˌkɑɡ]	[ɑp̊ɑˌk͡ɑd]
[caːɜģˈˈ <mark>ˈˈ</mark> d]	[caːɜģˈ <mark>ˈsd</mark> ]	[ca <mark>3</mark> gˈˈ <mark>ˈ</mark> d]	[ຣວ <mark>3</mark> gໍ <mark>ອd</mark> ]	[b̥ɐˈˈåʌːɛə]
[pa'n ea]	[ <mark>p</mark> ạˈn <mark>iə</mark> ]	[p <mark>ɐˈn̪iə</mark> ]	[ <mark>paˈniə</mark> ʌ]	[þʰɐˈnẹ <mark>ʁฺə</mark> ]
[b̥ <mark>ɪˈs̪leg̊təð̪</mark> ˠ]	[ <mark>bɪˈs̪l<sup>j</sup>eģd̥əð̪ɣ]</mark>	[b̥ <mark>ɪˈs</mark> leģd̥əð̪ˠ]	[ <mark>bıˈslɛģd̥əð̪</mark> ˠ]	[b̥eˈsleģd̥əð̪ˠ]
[f <mark>ɔ</mark> 'b̥ <sup>j</sup> iləð̪ˠ ə]	[f <mark>oˈb<sup>j</sup>i</mark> ləð̪ˠ ə]	[ˌf <mark>ə</mark> ˈb̥ <mark>i</mark> ːləð̪ˠ ə]	[fɔˈ <mark>bi</mark> ləð̪ˠ ə]	[ˌfɔˈb̥eləð̪ˠ ə]
[ɐˈliːnə]	[əˈl <sup>j</sup> iːnə]	[ạˈl <mark>i</mark> ˈnə]	[ạˈ <mark>li</mark> ːnə]	[əˈliːnə]
['mạ ð̪ˠ ]	[ˈmạð̪ˠ ]	[ˈmąð̪ˠ ]	['mạ ð̪ˠ ]	[ˈmạð̪ˠ ]
[ˈsiːnə]	[ˈ <mark>s</mark> iːnə]	[ˈ <mark>siˈn</mark> ə]	[ˈsiːnə]	[ˈ <mark>s</mark> iːnə]
[ˈfalsk]	[ˈfal <mark>s</mark> g]	[ˈfalsg̊]	[ˈfalsg]	[ˈfalsg̊]
[ˈb̥ <mark>u</mark> ə]	[ˈ <mark>bu</mark> ːə]	[ˈb̥uːə]	[ˈb̥uːə]	[ˈb̥uːə]
[ˈdɔ̃uən]	[ˈdɔ̃uən]	[ˈdɔ̃u̯ən]	[ˈdɔ̃u̯ən]	[ˈdɔ̃uən]
[ˈģemə]	[ˈģemə]	[ˈģemə]	[ˈ <mark>g</mark> ʲemə]	[ˈģemə]
[ˈģ <mark>ä</mark> nskə]	[ˈģansġə]	[ˈġ <mark>a</mark> nsġə]	[ˈ <mark>ģ</mark> ʲa̯nsɡ̊ə]	[ˈġʌnsġə]
[ˈflämə]	[ˈf <mark>ła</mark> mə]	[ˈflämə]	[ˈflämə]	[ˈflämə]
[ˈ <mark>ku</mark> ²l]	['gul <sup>f</sup> ]	[ˈģul <sup>j</sup> ]	[ˈ <mark>g</mark> ul]	[ˈ <mark>ģọ</mark> l]
[ <mark>pıˌäġ</mark> ĭsģ <b>ҟə</b> ğ <sub>i</sub> ]	[ˈ <mark>bäi̯əˈsd͡</mark> təð̪v]	[ <mark>pɪˌāġĭ</mark> ðģrə͡ʃ <sub>ð</sub> ]	[pı, a <sub>j</sub> eĭs¢řə <sub>Q</sub> ,]	[ˌp̊ɑĭəɛɡ̊kə͡gʌ]
[ˈsg̊inə]	[ˈ <mark>s̪g</mark> ʰȩnə]	[ˈ <mark>ski</mark> nə]	[ˈs <mark>ki</mark> nə]	[ˈsɡ̊e̯nə]
[ˈhɛmelit̪]	['heməlit]	[ˈhɛməlit]	[ˈ <mark>h</mark> ʲeməlid̯]	[ˈhɛməlit]
[ˈfi²lm]	[ˈfil <sup>j</sup> m]	[ˈfilm]	[ˈfi <mark>l<sup>i</sup>m]</mark>	[ˈfilm]
[ˈvilə]	[ˈvelə]	[ˈvilə]	[ˈvelə]	[ˈvelə]
[ˈģʰ äi̯]	[ˈġʰäi̯]	[ˈģʰäi̯]	[ˈĝʰ äį]	[ˈġʰäi̯]
['humus]	[ˈhumu <mark>s</mark> ]	[ˈhumu <mark>s</mark> ]	['humus]	['humus]
[viˈ <mark>k</mark> ä <sup>?</sup> ]	['vi <mark>ka</mark> ]	['vi <mark>k</mark> ä]	[viˈġʰɔ]	['viĝ <sup>h</sup> a]

[ˈliu̯li]	[ˈ <mark>l<sup>j</sup>iu̯l</mark> ʲi]	[ˈ <mark>l<sup>j</sup>iu̯l</mark> ʲi]	['l <sup>j</sup> ivl <sup>j</sup> i]	[ˈliu̯li]
[ˈsk <mark>ʁi</mark> ːvə]	[ˈ <mark>s</mark> ģķiːvə]	[ˌs͡kɨ.ʌᢒ]	[ˈsģʁฺiːvə]	[ˈsg̊ʁฺiːvə]
[ˈdäi̯]	[ˈd <mark>a</mark> i	[ˈdäi̯]	['äi̯]	[ˈ <mark>d</mark> äi̯]
[ˈjo <mark>u</mark> ]	[ˈj <mark>u</mark> ð̪ˠ ]	[ˈjǫðᢩˠ, ]	['j <b>o</b> ð̪ˠ,]	[ˈjoᢩðˠ]
[øgʰoloʻgi?]	[øg̀ʰ <mark>əloˈg</mark> i]	[ø <mark>k</mark> ǫloˈģiˀ]	[øgʰ <mark>ołoˈg</mark> i]	[øg̊ʰəlo̥ˈġiˀ]
[dɪˈ <mark>ts</mark> aljə]	[dɪˈd͡saljə]	[ <mark>dı</mark> ˈd͡sa̯ljə]	[ <mark>dı</mark> ˈd͡sa̞ljə]	[d <mark>ıdel</mark> j'ji]
[ˈkiɐ̯səˌb <mark>ea</mark> ̯]	[ˈģʰisəˌ <mark>b<sup>j</sup>e</mark> ɐ̯]	[ˈ <mark>kis</mark> ˌbeɐ̯]	[ˈģʰiɐ̯səb̥ɛɐ̯]	[ˈ <mark>kiʔ</mark> səˌb <mark>e</mark> ɐ̯]
[ˈleŋə]	[ˈ <mark>l<sup>j</sup>eŋg</mark> ə]	[ˈleŋə]	[ˈleŋə]	[ˈleŋə]
[po'pe <sup>?]<sup>j</sup>]</sup>	['b <mark>u p<sup>j</sup>el<sup>j</sup>]</mark>	['bu <mark>pel<sup>j</sup>]</mark>	[ˈbopəl <sup>j</sup> ]	[' <mark>b</mark> oˌb̥ʰel]
[mɐˈ <mark>s</mark> ĝi <sup>ʔ</sup> ]	[mɔ̯ˈs <mark>g̊</mark> ʰi]	[moˈsɡ̊eˀ]	[mɔ̯ˈsɡ̊ə]	[mɔ̯ˈsɡ̊e̯ˀ]
[ˈɡ̊ʰuːlə]	[ˈġʰulə]	['kul <sup>j</sup> e]	[ˈģʰulə]	[ˈģʰulə]
[ˈläməˌ <mark>k</mark> øð̪ˠ]	[ˈläməˌġʰø̯ð̪ˠ]	[ˈläməˌ <mark>k</mark> øð̪ˠ]	[ˈlaməˌġʰøð̪ˠ]	[ˈläməˌġʰøð̪ˠ]
[ˈb̥äŋ <mark>k</mark> ə]	[ˈ <mark>baŋk</mark> ə]	[ˈb̥äŋ <mark>k</mark> ə]	[ˈb̥äŋɡ̊ə]	[ˈb̥äŋ <mark>k</mark> ə]
[ˈsmaʁ <mark>̯ə</mark> ]	[ˈ <mark>s̪mą</mark> ðˠᢩ <b>\$ə</b> ]	[ˈsm <mark>a</mark> ð̪ˠ ឝ͡ᢒ]	[ˈsmað̪ˠ ʁ̥ə]	[ˈsmäð̪ˠ ̯¥ə]
[ˈ <mark>ku</mark> nə]	[ˈġʰ <mark>u</mark> ˈnə]	[ˈģʰ <mark>o</mark> nə]	[ˈġʰ <mark>uː</mark> nə]	[ˈġʰ <mark>uː</mark> nə]
[ˈnesd̥ə]	['nesdə]	[ˈ <mark>n<sup>j</sup>esd</mark> ə]	[ˈnesd̥ə]	['nesdə]
[ˈģäŋə]	[ˈ <mark>g</mark> äŋə]	[ˈģäŋə]	[ˈ <mark>g</mark> äŋə]	[ˈ <mark>g</mark> äŋə]
[ˈģ <mark>u</mark> ð̪ˠ ]	[ˈ <mark>gu</mark> ð̪ˠ]	[ˈģo <mark>ð</mark> ]	[ˈ <mark>gu</mark> ð̪ˠ]	[ˈ <mark>g</mark> oðv̯ː᠈]
[ˈģ <mark>æ</mark> ]	[ˈģœː]	[ˈģ <mark>ŗ</mark> ʌ]	[ˈģœː]	[ˈ <mark>g</mark> Œː]
[ˈmenəsɡ̊ə]	[ˈmen <mark>əs</mark> g̊ə]	['m <sup>j</sup> enəsģə]	[ˈmenesɡ̊ə]	[ˈmenəsɡ̊ə]
[ˈb̥äŋ]	[ˈ <mark>b</mark> äŋ]	[ˈb̥äŋ]	[ˈb̥äŋ]	[ˈb̥äŋ]
[ikoˈkjesv]	[i'Ř <mark>ok<sub>j</sub>e,</mark> Ži <sub>,</sub> V]	[iʁ <mark>̯ok</mark> ʲɪˈsiə]	[ikoˌ <mark>k</mark> jesv]	[irqˈġʰęsʌ]
[b <mark>o</mark> 'liviənsg]	[b <mark>olivi'ɛns</mark> g]	['boliviənsk]	[bolivi'ensĝ]	[boˈliviɛnsg]
[ˈsģoːlə]	[ˈs <mark>gʰu</mark> ːlə]	[ˈs <mark>k</mark> ǫlə]	[ˈsg̊ <mark>u</mark> ːlə]	[ˈsģoːlə]
[ˈɡ̊ʰ <mark>o</mark> ːnə]	[ˈġʰ <mark>uː</mark> nə]	[ˈġʰ <mark>u</mark> nə]	[ˈɡ̊ʰ <mark>u</mark> ːnə]	[ˈġʰoːnə]
[ˈ <mark>p</mark> isģəˌfløːð̪ˠə]	[ˈ <mark>pis</mark> ɡ̊əˌfløːð̪ˠə]	[ˈb̥ʰis <mark>k</mark> əˌfl <mark>ø</mark> ðᢩvə]	[ˌb̥ʰisɡ̊əˈflø̣ːð̪ˠə]	['bʰisɡ̊ə, flø̣ːð̪ˠə]
[ <mark>kɔʁ</mark> əɡ̊ˈ <mark>ts</mark> u'ɐ̯]	[g̊ʰɔ઼̃k <mark>e</mark> g̈́ˌ <mark>tsʊɐ</mark> ̯]	[g̊ʰə̣̥ʁəg̈ˈ <mark>ts</mark> uɐ̯]	[g̊ʰɔ <mark>̥</mark> ʑəg̊ˈd͡suɐ̯]	[g̊ʰɔ <mark>̣ʀe</mark> g̊ˈd͡suɐ̯]
[ˈ <mark>n̪ɔ</mark> ːsǧ]	[ˈnɔ̞ː <mark>s</mark> ̥ĝ]	[ˈn <mark>ə</mark> sġ]	[ˈnɔ̞ːsǧ]	[ˈnɔ <mark>̥</mark> sĝ]
[gä'pi <sup>9</sup> g]	[ <mark>pɐ</mark> ˈb̥ʰiɐ̯]	[þʰɐˈˈb̥ʰiɐ̯]	[þʰäˈþʰiɐ̯]	[päˈpiɐ̯]
[ˈʁ <mark>̃</mark> ɣiːˌ <mark>tə</mark> m²]	[ˈ <mark>ʔi</mark> ˈdʌm]	[ˌ <code>Ri,ˈď<sup>v</sup>ɯ]</code>	[ˌᠷ <mark>i</mark> ˈð <mark>ə</mark> m]	[ˌឝ <mark>i</mark> , dv͡m]
[ˈfɨ <mark>e</mark> ĝ]	[ˈf <mark>ᠯ</mark> ɛġ]	[ˈf <mark>ʁæ</mark> ĝ]	[ˌtɨ <mark>ɛ</mark> g]	[ˈf <mark>re</mark> ĝ]
[ˌŔi <sub>s</sub> 2]	[' <b></b> is]	[ˌřiɛ]	[ˌᠷis]	[' <b>ĸ</b> i?s]
['b̥eɐ̯ <mark>`</mark> ]	[ˈ <mark>b<sup>i</sup>e</mark> ɐ̯]	[ˈŷɛɐ̯]	[ˈy͡ʒd']	[ˈb <mark>̥e</mark> ɐ̯]
[ˈģʰi ɐ̯ġə]	[ˈġʰiɐ̯ġə]	[ˈģʰiɐ̯ɡ̊ə]	[ˈġʰiɐ̯ġə]	[ˈġʰiɐ̯ġə]
[ˈmoɐ̯²]	[ˈm <mark>o</mark> ʔ]	[ˈmoɐ̯]	[ˈmoɐ̯]	[ˈmoɐ̯]
[ˌ <mark>p</mark> ̀ƙesə]	[ˌĥ̥ɨ <mark>ɛ</mark> ɛə]	[ˌĥ̥ʰk͡ɛɛsə]	['ĥŀkasə]	[ˌ <mark>b</mark> kesə]
[ˈlyŋbɣ]	[ˈlyŋbɣ]	[ˈlyŋbɣ]	[ˈlyŋbɣ]	[ˈlyŋb̪y]
[säunə]	[ <mark>s</mark> äun]	[säunə]	[säunə]	[ <mark>s</mark> äunə]
[ˌŔλːᢒ]	[ˈ <mark>?</mark> y·ə]	[ˌŔÀːᢒ]	[ˌŔAː9]	['sue]
[c <mark>oģo</mark> 'lɛːð̪ˠ ə]	[ <mark>∫oĝo</mark> ˈlaːð̪ˠ̥ə]	[¢ <mark>ýðْ 3</mark> l' <mark>çĝ</mark> ə]	[coģoˈl <mark>ạ</mark> ːð̪ˠ ə]	[ <mark>çoĝoˈlɛð̪</mark> ɣə]

[ʁ <mark>ʊˈts</mark> iːnə]	[ʁ <mark>ʊˈˈd͡siːnə]</mark>	[ʁฺuˈt͡siˈnə]	[ʁ <mark>ʊ</mark> ˈd͡siːnə]	[ <mark>ʁuˈtiːnə]</mark>
[ˈb̥äːsĝ]	[ˈ <mark>b</mark> äːsĝ]	[ˈb <mark>̥ɑ</mark> ːsĝ]	[ˈb̥äːsĝ]	[ˈ <mark>b</mark> äːsĝ]
[ˈ <mark>ts</mark> oj]	[ˈd͡ <mark>s</mark> əj]	[ˈ <mark>ts</mark> ʌj]	[ˈd͡sʌj]	[ˈ <mark>ts</mark> ʌj]
[ˈ <mark>çœ</mark> u̯]	[ˈɕə̯u]	[ˈ <mark>ç</mark> ə̯u]	[ˈɕə̯u]	[ˈɕə̯u]
[ʊmˈ <mark>tso</mark> ʲəli]	[ <mark>ə</mark> mˈd͡sɔ̯ləl <sup>j</sup> i]	[ˈømˌ <mark>ts</mark> ɔ̯ləli]	[ <mark>e</mark> mˈd͡sɔ̯ləli]	[em'tsoleli]
['muːliˌvis]	['mu'li'vi <mark>s</mark> ]	['muli'vis]	['muːli'vis]	['mu <sup>?</sup> li'vis]
[ˈɡ̃ɛɐ̯nə]	[ˈġ <mark>e</mark> ɐ̯nə]	[ˈģɛɐ̯nə]	[ˈ <mark>ɡ</mark> ɛɐ̯nə]	[ˈģɛɐ̯nə]
[ˈm <mark>o</mark> ːlə]	[ˈm <mark>o</mark> ˈlə]	['mo <sup>·</sup> le]	[ˈm <mark>o</mark> ːlə]	[ˈmɔ̞ːlə]
[äg̊ʰub̥ʰuŋ ˈ <mark>t͡s</mark> uɐ̯²]	[ <mark>ɐkʊpʊŋˈd͡s</mark> uɐ̯]	[äkupuŋˈ <mark>ts</mark> uɐ̯]	[ɐg̊ʰub̥ʰuŋˈd͡suɐ̯]	[äg̊ʰu'b̥ʰuŋd͡suɐ̯]
[ˈväģ <mark>t</mark> ]	['väģt]	['vä <mark>kt</mark> ]	[ˈväģd]	['väģ <mark>ģs</mark> ]
[ˈkvoːtạ]	[ˈ <mark>kvo</mark> :d̥a̯]	[ˈ <mark>k</mark> vo̞·d̥a̞]	[ˈġʰvọː <mark>d͡s</mark> ạ]	[ˈġʰv <mark>oːd͡sə</mark> ]
[ˈhäu̯nə]	[ˈhäu̯nə]	[ˈhäu̯nə]	[ˈhäu̯nə]	[ˈhäu̯nə]
[ˈhœːnə]	[ˈhœːnə]	[ˈhøːnə]	[ˈh <mark>œ</mark> nə]	[ˈh <mark>œ</mark> ːnə]
[ˈniu̯]	[ˈkniu̯]	[ˈġʰniu̯]	[ˈġʰniu̯]	[ˈĝʰniu̯]
[ˈģʰɔ̯ːwə]	[ˈģʰ <mark>ɔ</mark> ːwə]	[ˈġʰɔ̯ːwə]	[ˈģʰɔ̯ːwə]	[ˈģʰɔ̯ːwə]
[ˈuːə]	[ˈuːə]	['uːə]	[ˈuːə]	[ˈuːə]
[ <mark>ko</mark> ˈʁ̞ub̥t]	[ĝ <sub>p</sub> oˌŧnþď]	[g̊ <sub>µ</sub> oˌᠷ̀np̊d]	[ģʰ <mark>ǫ</mark> ˈʁ̞ub̥ <code>t]</code>	[ĝ <sub>p</sub> oˌʀɔnþt]
[ˌp̊ҟeň]	[ˌp̊ <mark>ɛ</mark> ð]	[ˌp̂ƙeň]	[ˌp̊keň]	[ˌp̂ƙeň]
[ˈ <mark>ts</mark> œgʰleːðə]	[ˈd͡sœɡ̊ʰˌl <mark>e</mark> ːð̪ˠə]	[ˈ <mark>ts</mark> œg̊ʰleːð̪ˠə]	[d͡sœ઼'g̊ʰleːð̪ˠə]	[¦d͡sœ઼ˈg̊ʰl <mark>e</mark> ð̪ˠə]
[ˈmʊːnə]	[ˈmunə]	[ˈmuːnə]	[ˈmunə]	[ˈmɔ̃nə]
[tsiulit]	[ˈ <mark>d͡s</mark> iðᢩv <mark>l</mark> ıt]	[ˈ <mark>ts</mark> iðᢩvlits]	[ˈd͡sið̪ˠlid]	[ˈd͡sið̪vl̯id͡s]
[fycoloˈġi?]	[fy <mark>sioloˈk</mark> i]	[fy <mark>sio</mark> lo'ĝi <sup>?</sup> ]	[fycoloˈģi?]	[fy <mark>si</mark> ǫlọˈģiʔ]
['veĝ]	['v <sup>j</sup> eke]	['vekə]	[ˈv <mark>ɛk</mark> ə]	[ˈveɡ̊ə]
[fo'sge:li]	[fo'sghelji]	[fəˈskeli]	[fʌ̯ˈsɡ̊ɛli]	[fəˈsgeli]
[ˈ <mark>ts</mark> ʁฺyg̊ə]	[ˈ <mark>ts</mark> ʁykə]	[ˈ <mark>ts</mark> ʁฺyg̊ə]	[ˈd͡sʁ̃yg̊ə]	[ˈd͡sʁygə]
[p̊eˌkeð̯ʎ ˈsgɛb̊]	[ˈ <mark>biɣəˌs</mark> ̪gɛp]	['b <mark>akə</mark> 'sgɛb]	[ˈp̊ŧɛð̯ᠯ ˌs͡ặɛb̊]	[p̊ <mark>ə</mark> ˌŘɨ͡d̯ᠯ ˈsgɛp̊]
[bheda'gou?]	[beda, gµoň]	[bʰ ld͡ɕ ˈgouʔ]	[bhede'gou]	[pedäˈģou̯?]
[nɛɐ̯ˈvɜsitəð̪ˠ]	[nəvøsiˈd͡sit]	[nɛɐ̯vøˈsitəd]	failed	[ne v3si ti]
[ˈleːsə]	['leːsə]	['meːsə]	[ˈleːsə]	[ˈleː <mark>s</mark> ə]
[ˈfyˀsisĝ]	[fyˈsisĝ]	[fyˈsisĝ]	[fyˈsi²sĝ]	[ˈfy²sisĝ]
[ <mark>ʊ</mark> ŋˈd̥oֲmɐli]	[ˈuŋd̥əməli]	[ˈuŋˌdəməli]	[ <mark>ʊ</mark> ŋˈdomeli]	[ <mark>ʊ</mark> ŋˈdʌᢩməli]
[ˈģʰø̯sə]	[ˈģʰ <b>y</b> ːsə]	[ˈġʰ <mark>ợ</mark> ·sə]	[ˈģʰysə]	[ˈ <mark>kys</mark> ə]
[ˈģʰø̞ːb̥ə]	[ˈ <mark>k</mark> øːbə]	[ˈɡ̊ʰ <mark>ଡ଼</mark> ˈb̥ə]	[ˈģʰ <mark>y</mark> b̥ə]	[ˈ <mark>ky</mark> b̥ə]
[ˌåɨɕŋ]	[' <b>d?e</b> ŋ]	[ˈd̥ʁaŋ]	[ˌåɨ <mark>ɛ</mark> ม]	[ˈd͡¤aŋ]
[nøðv'vendi]	[nøð̪ˠˌˈ <mark>v<sup>j</sup>end</mark> i]	[nøðv'vendi]	[nøðv'vendi]	[nøðv, vendi]
[b̥ʰed̥a̯ˈɡ̊oˀisk]	[pedv gogisg]	[bhede'gogisg]	[bheda'gouisg]	[bıdɛˈå <mark>o²g</mark> isĝ]
[ˈsmœ²]	[ˈ <mark>s</mark> mæ]	[ˈsmæ͡]	[ˈsmæɐ̯]	[ˈsmæ̯ɐ]
[ˈl <mark>u</mark> ɡ̊ə]	['lukɛ]	[ˈl <mark>u</mark> ģə]	[ˈ <mark>łu</mark> ģə]	[ˈ <mark>łu</mark> g̊ə]
[ˌďŕɔ̇́meďk͡]	[ˈd͡͡͡] [ˈd͡͡]	[ˌd͡rɔ̄məq͡s]	[ˈd̥komədˈ <mark>a</mark> ]	[ˌå <b>r</b> ɔ̀məåɑ̈,]
['b <mark>o</mark> wmest <sub>4</sub> ]	['bəwˌmistə]	['bəw mesdʌ]	[ˌb̥əwˈmisd̪ʌ]	[ˈb <mark>oɐ</mark> ˈmesdʌ]

[ˈneɐ̯meɐ̯]	[ˈ <mark>n<sup>j</sup>eɐ̯mʌ̯ʌ]</mark>	['negmia	) [	'nɛɐ̯mʌ̯	]	[ˈnem <mark>əʁ̞ə</mark> ]
[ˈsd̥ɔːmə]	[ˈsd̥ <mark>ə</mark> ːmə]	[ˈsdᢩɔːmə	) [	ˈsdᢩɔːmə	)]	[ˈsd̥ɔ̞ːmə]
[ˈ <mark>ə</mark> ːnə]	[ˈ <mark>੨</mark> nʌ]	[ˈ <mark>ə</mark> nʌ]	[	ˈ <mark>ə</mark> nʌ]		[ˈʌ̯nə]
R 9	R 10		R 12		R 13	
[bʰa̧nə]	[b̥ʰạːnə]		[b̥ʰạːnə]		[ <mark>p</mark> ạnə]	
[ˈdɣnə]	[ˈd̪yːnə]		[ˈ <mark>d</mark> yːnə]		[ˈ <mark>d</mark> ynə]	
[ˈġʰɛɐ̯łi hẹð̪ˠ]	[ˈġʰɛɐ̯li hẹð̪ɣ]		['åʰɛɐ̯li hẹ	₫Ÿ]	['ĝʰɛɐ̯li]	h <b>ı</b> ðᢩv]
[udsgvk'nemeli]	['udsäg'neməli]		[ <mark>ʊ</mark> d͡s <mark>ɐ</mark> ĝˈnɛı	məli]	[ <mark>ʊd͡sɐ</mark> gˈr	n <mark>eme</mark> li]
[ˈb̥ <mark>ɛð̃</mark> v̥əˌv <mark>eə</mark> lsə]	[ˈb̥ɛːð̪ˠ əˌvɛɐ̯lsə]		[ˈbɛːð̪ˠ ə'v	[ealy3	e <sup>v</sup> õ3å']	v <mark>e</mark> ੲlsə]
[ˈmäŋlə]	[ˈmäŋlə]		[ˈmäŋlə]		[ˈmäŋlə]	]
[ˈf <mark>aʁฺə</mark> ]	[ˈfäːä]		[ˈfäːä]		[ˈfäːä]	
[ˈønsg̊əp]	[ˈo²n sġɛb]		['on <mark>s</mark> ĝe <sup>?</sup> b]	]	['on <mark>s</mark> ĝe	þ]
[ˈ <mark>sofɐ</mark> ]	[ˈsọːfɐ]		[ˈ <mark>s</mark> oːfạ]		['sofɐ]	
[ˌҟ <mark>ɑ</mark> ᠫ]	[ˌҟ <u>ɑ</u> ːɑ]		[ˌҟɑː <mark>ҟ</mark> Ӭ]		[ˌŔ <mark>ɑ̈Ŕ</mark> ᢒ]	
[b̥ɪˈsd̥emə]	[beˈsdemə]		[ <mark>bəˈs</mark> demə	]	[ <mark>bəˈs</mark> der	nə]
[ˈhʌᢩb <mark>e</mark> ]	[ˈhʌᢩbə]		[ˈh <mark>o</mark> b̥ə]		[ˈh <mark>ə</mark> b̥ə]	
[ <mark>d</mark> um]	[dum]		[ <mark>d</mark> um]		[ <mark>d</mark> um]	
[ <mark>s</mark> ġäːþ]	[sġäːþ]		[ <mark>s</mark> gåb]		[sĝäːþ]	
[ˌୡ <sub>p</sub> ɛၗ <mark>ୖ</mark> ğåэҟэ]	[, ầ <sub>p</sub> ɛĸ̆vɛď̥ə' k̈əv̆]		['ầ <sub>p</sub> eksda	, ŕć <mark>ŕ</mark> 9]	['k <sup>j</sup> eķəsq	j́9 <mark>k</mark> ́9]
[ <mark>s</mark> dʻəm]	[sdɔm]		[ <mark>s</mark> d͡ɔm]		[ <mark>s</mark> d̥ɔm]	
['ɛɐ̯'g̊ʰenə]	['ɛɐ̯ˈgʰenə]		[əˈġʰenə]		[eɐ̯ˈĝʰen	ə]
[ˈ <mark>s</mark> ydn]	[ˈ <mark>s</mark> ʲyd̯n]		[ˈ <mark>sy</mark> dn]		[ˈ <mark>s</mark> ʲyd̥n]	
[ˌ <mark>ð</mark> řið̯ʾ ə]	[ˌsģᠷiː፬̯̯̆ ə]		[ˌ <mark>ð</mark> ģŔiː፴Å ð	]	[ˌ <mark>s</mark> ģŕ <mark>i</mark> 졫ź	ə]
[ˈ <mark>g</mark> äfəl]	[ˈģäfəl]		[ˈĝ <mark>a</mark> fəl]		[ˈģ <mark>a</mark> fəl]	
['kinv]	[ˈĝʰiːnɐ]		[ˈ <mark>k</mark> iːnạ]		[' <mark>kinɐ</mark> ]	
[ˈ <mark>xi</mark> ðᢩvə <mark>ð</mark> v]	[ˈhẹðᢩvəd]		[ˈh <mark>e</mark> ðᢩvəd]		[ˈh <mark>e</mark> ðᢩvəd̥	.]
[ˈvạ <mark>s</mark> gəð̪ˠ]	[ˈvạsɡ̊əð̪ˠ]		[ˈvạsɡ̊əð̪ˠ]		[ˈvạsɡ̊əð̪	Y]
[ˈb̥inə]	[ˈb̥enə]		[ˈ <mark>bi</mark> nə]		[ˈ <mark>bi</mark> nə]	
[ <mark>s</mark> ĥɑ̈ˌkɑ̈٫t]	[ɑɡ̊pɑˌkɑ٫d]		[ <mark>s</mark> β̀s, ҟ҅ɑ <sub>,</sub> ぱ]		[ <mark>s</mark> p̀s, k̀aď	<b>ļ</b> ]
[bɐˈg <sup>j</sup> eːcə]	[ç͡]:3ġ́'gġ]		[caː3ġˈ <mark>9d</mark> ]		[bɐˈgaşə	]
[ˈb̥ʰạ̯nə̯ʌ]	[bʰ ạˈniə]		[paˈniʁə]		['pạn <mark>əự</mark> a	)]
[bɪˈslegdsɪð̪v]	[b̥eˈs <mark>l<sup>j</sup>eg̊d̥əðႍv</mark> ]		[ <mark>bɪˈs̪l<sup>j</sup>eģ</mark> də	₫Ÿ]	[ <mark>bı</mark> ˈsleģç	ļəð <sup>y</sup> ]
[ˈfəːˌb <mark>ɪ</mark> ləð̪ˠə]	[fəˈb̥eləð̪ˠ ə]		[ˌfəˈbiləð̪ˠ	[ə]	[ˌf <mark>oˈbi</mark> lə	ðv́ ∍]
[əˈle̞ːnə]	[ạˈlẹːnə]		[əˈl <sup>j</sup> iːnə]		[ <mark>əˈl<sup>j</sup>i</mark> ːnə]	
[ˈmạð̪ˠ ]	[ˈmạð̪ˠ ]		[ˈmạð̪ˠ ]		[ˈmạð̪ˠ ]	
[ˈ <mark>s</mark> inə]	[ˈseːnə]		[ˈ <mark>s</mark> iːnə]		[ˈ <mark>si</mark> nə]	
[ˈfalsg̊]	[ˈfalsg]		[ˈfalsg̊]		[ˈfalsg̊]	
[ˈbuːə]	[ˈb̥uːə]		[ˈb̥uːə]		[ˈ <mark>bu</mark> ə]	
[ˈ <mark>do</mark> ən]	[ˈdɔ̯uən]		[ˈ <mark>d</mark> ̪ə̯uən]		[ˈ <mark>d</mark> ə̯uən]	

[ˈ <mark>g</mark> emə]	[ˈģemə]	[ˈģemə]	[ˈ <mark>g<sup>j</sup>emə]</mark>
[ˈ <mark>gans</mark> ɡ̊ə]	[ˈģạnsġə]	[ˈġ <mark>ans</mark> ġə]	[ˈ <mark>gan</mark> sģə]
[ˈflamə]	[ˈflämə]	[ˈflamə]	[ˈf <mark>la</mark> mə]
[ˈġul <sup>v</sup> ]	[ˈģɔl]	['gul]	['gul <sup>v</sup> ]
[pɪˌ <mark>ŝɛ</mark> jsďŕə፬̯ʌ]	[pໍeˈ <mark>aɑ</mark> ĭsďr̄v̄ <sub>d</sub> ]	[pɪˌşɑisɡ̈kv͡gʌ]	[pı, şajeq kvğ.]
[ˈ <mark>ski</mark> nə]	[ˈsɡ̊e̯nə]	[ˈ <mark>ski</mark> nə]	[ˈske̯nə]
[ˈ <mark>h</mark> ʲeməli]	[ˈ <mark>x</mark> ʲeməlid]	[ˈheməlid̯]	[ˈheməlid]
[ˈfi <mark>l<sup>j</sup>m]</mark>	[ˈfilm]	[ˈfi <mark>l<sup>j</sup>m]</mark>	[ˈfilm]
[ˈvilə]	[ˈvelə]	[ˈvelə]	[ˈvelə]
[ˈġʰai̯ʔ]	[ˈģʰäi̯]	[ˈ <mark>ka</mark> i]	[ˈĝʰ äį]
[ˈhumu <mark>s</mark> ]	['humus]	['humus]	['humus]
['vi_ĝhäː]	['viĝʰäː]	[vɪˈka]	['vi <mark>ka</mark> ]
[ˈ <mark>l<sup>j</sup>iu̯l<sup>j</sup>i]</mark>	[ˈl <sup>j</sup> iu̯l <sup>j</sup> i]	[ˈ <mark>l<sup>j</sup>iu̯l<sup>j</sup>i]</mark>	[ˈ <mark>l<sup>i</sup>iu̯l</mark> ʲi]
[ˈ <mark>s</mark> g̊ʁฺiːvə]	[ˈsg̊ʁฺiːvə]	[ˈ <mark>s</mark> g̊ʁฺiːvə]	[ˈ <mark>s</mark> g̊kivə]
[ˈ <mark>dai</mark> ]	[ˈd̥äi̯]	[ˈdäi̯]	[ˈd̥äi̯]
[ˈj <mark>o</mark> ð̪ˠ]	['jqðᢩ <sup>v</sup> ,]	[ˈjoð̪ˠ, ]	['j <mark>o</mark> ðᢩv,]
[øˈģʰ <mark>olog</mark> i]	[øgʰolojˈġiʔ]	[øgh <mark>əlo</mark> ˈġi]	[əgʰəłoˈgi]
[duˈtaljə]	[d̥eˈd͡saljə]	[dɪˈ <mark>ts</mark> aljə]	[dɪˈd͡saljə]
[ˈ <mark>kˈ</mark> iɹs̪ˌbɛɐ̯]	[ˈġʰiʊ̯səˌb <mark>e</mark> ʊ̯]	[ˈġʰiɐ̯ <mark>s</mark> əˌbɛɐ̯]	[ˈgʰiɐ̯səˌ <mark>b<sup>j</sup>e</mark> ɐ̯]
[ˈleŋə]	[ˈleŋə]	[ˈ <mark>l</mark> ʲeŋə]	[ˈl <sup>j</sup> eŋɡə]
['b <mark>o</mark> ,bʰel <sup>j</sup> ]	['b̥ọˌb̥ʰelʲ]	['boˌbʰel <sup>j</sup> ]	['boˌbʰel <sup>j</sup> ]
[məˈ <mark>sk</mark> i]	[moˈsg̊e²]	[məˈ <mark>s</mark> ki]	[ˌmɔ̯ˈsġi]
['kul <sup>j</sup> e]	[ˈģʰulə]	[ˈ <mark>k</mark> ulə]	[ˈ <mark>k</mark> ulə]
[ˈl <mark>a</mark> məˌġʰøð̪ˠ]	[ˈläməˌġʰøð̪ˠ]	[ˈlaməˌġʰøð̪ˠ]	[ˈ <mark>ła</mark> məˌġʰøð̪ˠ]
[ˈ <mark>b</mark> äŋ <mark>k</mark> e]	[ˈb̥äŋɡ̊ə]	[ˈb̥äŋɡ̊ə]	[ˈ <mark>baŋk</mark> ə]
[ˈ <mark>s</mark> mað̪ˠ kə]	[ˈsm <mark>ä</mark> ð̪ˠ ɨ͡ə]	[ˌ <mark>s</mark> mɑ̯ð̪ʎ kə]	[ˈsm <mark>a</mark> ð̪ˠ ᢩ <b>ʁə</b> ]
was not read	[ˈģʰ <mark>u</mark> nə]	[ˈ <mark>ku</mark> nə]	[ˈ <mark>ku</mark> nə]
[ˈne <mark>s</mark> d̥ə]	[ˈ <mark>n<sup>j</sup>esd</mark> ə]	[ˈne <mark>s</mark> d̥ə]	['nesdə]
[ˈ <mark>g</mark> äŋə]	[ˈģäŋə]	[ˈģäŋə]	[ˈġäŋə]
[ˈģoðᢩˠ,]	[ˈģo²]	[ˈģoðᢩˠ,]	[ˈ <mark>ɡu</mark> ð̪ˠ ]
[ˈġ <mark>æɐ</mark> ̯]	[ˈģœː]	[ˌậ͡硅ː <mark>ጵ</mark> ᢒ]	[ˈ <mark>g</mark> œː]
['m <sup>j</sup> enə <mark>sk<sup>j</sup>e</mark> ]	[ˈme <mark>n<sup>j</sup>e</mark> sġə]	[ˈmenə <mark>s</mark> ge]	['men <mark>ɛ</mark> sg̊ <mark>e</mark> ]
[ˈb <mark>̥a</mark> ŋ]	[ˈb̥äŋ]	[ˈ <mark>b</mark> äŋ]	[ˈ <mark>b</mark> äŋ]
[ikɔˈ <mark>k</mark> jesv]	[iŕó, ĝ <sub>p</sub> és <sup>†</sup> ]	[iŕoˈ, ĝŀċsv]	[iřo, ĝ <sub>p</sub> es9]
[ <mark>bo</mark> ˈliviənsĝ]	[boliviˈʌnsĝ]	[ˈ <mark>bol<sup>j</sup>iviə</mark> nsĝ]	[ˈ <mark>bol<sup>j</sup>iviə</mark> nsģ]
[ˈ <mark>sku</mark> ːlə]	[ˈsg̊uːlə]	[ˈ <mark>s</mark> g̊uːlə]	[ˈs <mark>g̊ʰu</mark> lə]
[ˈ <mark>ku</mark> ːnə]	[ˈĝʰ <mark>u</mark> ːnə]	[ˈ <mark>ku</mark> ːnə]	[ˈ <mark>ku</mark> nə]
[ˈ <mark>pis</mark> kəˌf <mark>łv</mark> ð̪ˠə]	['bʰisɡ̊əˌfløːð̪ˠə]	[ˌb̥ʰis̪ɡ̊əˈfløːð̪ˠə]	[ˈb̥ʰisɡ̊əˈfløːð̪ˠə]
[ <mark>kə઼</mark> ¥əg̈́ˈ <mark>ts</mark> uɐ̯]	[g̊ʰɔ઼ <mark>ਞ</mark> əg̊ˈd͡suɐ̯]	[g̊ʰəəg̈ˈ <mark>ts</mark> uɐ̯]	[g̊ʰɔˈʁɛg̊ <mark>ds͡ʊ</mark> ]

[ˈnoːs̪ĝ]	[ˈn <mark>ə</mark> sĝ]	[ˈn <mark>əٖs</mark> ɡ̊]	[ˈnə̥sĝ]
[þʰäˈþʰiɐ̯]	[b̥ʰ <mark>ạ</mark> ˈb̥ʰiɐ̯]	[ <mark>p</mark> ä'piɐ̯]	[ <mark>p</mark> ä'bʰiɐ̯]
[ˌᠷiːˈ <mark>d</mark> ᠯu]	[ˌҟ <mark>ı</mark> ˈåv͡ɯ]	[ˌᠷiːˈðəm]	[ˈʁ <mark>ɨ</mark> ˈdʌm]
[ˈf <mark>ɛ</mark> g]	[ˈfɨ <mark>e</mark> ĝ]	[ˈf <mark>ɛ</mark> g]	[ˈf <mark>ɛ</mark> g]
[ˌři]	[ˌři <sub>y</sub> 2]	[ˌҟi <mark>ಶ</mark> ]	[ˌřiɛ]
[ˈy͡ʒd']	['b <mark>e</mark> ɐ]	[ˈ <mark>b<sup>i</sup>e</mark> ɐ̯]	[ˈb <sup>j</sup> eɐ̯]
[ˈġʰiɐ̯ġʌ̯]	[ˈģʰiɐ̯ġə]	[ˈ <mark>k</mark> iɐ̯ɡ̊ə]	[ˈģʰiɐ̯ɡ̊ə]
[ˈmoɐ̯]	['moɐ]	[ˈmoɐ̯]	[ˈmoɐ̯]
[ˌĥ <sub>ŀ</sub> ƙ <mark>ɛ</mark> ᢒ]	[ˌĥ <sub>p</sub> k͡ɛsə]	[ˌ <mark>b</mark> ƙa <mark>z</mark> ə]	[ˌĥŀƙa <mark>s</mark> ə]
[ˈlyŋbɣ]	[ˈlyŋbɣ]	[ˈlyŋbɣ]	[ˈlyŋ <mark>b</mark> y]
[säunə]	[säunə]	[ <mark>s</mark> äunə]	[ <mark>s</mark> äun]
[ˌᠷ <mark>̀n</mark> ːᢒ]	[ˌŔÀ:9]	[ˌŔÀː9]	[ˌŔ <mark>٨</mark> ᡬ]
[∫oģoˈląːðၘə]	[∫oĝoˈlạːð̪ˠə]	[ <mark>şoģoˈlạ</mark> ːð̪ˠ ə]	[ <mark>şoģoˈlaːð̪ˠ</mark> ə]
[ʁฺuˈtiːnə]	[ʁ <code>ʊˈd͡siːnə]</code>	[ʁฺuˈd͡siːnə]	[ʁฺuˈ <mark>t</mark> iːnə]
[ˈb̥äː <mark>s</mark> ĝ]	['b̥äːsĝ]	[ˈb̥ <mark>as</mark> k]	[ˈ <mark>b</mark> äːsĝ]
[ˈt͡sɔj]	[ˈd͡͡sʌj]	[ˈ <mark>tsə</mark> j]	[ˈ <mark>ts</mark> ʌj]
[ˈ <mark>ɛj</mark> ɔu̯]	[ˈɕ <sup>i</sup> ə̯u]	[ˈsœu̯]	[' <mark>ç</mark> өµ]
[em'tsojlel <sup>j</sup> i]	[ˈømˌd͡sə̯ləli]	[əmˈtsoləl <sup>j</sup> i]	[əmˈd͡sə̯ləl <sup>j</sup> i]
['muː <mark>lj</mark> ivi <mark>s</mark> ]	['mu²li'vis]	[ˈmul <sup>j</sup> ɪvɪs̪]	['mułıvıs]
[ˈ <mark>ge</mark> ɐ̯nə]	[ˈģɛɐ̯nə]	[ˈģɛɐ̯nə]	[ˈġ <mark>e</mark> ɐ̯nə]
[ˈmɔ̃ːlə]	[ˈmɔ̞ːlə]	[ˈmɔ̞ːlə]	[ˈm <mark>ə</mark> ːlə]
[äg̊ʰub̥ʰuŋˈd͡suɐ̯]	[äg̊ʰub̥ʰuŋˈd͡suɐ̯]	[ekopoŋ'tsuɐ]	[ <mark>ɐkʊpʊŋˈd͡s</mark> uɐ̯]
['väĝd]	[ˈväģd]	[ˈväģd]	[ˈväģd]
[ˈġʰvotɐ]	[ˈġʰv <mark>oːt</mark> ə]	[ˈ <mark>kvo</mark> ːd̥a̯]	['kvoːt͡sɐ]
[ˈhäu̯nə]	[ˈhäu̯nə]	[ˈ <mark>x</mark> äu̯nə]	[ˈhäu̯nə]
[ˈ <mark>x<sup>j</sup>u</mark> ːnə]	[ˈhøːnə]	[ˈh <mark>y</mark> ːnə]	[ˈh <mark>œ</mark> ːnə]
[ˈġʰniu̯ʔ]	[ˈĝʰniu̯]	[ˈ <mark>k</mark> niu̯]	[ˈ <mark>g</mark> niu̯]
[ˈ <mark>ko</mark> wə]	[ˈģʰɔ̯ːwə]	[ˈ <mark>k</mark> ɔːwə]	[ˈ <mark>kɔ</mark> ːwə]
['uːə]	['uːə]	[ˈuːə]	[ˈ <mark>u</mark> ːə]
[g̊ <sub>p</sub> oˌŧnþ̊d]	[g̊ʰ <mark>ɐ</mark> ˈʁ̞ub̥ <code>t]</code>	[ <mark>ko</mark> ˌҟnþ̊d]	[ <mark>ko</mark> ˌŧnþ̊d]
[ˌp̂ƙeň]	[,p̂keň]	[ˌĥŧeñ]	[ˌp̊ၨ <mark>ɛ</mark> n̆]
[ˈ <mark>ts</mark> œɡ̊ʰlɛːð̪ˠə̞]	[ˈd͡sœᢩɐg̊ʰleːð̪ˠə]	[ˈ <mark>ts</mark> œɡ̊ʰˌleð̪ˠə]	[ˈd͡͡sœ̃ġʰˌl <mark>e</mark> ðᢩvə]
[ˈmunə]	[ˈmunə]	[ˈmunə]	[ˈmunɛ]
[ˈ <mark>ts</mark> iðlit]	[ˈd͡sið̪ˠlʲt]	[ˈ <mark>ts</mark> iðᢩvl̥ <sup>j</sup> ɪd̥]	[ˈ <mark>d͡s</mark> iðᢩˠ <mark>l</mark> ʲɪd̯]
[fy <mark>zio</mark> 'logi]	[fy <mark>siolo</mark> ˈġiʔ]	[fy <mark>si</mark> ǫlọˈģi²]	[fy <mark>si</mark> olo'gi <sup>?</sup> ]
[ˈveɡ̊ə]	[ˈ <mark>v</mark> ʲeɡ̊ə]	[ˈveɡ̊ə]	['v <sup>j</sup> eke]
[f <mark>oˈş</mark> ģeli]	[foˈsġeli]	[fəˈ <mark>s</mark> g̊el <sup>j</sup> i]	[fo'sghelji]
[ˈd͡sʁ̞y <mark>k<sup>j</sup>e</mark> ]	[ˈd͡sʁٜyg̊ə]	[ˈd͡sʁ̞yɡ̊ə]	[ˌ <mark>ī</mark> kʌ <mark>k</mark> ə]
[ <mark>pə</mark> ˌጵiďˈ <mark>ಶ</mark> ͡ਫɛ͡p]	[p̊eˌᠷ <mark>ɨ</mark> ᡚᢩᡝ ˈsǥ̃ɛb]	[ <mark>pɛ</mark> ˌጵɛð̪ᠯ ˈ <mark>s</mark> ͡ਫɛp̊]	[ <mark>p១</mark> ˌጵɛ፬̯ʎˈˈឧឆ <mark>ၞəb</mark> ]
[pedv'gou]	['bʰed̥ <mark>ɐ</mark> ,g̊ou̯']	[pedˈɐˈˈgo²u̯]	[peds¦gou?]

	-	-	-
[ner vøsi ded]	[nı <sup>°</sup> vøsi'dse?d]	[nəvøsi'dse'd]	[nəvøsi'dsed]
[ˈleː <mark>s</mark> ə]	[ˈleːsə]	[' <mark>l<sup>j</sup>e:s</mark> ə]	[ˈ <mark>l<sup>j</sup>e</mark> ·sə]
[ˈfysisĝ]	[fyˈsisĝ]	[ˈfysisĝ]	[fyˈsisĝ]
[ <mark>ʊŋģ</mark> ˈd̪ʌməli]	[ˈ <mark>o</mark> ŋdəməli]	[ʊŋˈ <mark>d</mark> əməli]	[ʊŋˈ <mark>d</mark> əməli]
[ˈ <mark>kys</mark> ə]	[ˈģʰøsə]	[ˈ <mark>kys</mark> ə]	[ˈ <mark>ky</mark> sə]
[ˈ <mark>kø</mark> bə]	[ˈģʰøːb̥ə]	[ˈ <mark>ky</mark> ːb̥ə]	[ˈ <mark>kø</mark> bə]
[ˈd̥kaŋ]	[ˈd̥ʁaŋ]	[ˌďřau͡,]	[ˌďřeû]
[nødˈvendjʲi]	[nøð <sup>v</sup> , ve <sup>2</sup> ndi]	[nøˈvendid]	[nøðǐ vjendi]
[bhedr'gogisg]	[bhedr'go'uisg]	[pedv'go <sup>?</sup> isg]	[pedv'gogisg]
[ˈ <mark>sm<sup>j</sup></mark> əɐ̯]	[ˈsmæ]	[ˈ <mark>s</mark> mæ]	[ˈsmæɐ̯]
[ˈ <mark>łuk</mark> ə]	[ˈ <mark>łu</mark> g̊ə]	[ˈ <mark>łu</mark> g̊ə]	[ˈl <mark>y</mark> ĝɛ]
[ˈ <mark>q</mark> ͡kɔ̀meˌ <mark>q͡ɛk</mark> ]	[ˌģጵɔૅməɡ <mark>͡s</mark> ]	[ˌåk͡ɔm <mark>e</mark> ၛႆၷ]	[ˈtk͡ɔmeqs]
[ˈbəwˌme <mark>s</mark> dʌ]	[ˈb̥əwˌmesd̥ʌ]	[b̥o̯wˈmisd̪ʌ]	['b̥əwmes <mark>tə</mark> ]
[ˈnɛʊ̯mʌ̯ʌ]	[ˈnɛɐ̯mʌ̯ʌ]	[ˈnɛɐ̯mᠯ <mark>k</mark> ə]	['neğme <mark>k</mark> v]
[ˈsd <mark>əᢩv</mark> mə]	[ˈsd̥ɔ̞ːmə]	[ˈ <mark>s</mark> d̥ɔ̞ːmə]	[ˈsd <mark>̥ə</mark> ːmə]
[ˈ <mark>ə</mark> nʌ]	[ˈʌʌnʌ]	['onʌ]	['onə]

### Appendix 20. Transcriptions of the vowel and consonant targets in the Dgroup

D 2	D 3	D 4	D.C
	D 3	D 4	D 5
[ <mark>p</mark> ạnə]	[ <mark>pạṇ</mark> ə]	[ <mark>paṇ</mark> ə]	[ <mark>p</mark> ạnə]
[ˈ <mark>d</mark> ynə]	[ˈ <mark>d</mark> y <u>n</u> ə]	[ˈ <mark>dyn</mark> ə]	[ˈ <mark>d</mark> ynə]
[ˈġʰɛɐ̯liˌhẹ̯ð̪ˠˀ]	[ˈ <mark>k</mark> ɛɐ̯liˌhəð̪ˠ]	[ˈ <mark>k</mark> ɛɐ̯liˌhɐ̯ð̪ˠ]	[ˈĝʰɛɐ̯liˌh <mark>ɪ</mark> ðᢩˠ]
[utäg'nemeli]	[utäg'nemli]	[u <mark>d</mark> ägˈneməli]	[ <mark>v</mark> ds͡ <mark>ɐ</mark> gˈnɛməli]
['bɛːð̪ˠ əˌvɛɐ̯lsə]	[ˈ <mark>be</mark> ð̪ˠ əˌvɛl <mark>s</mark> ə]	[ˈ <mark>ba</mark> ð̪ˠ vɛls̪ə]	[ˈ <mark>bɛð</mark> ˠəˌveɐ̯l <code>sə]</code>
[ˈm <mark>a</mark> ŋlə]	[ˈmäŋ <sup>g</sup> lə]	[ˈmäŋglə]	[ˈm <mark>a</mark> ŋlə]
[ˈfäːä]	[ˈf <mark>aə</mark> ]	[ˈfāːä]	['fäːä]
[ˈo̯nsɡ̊əb̥]	['o <mark>nˌs</mark> ġạp]	[ˈ <mark>yn</mark> ˌs <mark>kę</mark> b̥]	[ˈo̯nsɡ̊əb̥]
[ˈso <sup>ʊ</sup> ːfə]	[ˈso·fɐ]	[ˈso <sup>ə</sup> ːfɐ]	['sofɐ]
[ˌҟ <mark>ɑ</mark> ᢒ]	[ˌҟɑː <mark>&gt;</mark> ]	[ˌҟ <mark>ช้Ӭ</mark> ]	[ˌħɑː <mark>9</mark> ]
[b̥ɪˈsd̥emə]	[þɛˈdɛmɛ]	[ <mark>bə</mark> ˈstemə]	[ <mark>bı</mark> ˈsd̥emə]
[ˈhʌᢩbə]	[ˈh <mark>opɛ</mark> ]	[ˈhʌᢩbə]	[ˈhʌᢩbə]
[ <mark>d</mark> um]	[dum]	[ <mark>d</mark> um]	[ <mark>du</mark> m]
[sġäːþ]	[sģäːp]	[sģäːþ]	[sắ <mark>ä</mark> þ]
[ˈ <mark>k</mark> ɛɐ̯sd̥ <mark>əʌ</mark> ]	[kɛɐ̯sd̥əə]	[ˌḁ̂̀ʲɛၗၴၩၛႆጛႜ႞ᠯ]	[ˈ <mark>k<sup>j</sup>e</mark> ɐ̯sd̥ə <mark>e</mark> ᢩʌ]
[sd̥ə̣m]	[ <mark>s</mark> dom]	[stəm]	[sdɔm]
[ˈcpɑ̂hopo]	['spkeno]	[ɛɐˈˈøʰeŋə]	[əˈk <sup>j</sup> enə]
	<pre>[panə] ['dynə] ['dynə] ['gʰɛ¤li,hẹðˠ?] [utäġ'nemeli] ['bɛːðˠ,ə,vɛ¤lsə] ['maŋlə] ['fä:ä] ['qnsġəb] ['so°:fə] ['käə] [bı'sdemə] ['h,bə] [dum] [sġä:b] ['kɛ¤sdəʌ] ['amöhənə]</pre>	[pạnə]       [pạnə]         ['dynə]       ['dynə]         ['gʰɛɣli,hẹð̪ˠ,²]       ['kɛɣli,həð̪ˠ,]         [utäġ'nemeli]       [utäġ'nemli]         [utäġ'nemeli]       [utäġ'nemli]         ['bɛːð̪ˠ,ə,vɛɣlsə]       ['beð̪ˠ,ə,vɛls̪ə]         ['bɛːð̪ˠ,ə,vɛɣlsə]       ['beð̪ˠ,ə,vɛls̪ə]         ['mạŋlə]       ['mäŋs <sup>°</sup> lə]         ['maŋs²lə]       ['fạə]         ['oŋsġəb]       ['oŋ,sġap]         ['sö <sup>°</sup> :fə]       ['sö'fv]         ['yäa]       ['yä:ə]         ['yäa]       ['kɛï:ə]         [bi'sdemə]       [bɛ'dɛmɛ]         ['h,bə]       ['hopɛ]         [dum]       [dum]         [sġä:b]       [sġä:p]         ['kɛɣsdəʌ]       [kɛɣsdəə]         [sdəm]       [sddm]	[pạnə]       [pạnə]       [pạnə]         ['dynə]       ['dynə]       ['dynə]         ['gʰɛgli,hẹð̪ˠ]       ['kɛgli,həð̪ˠ]       ['kɛgli,hẹð̪ˠ]         [utäġ'nemeli]       [utäġ'nemli]       [udäġ'neməli]         [utäġ'nemeli]       [utäġ'nemli]       [udäġ'neməli]         ['bɛːð̪ˠ,ə,vɛglsə]       ['beð̪ˠ,ə,vɛls̪ə]       ['bạð̪ˠ,vɛls̪ə]         ['mɨŋlə]       ['mäŋglə]       ['mäŋglə]         ['mɨŋlə]       ['mäŋglə]       ['mäŋglə]         ['mɨŋlə]       ['gŋ,gǧậŋ]       ['mäŋglə]         ['gosġəb]       ['gŋ,gǧậŋ]       ['mäŋglə]         ['sö <sup>0</sup> :fə]       ['sö'fɛ]       ['sö <sup>3</sup> :fɛ]         ['yäə]       ['yä:ə]       ['yäə]         ['yäə]       ['yä:ə]       ['yɨ]         ['yäə]       ['bɛ'dɛmɛ]       [bə'stemə]         ['h,bə]       ['hopɛ]       ['h,bə]         [dum]       [dum]       [dum]         [sģäːb]       [sġäːp]       [sġäːb]         ['kɛgsdəa]       ['ġʰɛɛɡsdəʁʌ̯]       ['ġʰɛɛɡsdəʁ̯ʌ̯]         ['kɛɡŋ]       [ɣdom]       [stəm]

[ˈsø <mark>te</mark> n]	[ˈsød̥n]	[ˈ <mark>s</mark> ʲydn]	[ˈ <mark>s̪ʏt</mark> ən]	[ˈs <mark>y</mark> dn]
[ˈst <mark>ṛ</mark> iː <b>ṛ</b> ə]	[ˈs <mark>t</mark> kijə]	[ˌ <mark>ð</mark> ģ祚 <u>i</u> ðʌə]	[ˈs <mark>tɪi</mark> ð̪ˠ ə]	[ˌsďr <mark>ɨ</mark> ð̯ːə]
[ˈ <mark>g</mark> äfəl]	[ˈ <mark>g</mark> äfl]	[ˈģäfəl]	[ˈģäfəl]	[ˈ <mark>ġa</mark> fəl]
[ˈ <mark>k</mark> iːna]	[ˈĝʰiːnə]	[' <mark>kin</mark> ɐ]	[ˈġʰiː <code>ŋɐ]</code>	[ˈġʰinə]
[ˈh <mark>eð</mark> ət]	[ˈhẹðᢩv]	[ˈh <mark>e</mark> ð̪ˠəd̥]	[ˈhẹð̪ˠət]	[ˈhẹ̃ðᢩvəd̥]
[ˈv <mark>a</mark> skət]	[ˈvạsġəð̪ˠ]	[ˈvḁs̪ɡ̊əd̯] dialect	[ˈv <mark>ask</mark> ət]	[ˈvạ <mark>s</mark> ġəð̪ˠ]
[ˈbinə]	[ˈb̥inə]	[ˈb <mark>in</mark> ə]	[ˈb̥enə]	[ˈb̥inə]
[ <mark>s</mark> þaˌkaď]	[ <mark>s</mark> þəˌŧåģ]	[ <mark>aþa</mark> ˌ <mark>ƙa</mark> ၛ]	[ɑ͡þɑ̈ˌkɑ̈t]	[ <mark>sþs</mark> ,řgď]
[b <mark>aˈga</mark> ːɛə]	[b̥ɐˈɡaʒː]	[ç <mark>ɛˈsɛːʒə</mark> ]	[b̥ɐˈɡ <mark>ɛːʒ</mark> ]	[cə <mark>3</mark> ğ' <mark>yd</mark> ]
['bʰɛnɛɛ]	[pạˈnęʌ]	[paˈneɐ]	[b̥ʰ aˈn <mark>e</mark> ʌ]	['pɐni²ʌ]
[bɛˈ <mark>sɬɛgt̪ɛt</mark> ]	[b̥ɪˈslejd̥əd̥]	[b̥ɛˈ <mark>s</mark> leg̊d̥ə <mark>d</mark> ]	[bɪˈslag̊tət]	[vႍðeþĝ3l <mark>ː</mark> ːd]
[ˌfəˈbilədə]	[fəˈb̪eləð̪ˠ ə]	[f <mark>ə̯ˈbi</mark> ləð̪ˠə]	[ˈfɔːˌbeləð̪ˠ ə]	[ˈf <mark>ob̥ı</mark> ləð̪ˠ ə]
[əˈlinə]	[əˈle̞ːnə]	[əˈl <sup>j</sup> in̪ːə]	[aˈleːnə]	[əˈliːnə]
['mạ <mark>l</mark> <sup>j</sup> ]	[ˈməð̪ˠ ]	[ˈmɛð̪ˠ ]	['mạ ð̪ˠ ]	[ˈmąð̪ˠ ]
[ˈsiˈnə]	[ˈsɐ̯nə]	[ˈ <mark>s</mark> iː <u>n</u> ə]	[ˈseːnə]	[ˈ <mark>s</mark> eːnə]
[ˈfalsk]	[ˈfalsg̊]	[ˈfalsg̊]	[ˈfalsg̊]	[ˈfalsĝ]
[ˈbui]	[ˈ <mark>b</mark> uə]	[ˈbuː <mark>ɛ</mark> ]	[ˈb <mark>̥u</mark> ə]	[ˈ <mark>bu</mark> ˀ]
[ˈ <mark>dou</mark> n]	[ˈ <mark>d</mark> oən]	[ˈd̥ə̯n]	[ˈ <mark>d</mark> ə̯uən]	[ˈ <mark>d</mark> ou̯n]
[ˈgeːmə]	[ˈg͡ɛmə]	[ˈģemə]	[ˈģemə]	[ˈ <mark>g</mark> ʲemə]
[ˈ <mark>gä</mark> ˈns <mark>k</mark> ə]	[ˈ <mark>g</mark> ạns <mark>k</mark> ə]	[ˈˈġ <mark>aŋs</mark> ġə]	[ˈġ <mark>a</mark> nsġə]	[ˈ <mark>gạ</mark> nsɡ̊ə]
[ˈfl <mark>ɛ</mark> ·m <sup>j</sup> ə]	[ˈflämə]	[ˈflämə]	[ˈfləmə]	[ˈflämə]
[ˈgulɣ]	[ˈ <mark>g</mark> ə̯l]	[ˈģ <mark>ʉl<sup>j</sup></mark> ]	[ˈģul <sup>f</sup> ]	[ˈgul]
[ <mark>pɪˈgɑi</mark> stʁ <mark>əț</mark> ]	[ˌpɑjəstkət]	[beˈgäi̯strəț̪]	[ <mark>pıˌāġisģis</mark> ði]	[pı, agisqkvğ <sub>i</sub> ]
[ˈsg̊inə]	[ˈsģenə]	[ˈ <mark>skin</mark> ə]	[ˈskenə]	[ˈ <mark>ski</mark> nə]
[ˈh <mark>i</mark> m <sup>j</sup> eli ]	[ˈhɛməlid]	['hemlit]	[ˈh <mark>ɛmɛlit</mark> ]	[ˈheməlid]
[ˈfilm]	[ˈfilm]	[ˈfilm]	[ˈfilm]	[ˈfilm]
['vil <sup>·</sup> e]	[ˈvelə]	[ˈvelə]	[ˈvelə]	[ˈvilə]
[ˈġʰ <mark>ɛ</mark> į]	[ˈ <mark>k</mark> äi̯]	[ˈ <mark>k</mark> äi̯ʔ]	[ˈĝʰ äį]	[ˈ <mark>k</mark> äi̯]
['humus]	['humus]	['humus]	['humus]	[ˈxumus]
[viˈġʰä·]	[viˈkä]	[viˈ <mark>ka</mark> ]	[viˈ <mark>ka</mark> ]	[viˈ <mark>k</mark> ä²]
[ˈlivli]	[ˈliu̯li]	['l <sup>j</sup> iul <sup>j</sup> i]	[ˈliu̯li]	[ˈlʲiu̯lʲi]
[ˈskɪiwə]	[ˈsģ <mark>ʁ</mark> ɨvə]	[ˈ <mark>s</mark> kˠɨːvə]	[ˈs <mark>k</mark> ʁฺɨvə]	[ˌ <mark>ð</mark> §Ŕŗːň]
[' <u>dei</u> ]	[ˈ <mark>d</mark> äi̯]	[ˈdai̯]	[ˈd͡ɛi̯]	[ˈ <mark>d</mark> äi̯]
[ˈjoðᢩˠ]	[ˈjoð̪ˠ]	['joðᢩ¥]	[ˈjəð̪ˠ ]	[ˈjoᢩðᢩv,]
[ <mark>əko</mark> lo'gi]	[øgʰo̯lo̯ˈġi]	[ <mark>ə</mark> ģʰo̯loˈġi]	[ˌøɡ̊ʰo̯lo̯ˈgi]	[øåholo'gi?]
[d̪ɪˈt̪aljə]	[dɪˈt̪aljə]	[d̪ɪˈt̪ɛljə]	[d̥ɪˈd͡sa̯ljə]	[ <mark>d</mark> ɪˈd͡͡sa̯l <sup>j</sup> jə]
[ˈkisəˌbɛ <sup>j</sup> a]	[ˈġʰiʊ̯səˌb̥ɛv̯]	[ˈɡ̊³d̥ <code>aˌbɛɐ</code> ]	[ˈġʰisəˌbɛɐ̯]	[ˈģʰiɐ̯səˌ <mark>b<sup>j</sup>e</mark> ɐ̯]
[ˈ <mark>ᠯ</mark> ɛŋə]	[ˈleŋə]	[ˈlɛŋɡə]	['leŋgɛ]	[ˈłeŋə]
['bopel]	[ˌb̥oˈ <mark>p</mark> el]	[po'pel]	['b <mark>o'p</mark> el]	['bobhelj]
[mɔ̯ˈski]	[mɔ̯ˈsɡ̊əˀ]	[moˈ <mark>ske</mark> ]	[mɔ̯ˈsɡ̊əˀ]	[mɔ̯ˈsɡ̊eˀ]

[ˈģʰulə]	[ˈ <mark>k</mark> ulə]	['kul <sup>j</sup> e]	[ˈģʰulə]	[ˈģʰulə]
[ˈ <mark>ł</mark> am <sup>j</sup> eˌkøl]	[ˈläməˌġʰø̯ð̪ˠ]	[ˈläməˌġʰø̯ð̪ˠ]	[ˈläməˌġʰø̯ð̪ˠ]	[ˈ <mark>ła</mark> məˌġʰợ̯ð̪ˠ]
['baŋ <mark>k<sup>j</sup>e</mark> ]	[ˈb̥äŋ <mark>k</mark> ə]	[ˈb̥äŋ <mark>k</mark> ə]	[ˈˈb <mark>̥a</mark> ŋg̊ə]	[ˈ <mark>ba</mark> ŋkə]
[ˈsmąð̪ˠ ɟə]	[ˈsmąð̪ˠ ឝə]	[ˈ <mark>sma<sup>ð</sup>ʁə</mark> ]	[ˈsməð̪ˠ ឝə]	[ˈsmąð̪ˠ ម៉ <mark>ə</mark> ]
[ˈ <mark>ku</mark> nə]	[ˈ <mark>ku</mark> nə]	[' <mark>k</mark> unɛ]	[ˈ <mark>kun</mark> ə]	[ˈ <mark>k</mark> unə]
[ˈ <mark>'nɛstə</mark> ]	[ˈnesd̥ə]	[ˈ <mark>n</mark> esdə]	[ˈnesd̥ə]	[ˈnesd̥ə]
[ˈ <mark>gäng</mark> ə]	[ˈģäŋ <b>!</b> ]	[ˈģäŋ <mark>g</mark> ə]	[ˈģäŋ <mark>g</mark> ə]	[ˈ <mark>g</mark> äŋə]
[ˈ <mark>go<sup>u</sup>]</mark>	[ˈ <mark>ġo</mark> ðᢩŸ	['ko <sup>u</sup> ]	['ĝo]	[ˈ <mark>gu</mark> ð̪ˠᢩ]
['gœ]	['ģ <b>œ</b> ː]	[ˈģ <mark>æ</mark> ə]	['ĝ <b></b> []	[ˈ <mark>g</mark> Œː]
[ˈmenəsɡ̊ə]	[ˈmenəsɡ̊ə]	[ˈmʲeneskə]	[ˈmensɡ̊ə]	[ˈmenəsɡ̊ə]
['baŋk]	[ˈ <mark>b</mark> ạŋ]	[ˈb̥äŋ]	[ˈb <mark>̥aŋk</mark> ]	[' <mark>b</mark> äŋ]
[iro'keza]	[iko, g <sub>µ</sub> esv]	[īkoˌke. <mark>sə</mark> ]	[ioˈgʰesə]	[ikoˈkjesv]
[boliviˈansk]	[bolivi'ensg]	[bolivi'en <mark>sk</mark> ]	[þoliviˈ <mark>ä</mark> nsk]	[þ <mark>ol<sup>j</sup>iviˈɛn</mark> sǧ]
[ˈsk <mark>ə</mark> lə]	[ˈs <mark>k</mark> ələ]	[ˈ <mark>sko</mark> ̞·lə]	[ˈsɡ̊ <mark>o</mark> ̥·lə]	[ˈs <mark>ɡ̊ʰo</mark> ːlə]
[ˈ <mark>kə</mark> nə]	[ˈ <mark>kə</mark> nə]	[ˈ <mark>koːn</mark> ə]	[ˈġʰœːnə]	[ˈĝʰoːnə]
[ˈpiskəfløl <sup>j</sup> ]	[ˈ <mark>p</mark> isɡ̊əˌflø̞ːð̪ˠə̞]	[ˈ <mark>pis</mark> kəˌfløːð̪ˠə]	['b̥ʰisɡ̊əˌflø̥·ð̪ˠə]	[ˌb̥ʰisɡ̊əˈfløːð̪ˠ;ə]
[g̊ <sup>h</sup> ə̣̯ʁəg̈ˈtuɐ̯]	[ <mark>kວ</mark> əgˈˈ <mark>t</mark> uɐ̯]	[ <mark>kokə</mark> k tnš]	[g̊ʰɔ <mark>ၨҟ</mark> əg̈ˌd͡snʁ̆]	[g̊ʰɔˈeg̊ <mark>ds</mark> u?]
[' <mark>no</mark> 'sk]	[ˈnəsġ]	[ˈ <mark>nos</mark> k]	[ˈn <mark>o</mark> ˈsĝ]	[ˈn <mark>ə</mark> sĝ]
[þʰäˈþʰi <sup>ª</sup> ]	[papiy]	[paˈpiɐ̯]	[paˈpiɐ̯]	[ <mark>p</mark> äˈ <mark>þ</mark> iɐ̯]
[ˌʀi.ˈ <mark>d</mark> om]	[ˈ <mark>ʁid̥ə</mark> m]	[ˌkɨˈďvɯ]	[ˈɹiːd̥əm]	[ˈʁ <mark>i</mark> ˈd͡v͡m]
[ˈfʁ <mark>ɛk</mark> ]	[ˈf <mark>k</mark> ĝ]	[ˈfʁ <mark>ɛ</mark> ĝ]	['fřa <sub>Å</sub> ]	[ˈfʁ̞aġ]
[ˈ <mark>r<sup>j</sup>is]</mark>	[ˌři,z]	[ˌᠷඁi <mark>Ⴧ</mark> ᢄ]	[ˌŔ <u>i</u> ːɛ]	[ˌři <sub>s</sub> 2]
[ˈb <mark>a</mark> ɐ̯]	[ˈy͡əd']	[ˈb̥ɛɐ̯]	[ˈɣɜɣˈ]	[ˈ <mark>ˈyʒd</mark> ']
[ˈģʰi ɐ̯kə]	[ˈ <mark>kiʊ̯k</mark> ə]	[ˈ <mark>kik</mark> ə]	[ˈģʰi ɐ̯ɡ̊ə]	[ˈģʰiɐ̯ɡ̊ə]
['m <mark>o</mark> ']	[ˈmoɐ̯]	[ˈmoɐ̯]	[ˈm <mark>o</mark> ɐ̯]	[ˈmoɐ̯]
[ˈĥ̥ʰk͡ <mark>e</mark> sə]	[ˈb̥ʰʁ̞asə]	[ˌβ̥ʰҟɛ <mark>ɐ</mark> ə]	['b̥ʰʁ̞asə]	['b̥ʰʁ̞asə]
[ˈlyŋ <mark>gb</mark> y]	[ˈl <b>y</b> ŋb̓y]	[ˈłvŋby]	[ˈlyŋb̥y]	[ˈlyŋ <mark>b</mark> y]
[säunə]	[säunə]	[ <mark>s</mark> äunə]	[säunə]	[ <mark>s</mark> äun]
[ˌҟ <mark>n</mark> :6]	[ˌᠷ <mark>ᠯ</mark> ᢒ]	[ˌŔ <mark>٨</mark> \$)	[ˌŔ <mark>λ</mark> 9]	[ˌŔÀːᢒ]
[coģoˈlal <sup>j</sup> ]	[coģoˈlɛːð̪ˠ ə]	[cokoˈlɛ ð̪ˠ ə]	[cokoˈlɛð̪ˠ]	[coģoˈlɛːð̪ˠ ə]
[ʁฺuˈtiːnə]	[ʁฺuˈtiːnə]	[kɨnˌðsi. <mark>b</mark> ə]	[ʁฺuˈd͡siːnə]	[ʁ <mark>ʊ</mark> ˈd͡siːnə]
[ˈb̥äːsġ]	[ˈb̥äːsġ]	[ˈb <mark>ɑ̈sk</mark> ]	[ˈb̥äːsġ]	[ˈ <mark>b</mark> äːsĝ]
[ˈ <mark>to</mark> j]	[ˈ <mark>t</mark> ʌj]	[ˈtəj]	[ˈd͡sʌj]	[ˈtʌj]
[ˈçœu̯]	[ˈ <mark>sj</mark> ɔu̯]	[ˈ <mark>sj</mark> ə̯u]	[ˈcœu̯]	[ˈ <mark>ç</mark> өµ]
[om'tolili]	[ømˈtələli]	[ <mark>u</mark> mˈt̪ɔl <sup>j</sup> el <sup>j</sup> i]	[ˈœmˌd͡sə̯ləli]	[œmˈ <mark>t</mark> ələli]
['mul <sup>j</sup> ivis]	['muli_vis]	['mul <sup>j</sup> i'vis]	['muːliˌvi²s]	['muːl <sup>j</sup> ivis]
[ˈģɛːʊ̯nə]	[ˈ <mark>g</mark> ɛnə]	[ˈģɛɐ̯nə]	[ˈg̊ <mark>ɛ</mark> ːɐ̯nə]	[ˈ <mark>ɡe</mark> ɐ̯nə]
[ˈ <mark>m<sup>j</sup>œ</mark> ːle]	[ˈmɔːlə]	[ˈmoː <mark>l<sup>i</sup>ə]</mark>	[ˈm <mark>o</mark> ːlə]	[ˈməːlə]
[äkupuŋ' tuɐ]	[ekupun'tue]	[äkubhun'dsue]	[äghubhun'dsuv]	[vkopon'dsuv]

[ˈväģd]	[ˈväġ <mark>t</mark> ]	[ˈ <mark>v<sup>f</sup>ɛ</mark> ġd]	[ˈväg̊t]	[ˈväg̊t]
[ˈ <mark>kvota</mark> ]	[ˈġʰv <mark>ot</mark> ə]	[ˈģʰvọːd̥ə]	[ˈģʰvoːtə]	['kvotɐ]
[ˈhäu̯n]	[ˈhäu̯nə]	[ˈhäu̯ <mark>ŋ</mark> ə]	[ˈhäu̯nə]	[ˈhäu̯nə]
[ˈh <mark>œฺne</mark> ]	[ˈh <mark>œ</mark> ːnə]	[ˈhœːə]	[ˈhœːnə]	[ˈhœːnə]
[ˈġʰniu]	[ˈġʰniu]	[ˈġʰniu]	[ˈģʰniu̯]	[ˈġʰniu̯]
[ˈkɔ̣̣̣̣́́uje]	[ˈ <mark>k</mark> ɔə]	[ˈkoə]	[ˈģʰ <mark>ɔ</mark> ə]	[ˈġʰɔ̯ːə]
[ˈuːə]	[ˈuːə]	[ˈuːə]	['uːə]	['u: <sup>ə</sup> ]
[ģʰ <mark>ọ</mark> ˈʁ̞ɔub <code>t]</code>	[ <mark>kວ</mark> ˈʁ̞ub̥ <code>t]</code>	[ <mark>ko</mark> ˈʁub̊d]	[Ĝŗ <mark>o</mark> ,Řvβt]	[ <mark>g̊µ</mark> oˈʁʌb͡t]
[ˌpɨcň]	[ˌp̂ŧeň]	[ˌp̂ŧeň]	[ˌp̂ <b>ĸ</b> ձeň]	[ˌp̊ <mark>kɛ</mark> й]
[ˈ <mark>tœฺklɛ</mark> ʲə]	[ˈ <mark>t</mark> œɡ̊bleːð̪ˠə]	[ˈd͡ <mark>sœ</mark> ːg̊ʰleːð̪ˠə]	[ˈd͡s <b>y</b> ɐ̯ğʰleːð̪ˠə]	[ˈd͡͡sœɡ̊ʰˌl <mark>e</mark> ːð̪ˠə]
[ˈm <mark>unn</mark> ]	[ˈmʉnə]	[ˈm <mark>un</mark> ə]	[ˈmunə]	[ˈmunə]
[ˈ <mark>tið</mark> lid]	[ˈ <mark>ˈt</mark> ið̪ˠlʲi <code>t]</code>	[ˈd͡sid̪lit]	[ˈd͡sið̪ˠli]	[ˈ <mark>d͡s</mark> iðᢩv <mark>]</mark> ʲɪd]
[fiziolo'gi <sup>?</sup> ]	[fyci <mark>ələ</mark> ˈģi]	[fyzɪoloˈɡi]	[fɪzɪoloˈġi]	[fy <mark>si</mark> ǫlǫˈgi²]
[ˈve <mark>k</mark> ə]	[ˈve <mark>k</mark> ə]	[ˈvekə]	[ˈvɛkə]	[ˈveɡ̊ə]
[fʌˈskeli]	[fəˈskeli]	[fʌˈskeli]	[fɐˈsg̊eli]	[fəˈskeli]
[ˈtựyke]	[ˈ <mark>t</mark> yɡ̊ə]	[ˈt <mark>ʁu</mark> g̊ə]	[ˈd͡s͡k͡ʌg]	[ˈd͡sʁ <mark>̯y</mark> g̊ə]
[ <mark>b</mark> eˈʁɛlskɛb]	['b <mark>e<sup>y</sup>ɛð̪ˠ</mark> ˌsgɛb̥]	[ˈb <mark>eeskɐ</mark> b̥]	[þ <mark>ɛ</mark> ˌř <mark>ɛq</mark> ˈsลɛ͡b]	[p̊ <mark>e</mark> ˌkɛð̃ <sup>រ័</sup> ˈsg̃ɛ <mark>b</mark> ]
[bɪd͡ɕˌðo.͡ɕ]	[pedɐˈģou̯]	[ <mark>p</mark> ed̥a̯ˈɡ̊o̯u̯ˀ]	[ped̥ɐˈģou]	[pedˈ <mark>ɐ</mark> ˈgou̯²]
[nɛɐ̯vøsiˈ <mark>tet</mark> ]	[nɛvøsiˈ <mark>te</mark> ]	[ <mark>n</mark> əvøsiˈd͡s <mark>e</mark> t]	[nevøsi'dse?d]	[nəvøsi'dse²t]
['lɛze]	['leːsə]	['leˈsə]	['leːsə]	[ˈlesə]
[ˈf <mark>iz</mark> isĝ]	[ˈfysisĝ]	[ˈfyzisĝ]	[fy'zisĝ]	[ˈfysisĝ]
['oŋdom <sup>j</sup> e li]	[ə̯ŋˈ <mark>də</mark> məli]	[ˈuŋd̪omli]	[ʉŋd̥ <mark>ə</mark> meˈli]	[ʊŋˈ <mark>d</mark> oməli]
[ˈåʰyse]	[ˈkøsə]	[ˈ <mark>k</mark> øฺsə]	[ˈ <mark>k</mark> øsə]	[ˈġʰysə]
[ˈgʰø̞ːbe]	[ˈk <mark>ø</mark> ˈb̥ə]	[ˈ <mark>kø</mark> ːþə]	[ˈģʰợːþə]	[ˈģʰợːþə]
[ˈd̪ɪaŋġ]	[ˈd̥ <mark>ʁɛ</mark> ŋ]	[ˈd͡ҟa <mark>រ</mark> ]	[ˈd̥ʁ̞aŋġ]	[ˈ <mark>d</mark> ĸeŋ]
[nøl'ven <mark>d</mark> i]	['nøðvvendi]	[ <b>n</b> ợðvˈve <b>nd</b> <sup>j</sup> i]	[nøðv, ven, di]	[nøðv vendi]
[pedaˈgogisk]	[ped̥a̯ˈ <mark>go</mark> gisĝ]	[pedv'goisk]	[ped̥ɐˈɡ <mark>̊oːg</mark> isg]	[pedˈɐˈˈģoɡ̊isɡ̊]
[ˈsm <mark>o</mark> ·]	[ˈsmæ]	[ˈsmœ]	[ˈs <mark>m<sup>j</sup>œ</mark> ɐ̯]	[ˈsmæ̯ʲ]
[ˈlø̞k <sup>j</sup> e]	[ˈl <mark>u</mark> g̊ə]	[ˈl <mark>ụk</mark> ə]	[ˈl <mark>u</mark> g̊ə]	[ˈ <mark>łu</mark> g̊ə]
[ˈ <mark>d</mark> kɔmedaː]	[ˈd̥kəːmədɑ]	[¦ďřomэďa]	[ˈd̪ʁsmə <mark>d</mark> a]	[ˌq͡ҟɔ̀məq͡ <mark>ɕ</mark> ]
[b̥ə̯wˈmesd̯ʌ]	['b̥ɔmesd̯ʌ]	[ˈb̥əwˌmesd̪ʌ]	[ˈb̥əwmesd̥ʌ]	[ˈ <mark>bə̯</mark> ˌmesdʌ]
['nɛmee]	[ˈnɛɐ̯mʌ̯]	[ˈ <mark>n</mark> ɛɐ̯mə]	[ˈnɛɐ̯mʌ្]	[ˈnɛɐ̯mʌ̯ʌ]
[ˈstə̥ːmə]	[ˈsd̥ɔːmə]	[ˈstəmə]	[ˈstə̯ːmə]	[ˈs <code>tᢩɔːmə]</code>
[ˈʌ <mark>ᡅ</mark> ʌ]	[ˈʌʌnʌ]	[ˈʌ̯nə]	[ˈə̯ntə]	[ˈ <mark>ə</mark> nʌ]
D 6	D 11	D 12	D 13	
[pạnə]	[b̥ʰạnə]	[panə]	[panə]	
[ˈd̪yn̪ə]	[ˈdɣnə]	[ˈd̪yn̪ə]	[ˈ <mark>d</mark> yn]	
[ˈ <mark>k</mark> ɛɐ̯liˌhẹð̪ˠ]	['ġʰɛɐ̯li hẹð̪ɣ]	[ˌ <mark>ke</mark> liˈhẹð̪ˠ]	[ˈ <mark>k</mark> ɛliˌxəl <sup>J</sup> ]	
[u <mark>tak</mark> 'nɛmli]	['udsäg,nemli]	[utak'nɛmɛli]	['ut̪ɐkˌn̪emli]	

[ˈ <mark>bạ</mark> ð̪ˠ əˈvɛls̪ə]	[ˈb̥ <mark>ɛ</mark> ð̪ˠ əˈvɛɐ̯lsə]	[1
[ˈmäŋglə]	[ˈmäŋlə]	[
['yä:]	[ˈf <mark>a</mark> ːə]	[
['ọnˌsǧɛp]	['ọnˌsg̊ɛb̯]	[
[ˈ <mark>so</mark> ²ːʃɐ]	[ˈsoːfa]	[
[ˌŔ <mark>ġ</mark> ᢒ]	[ˌҟ <u>ɑ</u> ːᢒ]	[
[ <mark>bəˈ<mark>st</mark>emə]</mark>	[b̥eˈsd̥emə]	[
[ˈh <mark>ə</mark> bə]	[ˈhʌᢩbə]	[
[dɔm]	[dɔm]	[(
[sģäːþ]	[sģäːþ]	[
[ˌŜ <sub>p</sub> ɛၗ̃ɛď̥əҟᠯ]	[ˌg͡ <sub>p</sub> ɛၗ̆vɛɡ̊ˈək͡ <mark>ə</mark> ]	[
[ <mark>st</mark> om]	[sd̥ɔ̈́m]	[
[ɛɐ̯ˈkenə]	[ˈɛɐ̯ɡ̊ʰənə]	[
[ˈ <mark>s̪vt</mark> ̪ən]	[ˈ <mark>sy</mark> d̥n]	[
[ˌ <mark>ই</mark> t͡ҟɛð̪ʎə]	[ˌsģ祚 <mark>i</mark> ፬ᠯ ə]	fa
[ˈ <mark>ga</mark> fəl]	[ˈġäfəl]	[
[ˈkiŋə]	[ˈkinə]	[
[ˈh <mark>ɛ</mark> ð̪ˠə]	[ˈhạᢩ <sup>ð</sup> ə]	[
['v <mark>askət</mark> ]	[ˈvạs <mark>k<sup>j</sup>əð̪ˠ]</mark>	[
[ˈ <mark>bin</mark> ə]	['benə]	[
[ɑ̈b̥ɑˈk̈ɑt]	[ɑ̈b̥ɑˌk̈ɑ <mark>ð</mark> ͡ɛ]	[0
[bɐˈˈg <mark>aːz</mark> ə]	[bɐˈåɑz]	[]
[ea <mark>ŭ</mark> ad]	[ˈ <mark>p</mark> ạnəə]	[]
[bɪˈslaġtət]	[b̥ɪˈslɛɡ̊ <mark>d͡</mark> səd͡s]	[]
[ˌfɔˈb̥eləð̪ˠ ə]	[ˈfəːˌbeləð̪ˠ ə]	[
[əˈlɛːnə]	[aˈleːnə]	[;
[ˈmạð̪ˠ̯ ]	['mạ ð̪ˠ ]	[
[ˈ <mark>sin</mark> ə]	[ˈseːnə]	[
[ˈfal <mark>sk</mark> ]	[ˈfalsĝ]	[
[ˈb̥ <mark>u</mark> ˀ]	[ˈb̥ <mark>u</mark> ə]	[
[ˈ <mark>d</mark> əən]	[ˈd <mark>̥əุu</mark> n]	[
[ˈ <mark>g</mark> ʲe mə]	[ˈģemə]	[
[ˈg <mark>ans</mark> kə]	[ˈ <mark>ġ</mark> änsġə]	[
[ˈfl <mark>a</mark> mə]	[ˈflämə]	[
[ˈgul <sup>f</sup> ]	[ˈgul <sup>f</sup> ]	[
[pɪˌ <mark>ɑ</mark> äiɛɡˈk͡əðʌ͡]	[p̊ɪˌäɑisd̈kə <mark>d̈́s</mark> ]	[
[ˈ <mark>skiŋ</mark> ə]	[ˈ <mark>sk</mark> inə]	[
[ˈ <mark>xɛmɛlit</mark> ]	[ˈhiməlihəð̪ˠ]	[
[ˈfilm]	[ˈfilm]	Γ
[ˈvilə]	[ˈv <mark>i</mark> lə]	[
-	_	-

baə'vel<mark>s</mark>ə] mäŋ<mark>g</mark>lə] fɛə] <mark>ənˈsg̊ɛ</mark>b] '<mark>so</mark>?:fɐ] '<mark>.ıäə</mark>] bəˈ<mark>st</mark>emə] ˈh<mark>ɔ</mark>b̥ə] dum] sĝäːp] ˈୡ<sub>ʰ</sub>ɛʁฺɛsd̥əə] stom] ˈɛɐ̯kə<mark>n</mark>ə] '<mark>svt</mark>ən] ailed '<mark>ga</mark>fəl] kinə] ˈh<mark>ɛ</mark>ð̆ɣəd̯] vaskət] ˈ<mark>bin</mark>ə] äþä ¦ kät] be ga z] pəˈn<mark>ɛə</mark>] bɛˈslɛtət] fəˈ<mark>b</mark>eləð̪ˠ ə] əˈl<mark>ɛ</mark>ˈnə] mą] ˈse̯nə] fạlsg] <mark>b</mark>u<sup>ə</sup>] ˈ<mark>d</mark>ɔn] ge mə] ˈ<mark>g</mark>ạnsĝə] fl<mark>a</mark>mə] gul<sup>v</sup>] bı 'gejst<mark>şət</mark>] ˈs<mark>ki</mark>nə] 'heməlid] fi<mark>l</mark>m] vil<mark>d</mark>ə]

 $[, p\dot{a}\bar{Q}_{\lambda}^{\dagger} \circ ' \Lambda \epsilon \dot{k} \circ l\ddot{8} \circ ]$ ['manglə] ['f<mark>a</mark>] ['on sgap] ['<mark>s</mark>ofɐ] ['ra] [bəsˈtimə] ['h<mark>op</mark>] [<mark>d</mark>um] [skap] [ˈkɛrɛs̪dərə] [sdom] [əˈkɛn̪ə] ['s<sup>j</sup>yten] [ˈ<u>st</u>riðə] [ˈ<mark>ga</mark>fl] ['kiŋɐ] [ˈhɛð̪ˠəd] ['v<mark>askəl</mark>] [ˈbin̪ə] [apa'rad] [bɐˈgazə] [pg neg] [bɪˈslegtəl] [ˈf<mark>əb</mark>ɪlərə] [e'lenə] [ˈmɛl] [ˈ<mark>sɛn</mark>ə] [ˈfalsĝ] [ˈbuə] [ˈd<mark>̥ə</mark>n] [ˈ<mark>gɛ</mark>mə] ['ganskə] ['fläm] [ˈgul] [bi'gajstrəl] [ˈ<mark>skin</mark>ə] ['xɛmɛlit] [ˈfilm] ['vil]

[ˈ <mark>ka</mark> j]	[ˈĝʰ äį]	[ˈ <mark>ka</mark> j]	[ˈ <mark>k</mark> äj]
[ˈ <mark>xumʊs</mark> ]	[ˈh <mark>ə</mark> mus]	['humus]	[ˈxumu <mark>s</mark> ]
[viˈka]	[viˈkä]	[vɪˈ <mark>k</mark> a]	[viˈ <mark>ka</mark> ]
[ˈli <mark>ʊ</mark> li]	[ˈliu̯li]	[ˈliʊli]	[ˈliʊlɪ]
[ˈ <mark>skri</mark> ə]	[ˈsģʁ <mark>i</mark> və]	[ˈ <mark>skɹi</mark> və]	[ˈ <mark>skri</mark> və]
[ˈ <mark>d</mark> aj]	[ˈd <mark>e</mark> i]	[ˈ <mark>da</mark> j]	[ˈ <mark>d</mark> aj]
[ˈ <mark>œ</mark> ð̪ˠ]	['j <b>ọl</b> ]	[ˈjəð̪ˠ]	[ˈjɔ]
[ø <mark>k</mark> olo' <mark>g</mark> i]	[øgʰoloʻgi]	[øgʰoloˌˈ <mark>g</mark> i]	[əˈkologɪ]
[ <mark>d</mark> ıˈt̪a̯ljə]	[d̥ɪˈd͡sa̯ljə]	[d̥ɪˈta̯ljə]	[ <mark>d̪ɪˈt̪a</mark> ljə]
[ˈki <mark>s</mark> əˌb <sup>j</sup> ɛɐ̯]	[ˈġʰiɐsəˌb̥ɛɐ̯]	[ˈ <mark>ki</mark> ɪsəˌbɛɐ̯]	[ˈkirs̪əˌbeə]
[ˈlɛŋɡə]	[ˈleŋə]	[ˈlɛŋə]	[ˈ <mark>łɛng</mark> ə]
[ˈ <mark>bop</mark> əl]	['bopel]	[ˈb <mark>opɛ</mark> l]	['boˌpel]
[mɔˈ <mark>s</mark> kɪ]	[məˈsɡ̊ə²]	[məˈ <mark>skı</mark> ]	[moˈ <mark>ske</mark> ]
[ˈkuːlə]	[ˈĝʰulə]	[ˈ <mark>k</mark> ulə]	['kul <sup>j</sup> e]
[ˈ <mark>ł</mark> amˌkøð̪ˠ]	[ˈläməˌģʰø̞ <mark>l</mark> ]	[ˈla̪mˌkø]	[ˈläməˌģʰ əl]
[ˈ <mark>baŋk<sup>j</sup>ə]</mark>	[ˈb̥äŋġə]	[ˈb̥äŋ <mark>k<sup>j</sup>ɛ</mark> ]	[ˈbäŋk <sup>j</sup> e]
[ˈ <mark>smɛð̪ˠ ə</mark> ]	[ˌsmāgʌˈkv]	[ˈ <mark>s</mark> mąðᢩˠ <b>ıɛ</b> ]	[ˈ <mark>s</mark> ma <sup>ð</sup> ɾə]
[ˈ <mark>kuːn</mark> ə]	[ˈģʰə̯nə]	[ˈ <mark>k</mark> ənə]	[ˈ <mark>kun</mark> ə]
[ˈ <mark>n̪ɛs</mark> d̥ə]	[ˈnesdə]	[ˈnesd̥ə]	[ˈ <mark>n̪ɛs̪d</mark> ə]
[ˈ <mark>g</mark> äŋə]	[ˈģäŋə]	[ˈ <mark>g</mark> äŋ <mark>g</mark> ə]	[ˈ <mark>gɑŋg</mark> ə]
[ˈģ <mark>o</mark> ðᢩˠ]	[ˈģoᢩðᢩv]	['go]	[ˈ <mark>g</mark> o̞]
[ˈĝœ·]	[ˈĝœː]	[ˈġ <u></u> Ţ]	[ˈg <sup>j</sup> ə]
[ˈm <mark>ɛn̪ɛs</mark> ̪kə]	['menəsg̊ə]	['m <mark>ɛnɛ</mark> skɛ]	[ˈm <mark>ɛn̪ɛs</mark> kə]
[ˈ <mark>ba</mark> ŋ]	[ˈb̥äŋ]	[ˈ <mark>ba</mark> ŋ]	[ˈbaŋk]
[irok1'seə]	[iko, g <sub>p</sub> esv]	[iko¦ĝ <sub>p</sub> es9]	[Ito'ke'zə]
[ <mark>bo</mark> ˈliviənsk]	[þ <mark>o</mark> ˈliviənsĝ]	[bolivi'ansg]	[ <mark>boˈlivɪənsk</mark> ]
[' <mark>skole</mark> ]	[ˈsɡ̊ <mark>o</mark> lə]	[ˈs <mark>kolɛ</mark> ]	[ˈ <mark>sk</mark> o̞·lə]
[ˈk <mark>oːn</mark> ə]	[ˈģʰ <mark>o</mark> nə]	[ˈġʰ <mark>ə</mark> nə]	[ˈ <mark>koŋ</mark> ə]
[ˈ <mark>piskə</mark> ˌfləːð̪ˠ]	[ˈb̥ʰisɡ̊əˌf <mark>lø</mark> l]	[ˈb̥ʰisɡ̊əˌflø̯ð̪ˠə]	[ˈpi <mark>s</mark> kəˌf <del>l</del> əl]
[ <mark>kok</mark> əkˌĦnk]	[g̊ʰ <mark>ə</mark> ʀəg̊ˈd͡suɐ̯]	[g̊ʰ <mark>ɵɹ</mark> ˈktuɪ]	[ <mark>korək't</mark> u]
[ˈnoːsk]	[ˈnɔ̞ːsġ]	['nəsk]	[ˈ <u>nos</u> k]
[yiq'yq]	[b̥ʰäˈb̥ʰiɐ̯]	[papiə]	[paˈpiɐ̯]
[ˈ <mark>rid̥ə</mark> m]	[ˈ <mark>Ri</mark> ˌd̥ʌᢩm]	[ˈ <mark>.ɪidə</mark> m]	[ˈriˌdom]
[ˈfʁ̞aĝ]	[ˈfʁ̞ag̊]	[ˈfɪɛk]	[ˈf <mark>rɛk</mark> ]
[ˌŔ <mark>i</mark> 8]	['ʁis]	['ris]	[ˈ <mark>ːː</mark> ]
[ˈŷɛɐ̯]	[ˈɒ͡sɛ]	[ˈŷɛɐ̯]	[ˈɣ3 <mark>d</mark> ']
[ˈ <mark>k<sup>j</sup>iʁ̞k</mark> ə]	[ˈģʰiɐ̯ɡ̊ə]	[ˈġʰikə]	[ˈ <mark>ki.ɪk</mark> ə]
[ˈm <mark>o</mark> ]	[ˈmoɐ̯]	['mo]	['m <mark>o</mark> ]
[ˌbŕ <mark>ɛ</mark> ɛə]	['þʰʁ̯asə]	[ˈb̥ʰ <b>ɹɛ</b> sə]	[ˈ <mark>prɛs</mark> ə]

[ˈlyŋbɣ]	[ˈlybɣ]	[ˈlyŋb <mark>̪y</mark> ]	[ˈlyŋ <mark>gb</mark> y]
[ <mark>s</mark> äunə]	[säunə]	[ <mark>s</mark> äunə]	[säunə]
[ˈ <mark>ry</mark> ə]	[ˌŔ <mark>λ</mark> ᢒ]	[ˌ <mark>ˌˈ</mark> ]	[' <b>r</b> yə]
[ə͡skoˈlɛð̪ɣə]	[cogʻisi' <mark>o</mark> goi]	[tʃokoˈlɛə]	[ <mark>∫oko</mark> ˈlɛl]
[ <mark>ruˈt</mark> ʲiːnə]	[ʁฺuˈd͡ <mark>si</mark> nə]	[ <mark>.ruˈti</mark> nə]	[ruˈtiˈn̪ə]
[ˈb̥ä <mark>s</mark> ġ]	[ˈb̥äːsǧ]	[ˈbask]	['ba.isk]
[ˈ <mark>t</mark> oj]	[ˈd͡sʌj]	[ˈ <mark>to</mark> j]	[ˈ <mark>t</mark> əj]
[ˈ <mark>ç</mark> əu̯]	[ˈɛɔ઼̈́u]	[ˈ <mark>ʂ</mark> əu̯]	[ˈ <mark>sj</mark> ɔu̯]
[ <mark>ʊmˈt̪o</mark> ləli]	[omˈd͡sə̯ləli]	[ <mark>ˌə</mark> mˈtələli]	[ømˈ <mark>təl<sup>j</sup>el<sup>j</sup>i</mark> ]
[ˈmu·l <sup>j</sup> iˌvi <mark>s</mark> ]	['muli <sub>vi's</sub> ]	['mul <sup>j</sup> ivis]	[ˈmul <sup>j</sup> ɪvɪ <mark>s</mark> ]
[ˈģ <mark>ɛ</mark> ˈnə]	[ˈģɛɐ̯nə]	[ˈģɛənə]	[ˈ <mark>g</mark> ɛnə]
[ˈm <mark>o</mark> ːlə]	[ˈmə̯lə]	[ˈm <mark>ə</mark> ·lə]	[ˈmol <sup>j</sup> e]
[ekupuŋˈ <mark>t</mark> uɐ̯]	[ägub̥ʰuŋˈd͡suɐ̯]	[äkupuŋˈtu]	[ˈäkupuŋˌ <mark>t</mark> u]
['väģt]	[ˈväģ <mark>ds</mark> ]	[ˈväġt]	[ˈv <mark>ɛ</mark> ġ <code>t]</code>
[ˈkv <mark>oˈt</mark> ə]	[ˈg̊ʰvod͡͡sɐ]	[ˈģʰvod̥ə]	[ˈ <mark>kvot</mark> ə]
[ˈh <mark>ɛ</mark> u̯nə]	[ˈhäu̯nə]	[ˈh <mark>ɛ</mark> u̯nə]	[ˈ <mark>xa</mark> u̯nə]
[ˈh <mark>œ</mark> ˈnə]	[ˈhœːnə]	[ˈh <mark>œ</mark> nə]	[ˈh <mark>œ</mark> ːnə]
[ˈɡ̊ʰniu̯]	[ˈġniu̯]	[ˈġʰniv]	[ˈɡ̊ʰniu̯]
[ˈ <mark>kə</mark> ġə]	[ˈ <mark>k</mark> əə]	[ˈģʰ <mark>ɔ</mark> ə]	[ˈ <mark>koə</mark> ]
[ˈuːə]	[ˈuːə]	[ˈ <mark>u</mark> ə]	[ˈ <mark>u</mark> ə]
[ <mark>ko</mark> ˈʁ̞ub̥t]	[ୡ <mark>ʰo</mark> ˈʁuþ <mark>ð</mark> ͡s]	[ģʰ <mark>oˈ.ɪ</mark> ub̥t]	[koˈrupt]
[ˈb̥reu̯]	[,p̂řeň <sub>3</sub> ]	[ˈb̥ɪeu̯]	[ˈb <mark>rɛ</mark> u̯]
[ˈd͡ <mark>səġlɛ</mark> ːð̪ˠə]	[ˈd͡sœ <mark>ɡ̊ləl</mark> ]	[ˈ <mark>tœkleə</mark> ]	[ˈ <mark>t</mark> œkləl]
[ˈmɔːnə]	[ˈmunə]	[ˈmunə]	[ˈm <mark>unt</mark> ]
[ˈ <mark>t</mark> ʲiðᢩvʲlit̪]	[ˈd͡sið̪ˠlid͡ <mark>s</mark> ]	[ˈ <mark>t</mark> ilit]	[ˈ <mark>tilli</mark> ]
[f <mark>ızıolo</mark> 'ĝi]	[fy <mark>siolo</mark> 'ĝi]	[fy <mark>ziolo'g</mark> i]	[fɪsɪoloˈgi]
['v <mark>ɛkɛ</mark> ]	[ˈveɡ̊ə]	['ve <mark>k</mark> e]	[ˈve <mark>k</mark> ə]
[fəˈ <mark>s</mark> geli]	[fʌˈsɡ̊el <sup>?</sup> i]	[ˌfoˈsɡ̊eli]	[f <mark>əˈskel</mark> i]
[ˈ <mark>t</mark> ̪ryɡ̊ə]	[ˈd͡s͡k͡ʌ͡sə]	[ˈtɪʊɡ̊ə]	[ˈ <mark>t</mark> rykə]
[bɛˈ <mark>rɛð̆</mark> ɬ <mark>s</mark> ĝɛb̥]	[p̊əˌkɛ <mark>d</mark> ˈsg͡ɛp̊]	[ˈb̥eɛˌsɡ̊ɛb̥]	[bɪˈrelskəb]
[pedɐˈģou]	[pedˈ <mark>ˈvgou</mark> ]	[pedv'gog]	[pdv'gog]
[nɛvợsiˈd͡se²d]	[nəˈvɜsisd̥əd̯]	[nɛvœsi'tet]	[nə.vesiˈtɛl]
[ˈ <mark>łeːs</mark> ə]	[ˈl <mark>e</mark> sə]	[ˈl <mark>ɛs</mark> ə]	[' <mark>lɛːɛ</mark> ]
[fyˈ <mark>zisk</mark> ]	[ˈfysisĝ]	[ˈfysisĝ]	[ˈfysisĝ]
[ʊŋgˈd̪omeli]	[əŋˈdʌməli]	[ <mark>ʊ</mark> ŋˈdoməli]	[ʊŋgˈ <mark>d̪om</mark> əli]
[ˈ <mark>kø̞·s</mark> ə]	[ˈģʰ <mark>y</mark> sə]	[ˈģʰ <mark>y</mark> sə]	[ˈ <mark>kys</mark> ə]
[ˈ <mark>k</mark> øbə]	[ˈģʰ <mark>ợ</mark> 协ə]	[ˈģʰợ̃ːb̥ə]	[ˈ <mark>k</mark> ø̥bə]
[ˌģŕa <mark>Ĵ</mark> j]	[ˈd̥ʁaŋ]	[ˈd̪ɪaŋ]	[ˈd̪rɛŋk]
[nøl'venli]	[nøl'venli]	[nøˈwendi]	[ not'vend <sup>j</sup> i]
[pedvˈģ <mark>oģisk</mark> ]	[b̥ʰed̥ɐˈɡoisġ]	[pedv'gogisg]	[pedv'goigsk]

[ˈsm <sup>j</sup> œ]	[ˈsmæ̯ɐ]	[ˈsmo]	[ˈsmœ]
[ˈl <mark>ə</mark> ġə]	[ˈlɔ̥ġə]	[ˈlə̯ɡ̊ə]	[ˈl <mark>u̯kɛ</mark> ]
[ˈ <mark>dromədɐ</mark> ]	[ˌˈd͡kɔɛd͡s]	['dıomədɐ]	[ˈd̪roməd̪ɐ]
[b̥əˈmi <mark>st</mark> ə]	['b̥əwmesdʌ̯]	[bosˈmestə]	[ˈb̥ <mark>ɔ.ɪk</mark> ˌmaəstə.ɪ]
[ˈnɛm <mark>ɛə</mark> ]	[ˈnɛɐ̯mʌ̯]	['nemeə]	[ˈ <mark>ŋɛmɛɹə</mark> ]
[ˈstəmə]	[ˈs <mark>ds͡ə</mark> m]	[ˈ <mark>sd̥ə</mark> ːmə]	[ˈstəɪmə]
[ˈənə]	[ˈə̯nə]	[ˈon <sup>d</sup> ə]	[ˈənd̥ə]

# Appendix 21. Transcriptions of the diphthong and word stress targets in the R-group

R 1	R 4	R 5	R 6	R 7
['bilist]	['bilisd]	['bilisd]	['bilisd]	[biˈlisd]
[ɪvenˈ <mark>ts</mark> yʁ̞əð̪ˠ]	[ɪvənˈd͡syʁ઼əð̆ˠ]	[' <mark>e</mark> vənd͡ <mark>su</mark> kəð̆ <sup>y</sup> ]	[ <mark>1</mark> vən'dsykəð <sub>i</sub> ]	[ <mark>ə</mark> vən'd͡sykəð̪ˠ]
[ˈģʰvȩnəli]	[ˈģʰvẹnəli]	[ˈ <mark>k</mark> vȩnəli]	[ˈġʰvinəli]	[ˈġʰvinəli]
['u helt]	[v'x <sup>j</sup> el]	['uˌ <mark>x</mark> <sup>j</sup> el]	[ <mark>ʊˈx<sup>j</sup>el]</mark>	['uhel]
[kaˈjak]	[ģʰɐˈjäģ]	[ <mark>kɐ</mark> ˈjäk]	[ģʰɐˈjäĝ]	[ģʰɐˈjäĝ]
[misˈ <mark>ts</mark> äŋġə]	[misˈ <mark>ts</mark> äŋġə]	[ˈmis <mark>ts</mark> äŋ <mark>k</mark> ə]	[misˈ <mark>ts</mark> äŋģə]	[mis' <mark>ts</mark> äŋģə]
[ˈsɔ̣ːˌb̥ä]	[ˈsɔːˌb̥ä]	[ˈ <mark>s</mark> əːˌb̥ä]	[ˈsɔːˌb̥ä]	[ˈsɔːˌb̥ä]
[mis'tsen_som]	[mi <mark>s</mark> ˈd͡seŋˌsʌm]	[mi <mark>sˈts</mark> eŋ <u>s</u> ₄m]	[mi <mark>s</mark> ˈd͡seŋˌsʌm]	[misˈd͡seŋˌsʌm]
[b̥eˈ <mark>ab̥ajd</mark> ə]	[ˈb̥ɪäb̥äjd̥ə]	[ˈb <mark>eɐ</mark> ˌb̥äjð̪ə]	[ˌb̥ɪˈäb̥äjd̥ə]	[ˌb̥ɪˈäb̥äjd̥ə]
[ġ <mark>e</mark> ˈb̊y²ɐ̯]	[ģ <mark>ı</mark> ˈpyɐ̯]	[ģɪˈ <mark>þ</mark> yɐ̯]	[ġ <mark>ı</mark> ˈbyɐ̯]	[ġ <mark>ı</mark> ˈbyɐ̯]
['u <mark>kent</mark> ]	[ <mark>v</mark> 'ghend]	['u ghent]	[ʊˈġʰend]	['u,ghend]
[ˈsģjultə]	[ˈs <mark>g̊ʰ</mark> jultə]	[ˈ <mark>s̪g</mark> ʰjultə]	[ˈs <mark>k</mark> juld̥ə]	[ˈsgjuld̥ə]
[ģkɑˈtīģ]	[ậkɑˌtı̯ឆ]	[ģkɑˈtīģ]	[ģřaˌĮjĝ]	[ģkɑˌlıġ]
[ˈ <mark>s<sup>j</sup></mark> œwn løs]	[ˈ <mark>s̪<sup>j</sup>əə</mark> nˌløs]	[ˈ <mark>s</mark> əwnˌlø <mark>s</mark> ]	[ˌ <mark>s</mark> ʲəwnˈløs]	[ˈsœwnˌløs]
[vioˈli²n]	[vioˈlin]	[vioˈlin]	[vioˈlin]	[ <b>yio</b> ˈlin]
[kiɛˌtiːkə]	[ୡୢୖ୶ <mark>ୡ</mark> ୕୶diˌ <mark>ଌ</mark> ୢi୰]	[ <mark>k</mark> iļdi sēv]	[g̊ <sub>p</sub> ၨk <mark>ɨ</mark> ðɨː, sɨð	[ĜŗŔ <mark>i</mark> ġı, séřv]
[ˈjysĝ]	[ˈjysģ]	[ˈjysĝ]	[ˈjysģ]	[ˈjysģ]
[ˌuˈhel <sup>j</sup> i]	[uˈheldʲi]	['u heldi]	[u'heldi]	[u'heldi]
['bilist]	['bilisd]	[ˈbʲilisd]	[ˈbʲilisd]	['bilisd]
[øgoˈnomisg]	[øġoˈnomisġ]	[ø <mark>kəˈno</mark> misk]	[øgoˈnomisg]	[øgo'nomisg]
[ˈb̥äːnd̥ <mark>o</mark> m]	[ˈ <mark>p</mark> äːnd̥ʌm]	[ˈb̥äːnd̥ʌ̃m]	[ˈbäːndʌm]	[ˈbäːnˌdʌm]
[ˈjuːləfesd]	[ˈjuləˈfesd]	[ˈjuləˌfesd]	[ˌjuləˈfesd]	[ˈjuləˈfesd]
[tyvæˈʁiʔ]	[tsywe'?i]	[ <mark>ts</mark> ywe'ʁi <sup>?</sup> ]	[ģsàmv, ři,]	[ģ͡syweˈʁiˀ]
[uˈm <mark>uː</mark> lid̯]	[uˈmulit]	[uˈmuˈlit]	[uˈmuːlit]	[uˈmuːlit]
['bɛj efdtʌ]	['b̥ej'efd̥ʌ]	['b̥ <mark>e</mark> j'efd̥ʌ]	[ˌb̥ejˈefd̥ʌ]	[ˌb̥ejˈefd̥ʌ]
[ˈjewnˈ <mark>ald</mark> ʁ̞ənə]	[ˈjewn hạld͡͡͡]	[ˈjewnˌalʁ̞ənə]	[ˈjjewn <sup>ə</sup> ldˈʁɛnə]	[ˈjewnˈald̥ʁənə]
[ˌfemʌ̯ˈ <mark>ts</mark> ʁ̞eje]	[ˈfemʌ̯ˈd͡ <mark>s</mark> ²að̪ˠᢩvə]	[ˈfemʌ̯ˈd͡sʁ̞að̪ˠᢩvə]	[ˈfemʌ̯ˈd͡sʁ̞ɛð̪ˠvə ]	[ˈfeməˈd͡skaðʌvə]
[y ˈ <mark>ts</mark> ʁ̯eŋs ˌfʁ̯i ˈhəð̪ˠ]	[ˌyd͡͡͡²eŋsˈfʁ̞ihəð̪ˠ]	['Argenting, tripagal	ðِvj) [yd͡sˌʁiŋsˌfʁฺiˈhə	[ˈyd̥kɪŋsˈtɨtihəð̪ɨ]

[ˈ <mark>di</mark> ˈʁ଼a <mark>gt</mark> ə]	[d๋ <mark>i</mark> ,k͡sậtɔ]	[ <mark>ជ</mark> រˌᠷɛਙ͡d̥ə]	[ <mark>d</mark> ɪˈʁ <mark>ɛ</mark> g͡tə]	[d๋ıˌᠷ <mark>ɛ</mark> ਫ਼ੈtə]
[,å] ҟɛ:፬̃ <sup>៱</sup> ӭ]	[ˌåj <b>ŕ</b> 9̧Å <mark>ð</mark> ]	[ˌåʃ <mark>R9</mark> Qূź <mark>9</mark> ]	[ˌåj <b>ҟ</b> Ӭӯ <sub>҈</sub> ӭ]	[ˌåjk͡əð̪៱͡ə]
[ˈb̥äːnəˌvə̯wn]	[ˈb̥äːnəˌvə̯wn]	[ˈb̥äːnəˌvə̯wn]	[ˈ <mark>b</mark> äːnəˌvə̯wn]	[ˈb̥äːnəˌvə̯wn]
['ạnəðᢩێ'sb̥¤osb̥ʰed̥ɐ'goui sk]	['anaðy, sþ¢obhedæ, gog ik]	['ạnəðv, sþ <b>kobe</b> ås, ĝo ĝ <mark>ik</mark>	['å <mark>b</mark> ağ <sub>ň</sub> 'spřop <sub>r</sub> eďs, gogi <mark>y</mark> ]	[,ạnəðv ,sþ <u></u> kosþ <sup>h</sup> ed <mark>vgo</mark> 'gi k]
[ˈləwˌġiwnɪŋ]	[ˈləwˌġiwnɪŋ]	[ˈləwˌġiwnɪŋ]	[ˈlə̯wˌ <mark>g</mark> iwnɪŋ]	[ˈlə̯wˌ <mark>g</mark> iwnɪŋ]
[ˈiwnəˌsvaj²]	[ˈewnəˌsvɛ²j]	[ˈiwnəˌ <mark>s</mark> vɛj]	[ˈewnəˌsvɛj]	[ˈewnəˌswɛj]
[ˈb̥ʰẹ̃ː <mark>b</mark> ʌ̯]	['bʰe̯wʌ]	[ˈb̥ʰe̥b̥ʌ]	[ˈb̥ʰ <mark>ib</mark> ̥ʌ]	[ˈb̥ʰ <mark>ib</mark> ̥ʌ]
['äutso,metsisg]	[ <mark>əto</mark> ˈmɛtisǧ]	[ətsə'metsisg]	[äutoˈmɛti <mark>s</mark> ĝ]	[äuto'mɛti <mark>s</mark> ĝ]
[ˈd̃řimˈµn₅2]	[ˈd̪̃kiwˈhus]	[ˈd͡kiw hus]	[ˈd͡kiw hus]	[¦ďři,m,hn,z]
[ˈǿw <mark>ř</mark> 9]	[ˌۿゕŔᢒ]	[ˈǿwŔᢒ]	[ˌۿ <mark>ʌ</mark> ƙə]	[ˈǿwřə]
[ˈflə̥ə <sup>j</sup> u̯]	[ˈf <mark>lɔj</mark> əð̪ˠ]	[ˈfl <mark>ʉj</mark> əð̪ˠ]	[ˈf <mark>lɔj</mark> əð̪ˠ]	[ˈf <mark>lɔj</mark> əð̪ˠ]
[ˈdäjli]	[ˈ <mark>t</mark> ɑ̃jli]	[ˈdäjli]	[ˈ <mark>d</mark> äjli]	[ˈd̥äjli]
[ˈhujə]	[ˈhuə]	[ˈhui̯ə]	[ˈhui̯ə]	[ˈhujə]
[ˈġʰiɐ̯ġəli]	[ˈġʰiɐ̯ġəli]	[ˈģʰiɐ̯ġəli]	[ˈġʰiɐ̯ġəli]	[ˈġʰiɐ̯ġəli]
[ˈpɛɐ̯]	[ˈpɛɐ̯]	[ˈŷʰɛɐ̯]	[ˈpɛɐ̯]	[ˈpɛɐ̯]
[ˈfjɐ̯sɡ̊ən]	[ˈfe <mark>s</mark> g̊ən]	[ˈfeɐ̯sɡ̊ən]	[ˈfeɐ̯sɡ̊ən]	[ˈfeɐ̯sɡ̊ən]
[ˈdyɐ̯g̊ə]	[ˈdɣɡ̊ə]	[ˈdyɐ̯ɡ̊ə]	[ˈdyɐ̯ɡ̊ə]	[ˈdyɐ̯ɡ̊ə]
[ˈɡ̊ʰ <mark>œ</mark> səl]	[ˈģʰ <mark>ଡ଼</mark> ʲɛəl]	[ˈģʰ <mark>ø</mark> ʲsəl]	[ˈģʰ <mark>ø</mark> ʲsəl]	[ˈģʰ <mark>ø</mark> səl]
[ˌૡ <mark></mark> kəðᢩʎ]	[ˌ <mark>œ</mark> ჭə፬̃ <sub>Å</sub> ]	[ˌ <mark>ゐ</mark> ŔᢒQై <sub>Å</sub> ]	[ˌ <mark>œ</mark> ŔᢒQ <u></u> Å]	[ˌ <mark>œ</mark> Ŕᢒᡚ <sub>Å</sub> ]
[ubani'siɐ̯ʔ]	[ <mark>ʊpɐniˈs̪i</mark> ʌ]	[ˈʊb̥ɐniseʌ]	[opsui, <mark>žikə</mark> ]	[ <mark>ʊb̥ɐ</mark> niˈsɐ̯²ʌ]
[ˈb̥əɐ̯dʰ falə]	[ˈb <mark>̥o</mark> d̥ˈfalə]	[ˈ <mark>bo</mark> d̥ˌfa̯lə]	[ˈb <mark>̥o</mark> d̥ˈfa̯lə]	[ˈb <mark>̥o</mark> d̥ˌfa̯lə]
[ˈjạnuˌä]	[ˈjạnuˌä]	[ˈjạnuˌä]	[ˈjạnuˌä]	[ˈjạnuˌä]
[ˈjœːnə]	[ˈj͡œ̯nə]	[ˈj <mark>æᢩɐ</mark> nə]	[ˈj <mark>ឝ</mark> nə]	[ˈj <mark>ឝ</mark> nə]

R 10	R 12	R 13
['bilisd]	['bilisd]	[' <mark>be</mark> list]
[evən'd͡sy <sup>,</sup> kəð <sub>i</sub> ]	[evənˈd͡sy <mark></mark> ¥əð̆ <sup>ɣ</sup> ]	[ <mark>əvɪnˈd͡sy</mark> ¥əð̪ˠ]
[ˈġʰvinəli]	[ˈģʰvinəli]	[ˈkvẽnəli]
['u hel <sup>j</sup> ]	[ˌuˈhe <mark>l<sup>j</sup></mark> ]	['u'hel]
[ģʰɐˈjäģ]	[ģʰɐˈjäģ]	[ģʰ <mark>ä</mark> ˈjäĝ]
[misˈd͡säŋġə]	[mi <mark>sˈt͡s</mark> äŋġə]	[ˈmisd͡säŋġə]
[ˈsə̯ː ˌb̥ä]	[ˈ <mark>s</mark> əːˌb̥ä]	[ˈsəːˌb̥ä]
[mis'dsen,s₄m]	[mi <mark>s ˈts</mark> eŋ sʌm]	[mis'dsen sim]
['beˌäb̧äjd̥ə]	[ˌb̥ <mark>e</mark> ˈäb̥äjd̥ə]	[ <mark>be</mark>  'äbäj <mark>d</mark> ə]
[ģīˈbyɐ̯]	[ģɪˈb̥yɐ̯]	[ģeˈb̥yɐ̯]
['u, ghend]	[ʊˈk <sup>j</sup> end]	[ u'ghent]
[ˈsɡ̃juld̥ə]	[ˈsg̊ <sup>j</sup> ʊld̥ə]	[ˈsģjuld̥ə]
[ˈģ̃kɑ́t <mark>ɪs</mark> ɑ̃]	[ˈģ̃kätı͡ġ]	[ĝŧa ˈfiĝ]
[ˈsœ <sup>u</sup> nˌløs]	[ˌ <mark>s</mark> ʲəwnˈløs]	[ˈsœwnˌløs]
[vioˈli²n]	[vio'lin]	[viɔˈli²n]
[Ĝµҟiĝi zé,∛]	[ḁ̂̀ʰҟ <mark>i</mark> ̊diˌ <mark>ಶ</mark> եv]	[ḁ̂̀ʰҟiġiˌsե͡ҟᠯ]
[ˈjy²sģ]	[ˈjy <mark>s</mark> ġ]	[ˈjysģ]
[uˈheld̥i]	[ <mark>ʊˈx<sup>j</sup>eld̥i]</mark>	[uˈhe <mark>l<sup>j</sup></mark> di]
[bɨˈlisd]	[b̥iˈlisd̥]	[ˈbʲilisd]
[øgo'nomisg]	[ego'nomisg]	[øģoˈnomisġ]
	R 10         ['bilisd]         [evən'dsy'ţəðy]         ['gʰvinəli]         ['u,hel <sup>j</sup> ]         [gʰv'jäĝ]         [mis'dsäŋĝə]         [mis'dsâŋga]         ['sŷ:,bä]         [mis'dsâŋga]         ['bye,äbajda]         [gʰvidi]         ['sĝjulda]         ['skäfīsĝ]         ['sväfīsĝ]         ['sväfīsĝ]         ['jy'sĝ]         [u'heldi]         [bi'lisd]         [øĝo'nomisĝ]	R 10       R 12         ['bilisd]       ['bilisd]         [evən'dsy'ţəðy]       [evən'dsytəðy]         ['gʰvinəli]       ['gʰvinəli]         ['u,helʲ]       [,u'helʲ]         [jʰvijdš]       [gʰvijdš]         [bʰvijd]       ['gʰvijdi]         ['sʰvinəli]       ['u'helʲ]         ['u,helʲ]       [,u'helʲ]         ['u,helʲ]       [,u'helʲ]         ['sʰvidšaŋĝə]       [bʰvidšaŋĝə]         ['sɣ:,bä]       [mis'fsäŋĝə]         ['sɣ:,bä]       ['sɣ:,bä]         ['mis'dsaŋ,sʌm]       [mis'fseŋ,sʌm]         ['bɣ:dsajdə]       [,bɛ'äbäjdə]         ['bɣ:dsajdə]       [,bɛ'äbäjdə]         ['by:dsajdə]       [,bɛ'äbäjdə]         ['by!byt]       [g'byt]         ['u, gʰend]       [o'kʲend]         ['sĝjuldə]       ['sĝiddə]         ['sgildə]       ['sgifig̊]         ['sgifisg̊]       ['gwäfig̊]         ['sgidisg̊]       ['gwäfig̊]         ['swäfisg̊]       ['gʰwidi'sɛ̯?ʌ]         ['jy²sg̊]       ['jysĝ̊]         ['jy²sg̊]       ['jysĝ̊]         ['ju'sg³]       ['jusd³]         ['ju'sg³]       ['jysĝ³]         ['jysĝ³]       [

['bäːn,dʌm]	['bä:nd ,m]	['bä:ndʌm]	['bä:n,d,m]
['julə'fesd]	['ju:lə,fesd]	['julə,f <sup>i</sup> esd]	['julə,fesd]
[d͡sywe'ʁi²]	[dsyw, '¢i <sup>?</sup> ]	[tsywe'ʁi <sup>?</sup> ]	[dsyw,i'ţi]
[u'muːlit]	[u'mu:lid]	[u'mu:lit]	[u'mulid]
['bɛj,efdʌ]	['bɛj,efd,]	[,bej'efdʌ]	['bɛj,eft,]
[.jewn'aldʁənə]	['jewn,al¢,nə]	[,jewn'aldʁənə]	['jewn,aldţenə]
['Àğıus'fri'hiŷ <sub>i</sub> ]	['ydrens frihəð <sup>x</sup> ]	$['y \hat{\mathbf{ds}} d\mathbf{k} \eta \mathbf{s}, \mathbf{fk} \mathbf{h} \partial \mathbf{v}]$	[vdˈkiŋs fkiˈhəðv]
[dıˈʁɛɡ̊tə]	[diˈʁaġdə]	[dıˈʁ̞ɛğdə]	[diˈʁaĝdə]
[ˈalʁəð̪ˠə]	['aləˌʁ <mark>e</mark> ðˠə]	[ˈalʁ̞əð̪ˠə]	['alʁɛð̪ˠə]
[ˈbɑːnəˌvən]	['bɑːnəˌvə <sup>u</sup> n]	[ˈbɑːnəˌvəwn]	['bɑːnəˌvəwn]
[ˈa̯nəð̪v [ˈsɒˈkoɒˈʰedˈɐˈˈgogik]	['ạnəðv'sþ <b>ko</b> sþhed <mark>e</mark> 'yogisg]	[ˌa̯nəð̪v sb̥ʁos̪b̥ʰed̪ɐˈˈg̊og̊ik]	[,åuɔQ̄, sp̂ĸosbīd͡s, Bo'Big]
['ləw,giwniŋ]	['ləw,ğiwneŋ]	['łow,giwnŋ]	['ləw,giwniŋ]
['iwn,swɛj]	[' <mark>ja</mark> wnə,svɛj]	['ɛvnə,svɛj]	['ɛwnə,svɛ]
['bʰiþʌ]	['bʰęwʌ]	['piwʌ]	['pewʌ]
[öuto'mɛtisɐ̃]	[öudso'mɛdsisɐ̊]	[öuto'mɛtisɐ̃]	[ədsəmɐ'tisɐ]
['d̃kiw'hns]	[,ďřim'µn <sub>s</sub> ]	['dhiwhus]	['d <sup>s</sup> ʁiwˌhus]
[ˈøwʁə]	['øwġə]	[ˈøvʁə]	[, ŵmrờ]
[ˈflɔjəð̪ˠ]	['flʌ <sup>j</sup> əðˠ]	[ˈflɔjəð̪ˠ]	[, Įlý, ∋Q̃,]
[ dajii]	[ dajii]	[ dajii]	[ dajii]
['xujə]	['hujə]	['huə]	['hujə]
['kikəli]	['ğʰivğəli]	['kiekeli]	['gʰivgəli]
['bʰeɐ̯]	['bʰɛɐ̯]	[' <mark>pe</mark> ɐ̯]	['bʰɛɐ]
['feɐ̯s̪g̊ən]	['fɛɐ̯sgən]	['feɐ̯sg̊ən]	['fɛɐ̯sɡ̊ən]
[ˈdɣːɡ̊ə]	[ˈdɣv̯ġə]	[ˈ <mark>d</mark> ʊɐ̯ઙ૾ə]	[ˈdyɐ̯g̊ə]
[ˈk <mark>øِ<sup>ɐ</sup></mark> ̪səl]	[ˈgʰøɐ̯səl]	[ˈg̊ʰø <mark>ِ<sup>ɐ</sup></mark> səl]	[ˈg̊ʰøɐ̯səl]
[ˈw͡kəðv] [ˈbod falə]	['æ¥əð <sub>k</sub> ] ['po'¤q falo]	[ˈœ¥əðʌ] [opəui, <mark>ði</mark> kə]	[ˈœ̃kəðʌ] [npsuiˌsékə] [ˌpod tələ]
['jąnu,ä]	['jąnu,ä]	['jąnuˌä]	['jąnu,ä]
['jenə]	['j <mark>ę</mark> nə]	['j <b>գ</b> nə]	['jæ̯ɛnə]

# Appendix 22. Transcriptions of the diphthong and word stress targets in the D-group

D 1	D 2	D 3	D 4
[ˈ <mark>b</mark> ilisd]	['bili <mark>s</mark> t]	[bi'lisd]	['bʲili <mark>s</mark> t]
[eventy'et]	[əvənˈtyəd̯]	[ˈ <mark>ɛ</mark> ːvənˌd͡sy <mark>əd</mark> ]	[əvənˈd͡sy <mark>ʁʌ</mark> t]
[ˈġʰvinəli]	[ˈ <mark>k</mark> vȩnəli]	[ˈĝʰvęnli]	[ˈġʰvin <sup>d</sup> əli]
['uˌh <sup>j</sup> el <mark>d</mark> ]	['uel <sup>j</sup> ]	[u'helt]	[ˈ <mark>ʏhel<sup>j</sup>]</mark>
[ģʰąˈjäģ]	[ <mark>kɐ</mark> ˈjäĝ]	[kaˈj <mark>ak</mark> ]	[ģʰəˈjək]
[mis' <mark>tedŋke</mark> ]	[mi <mark>sˈd̥e</mark> ŋɡ̊ə]	[ˈmi <mark>s</mark> d͡säŋġə]	[ˈmi <mark>s</mark> d͡səŋġə]
[ˈsɔ̣ːˌb̥ä]	[ˈ <mark>s</mark> ə̯ˌb̥ä]	[ˈ <mark>s</mark> ɔːˌ <mark>bɐ</mark> ]	[ˈ <mark>s</mark> əːˌb̥ä]
[misˈt̪eŋˌsom]	[misˈd̥eŋˌs̪ʌm]	['mi <mark>s</mark> dseŋsəm]	[misˈd͡seŋ. <mark>s</mark> ʌm]
['be,beag] not valid	[ <mark>bəˈäbäjd</mark> ə]	[ˈb̥ <mark>ɛ</mark> äb̥äjd̥ə]	[ˈb̥ <mark>ı</mark> ˌäb̥äjd̥ə]
---	--	--	--
[geˈby·]	[gɪˈby]	[ģeˈbyɐ̯]	[ <mark>gıˈb</mark> yɐ̯]
[u'ghend]	['u ken]	[uˈgʰe <mark>nt</mark> ]	[ˈukənt]
[ˈs <mark>k</mark> jud̥ə]	[ˈsģjuld̥ə]	[ˈsģjuld]	[ˈsģjuld̥ə]
[ਫ਼ੈਸ਼ਂɑˈfiɡ̊]	[ਫ਼ੈk <mark></mark> ə, liğ]	[ĝŧɑˌᡶjፄ]	[ਫ਼ੈਸ਼ੰɑˌtıɡ]
[ˈsɒwnˌløs]	[ˈ <mark>s</mark> ʲœwnˌlø <mark>s</mark> ]	[ˈ <mark>sowp</mark> ˌl <mark>ys</mark> ]	[ˈ <mark>sʲyyn</mark> ˌlys]
[vioˈli²n]	[vio'li <sup>2</sup> n]	[vioˈlʲi̪n]	[vioˈlin]
[ĝµk <mark>eti</mark> ,sɛk]	[ģʰʁɨd̯iˈ <mark>siə</mark> ]	[ĝŀŧ <mark>it</mark> i' <mark>ŝla</mark> ]	[ĜŗŔŧġi,zĠv]
[ˈjysģ]	[ˈjysģ]	[ˈjysģ]	[ˈjysĝ]
[u'heldi]	[uˈ <mark>ɛld</mark> ̪i]	[ˌuˈ <mark>h<sup>j</sup>eld</mark> i]	[y'heldi]
[ˈbilisd]	[ <mark>b</mark> iˈlisd]	['bilist]	[biˈlisd]
[øåo'nomiså]	[øåo'nomiså]	[ø <mark>ko</mark> ˈnomisĝ]	[ <mark>e</mark> ģə'nəmisġ]
[ˈ <mark>bäŋˌd</mark> om]	[ˈbiäːnd̪əm]	[ˈb̥äːnˌd̥ ʌ̣ <b>m</b> ]	[ˈ <mark>b</mark> äːnd̥əm]
[ˈjuləˈfesd]	[ˈjuləˌfesd]	['jule'fesd]	[ˈj <mark>uː</mark> ləˌfesd]
[tywəˈ.ɪi]	[tywəˈʁi]	[ģsymv, ři,]	[ <mark>ģs</mark> ja <sub>m</sub> ə, ƙi]
[uˈmuˈli]	[ˈum <mark>ʊ</mark> li]	[uˈm <mark>u</mark> lit]	[ <mark>y</mark> 'myːlid]
[ bɛ'efdʌ] stress!	[ˈb̥ɛjˌefd̥ʌ]	[ˈbɛjˈefd̪ʌ]	[ˌbɛ <sup>j</sup> 'efdʌ]
[ˈjɐwnˈald̪ʁənə]	[ˈjewnˈal <mark>d</mark> ̥kənə]	[ˈj <mark>a</mark> wnˌaldʁənə]	[ˈj <mark>e</mark> ˀnˌ <mark>ald</mark> ̥ʁənə]
[ˈfem <mark>ətʁ̞e  ˈd</mark> iə]	[ˈfemʌᢩˈd͡sʁ̞að̪ <sup>s</sup> ɨvə]	[ˌfemʌ̯ˈtʁ̞að̪ˠvə]	[ˌfemʌ̯ˈd͡sʁ̞að̪ˠᢩvə]
[ˌ <mark>ut̪əhɪns</mark>  ˈfʁ઼iˌhi <sup>l</sup> ]	[ˌʌd̥kɪŋsˈt <code>kiəd]</code>	[ˈyd̥ <mark>ʁ</mark> yŋsˌfʁ̞ihəð̃ˠ]	[ˈ <mark>ůd</mark> iŋsˌfɣihəð̪v]
[d̪ɪˈreg d̪ə]	[d̥ɪˈʁ <mark>ɛ</mark> gɡ̊ə]	[d <mark>aj</mark> ˈĸagtə]	[diˈʁฺagdə]
[ˌåjř <mark>ɛ</mark> ð <sub>Å</sub> ə]	[ <mark>9</mark> ], ห <mark>้</mark> ย <sub>ู</sub> จี, 9]	[ål,Ř <mark>e9</mark> ]	[ˌˈájɪk͡əɡຼ <sub>ʌ</sub> ͡ə]
[ˈb̥ <mark>ឝ</mark> nəˌvə̯wn]	[ˈ <mark>b</mark> äːnəˌvə̯n]	['b៉äːnəˌv <mark>o</mark> ʷn]	[ˈbɑːnəˌvə̯wn]
['anə,spos pedge'gogik]	[ˈảuə፬ʌ <mark>ˈ</mark> z͡pᠷ <mark>oz</mark> ͡p <sub>p</sub> ed͡sɑo, aik]	['anssbuospedugo'gik]	['åuəỹ <sup><sup>k</sup> zpkop<sub>p</sub>edso, aik]</sup>
[ˈlə̥wgiwnɪŋ]	[ˈləwˌġiwnɪŋ]	[ˈləwˌgi <sup>v</sup> nɪŋ]	[ˈləwˌġiwnɪŋ]
[ˈiwnəˌsvɐj]	[ˈɛwnəˌsvɐ]	[ˈɛ <sup>v</sup> nˌs̪vɛj]	[ˈewnəˌsvɛ <sup>j</sup> ]
[ˈb̥ʰ <mark>ib</mark> ə]	[ˈb̥ʰęwə]	['bʰęwʌ]	[ˈb̥ʰ <mark>ẹ̃ːb̥</mark> ʌ]
[äydo'matisk]	[ <mark>ətəmɐˈti</mark> sġ]	[äuto'madsi <mark>sk</mark> ]	[ <mark>ɣutəˈmʌːti</mark> sġ]
[ˈ <mark>d</mark> ᢩʁiwhus]	[ˈd̥ʁiwhus]	[ˈd̥ʁiwhus]	[ˈd <mark>ˠju</mark> hus]
[ˈœə]	[ˌ <mark>ɕ</mark> mřť]	[ˌ <mark>Ġ</mark> MŔ <mark>9</mark> ]	[ˌۿːmŕə]
[ˈf <mark>lœed</mark> ]	[ˈf <mark>lə</mark> ʲə]	[ˈf <mark>lɔ̞əd</mark> ̯]	[ˈfl <mark>Ÿ</mark> jət]
[ˈ <mark>d</mark> ajli]	[ˈ <mark>d</mark> äjli]	[ˈd̥äjli]	[ˈd̥äjli]
[ˈhujə]	[ˈh <mark>uə</mark> ]	[ˈh <mark>uə</mark> ]	[ˈhuə]
[ˈ <mark>k</mark> iģəli]	[ˈ <mark>kiək</mark> əli]	[ˈģʰ <mark>iģ</mark> ʰəli]	[ˈ <mark>kɨək</mark> əli]
[ˈyɛɣ]	[ˈyəq']	[ˈ <mark>y</mark> 3q']	['bʰɛ¤]
[ˈfɛs <mark>k</mark> ən]	[ˈfɛɐ̯sɡ̊ən]	[ˈfɛɐ <mark>̯s</mark> kən]	[ˈfɛɐ̯sɡ̊ən]
[ˈ <mark>d</mark> yg̊ə]	[ˈd <mark>ɣ</mark> ːɡ̊ə]	[ˈd <mark>y</mark> g̊ə]	[ˈ <mark>dy<sup>ɐ</sup>kə</mark> ]

[ˈĝʰ <mark>ø</mark> səl]	[ˈkœːsəl]	[ˈɡ̊ʰ <mark>øᢩs</mark> əl]		[ˈ <mark>k</mark> øːsəl]	
[ˈૡ <mark>ə</mark> ]	[ˈ <mark>œəd</mark> ]	[ˌૡ <mark>୕</mark> kə͡Q <sub>Å</sub> ]		[ˌ <mark>œ.</mark> Ŕ9Qʌ]	
[ubeni'ze <sup>1</sup> ]	[ <mark>ʊbɐ</mark> niˈ <mark>şiə</mark> ]	[ubeni'ze	[9	[ <mark>ubɐniˈşe</mark> ʌ]	
['b̥od̥fạl]	[ˈb̥od̥fələ]	[ˈb <mark>o</mark> dʃalə	]	[ˈb <mark>o</mark> dˌfalə]	
[j <mark>ɐ</mark> nu'ä]	[jạnuˈä]	[ˈjąnuä]		[j <mark>e</mark> nu'ä]	
[ˈj̄œฺ̃n]	[ˈjœːnə]	[ˈhj <mark>æᢩn</mark> ə]		[' <sup>h</sup> jøːnə	)]
D 5	D 6	D 10	D 11		D 12
[' <mark>b</mark> ili <mark>s</mark> t]	['bil <sup>j</sup> i <u>st</u> ]	['bil <sup>j</sup> ig <mark>st</mark> ]	['bilisd]		[ˈbil <sup>j</sup> i <mark>st</mark> ]
[ɪvənˈd͡sykəð̪v]	[ɛv <sup>i</sup> enˈtyʁ̞ət]	[ <mark>ɛvenˈtyʁ̞ət</mark> ]	[evənˈd͡sy¥əð̪ɣ]		[ɛv <sup>j</sup> enˈtyʁ̞ət]
[ˈģʰvinəli]	['kv <mark>in</mark> li]	[ˈ <mark>kwin</mark> əli]	[ˈ <mark>k</mark> vinəli]		[ˈkv <mark>iŋ</mark> li]
['uˌhel <sup>j</sup> ]	[ˌuˈhɛl <sup>j</sup> ]	[ˈuˌhəl]	['u helt]		[ˌuˈhɛl <sup>j</sup> ]
[ģʰạˈjäĝ]	[kɐˈjak]	[kɐˈjak]	[ <mark>ka</mark> ˈjäĝ]		[kɐˈjak]
[mi <mark>s</mark> ˈd͡seŋġə]	[mi <mark>sˈt̪ɛŋk</mark> ə]	[mi <mark>sˈt̪aŋk</mark> ə]	[ˈmisd͡säŋġə]		[mi <mark>sˈt̪ɛŋk</mark> ə]
[ˈ <mark>s</mark> əˌba]	[ˈ <mark>s</mark> əˌbä]	[ˈ <mark>s̪ə</mark> ˌbä]	[ˈsəːˈþä]		[ˈ <mark>s</mark> əˌbä]
[misˈd͡seŋ s̪ʌm]	[mi <mark>sˈt̪ɛŋkˈs̪</mark> ʌm]	[ˈmi <mark>st</mark> əŋksʌm]	['mis dsansim]	]	[mi <mark>sˈt̪ɛŋkˈs̪</mark> ʌm]
[ <mark>bɪˈäbäjdə]</mark>	[ˈ <mark>beɐˈbäjd</mark> ̪ə]	[ˈbeɛkɨˈbäjd͡ə]	[ˈb̥ <mark>e</mark> äb̥äjd̥ə]		[ˈ <mark>beɐˈbäjd</mark> ̪ə]
[gıˈbyɐ̯]	[ge'by]	[gɪˈby]	[ģ <mark>e</mark> ˈb̥yɐ̯]		[ge'by]
[ˌuˈġʰent]	[ˌuˈ <mark>k</mark> ent]	['uˌ <mark>k</mark> en <mark>t</mark> ]	['u,ghent]		[ˌuˈ <mark>k</mark> ent]
[ˈsģjuld̥ə]	[ˈ <mark>sk</mark> juld̥ə]	[ˈ <mark>s</mark> ɡ̊juld̥ə]	[ˈsģjultə]		[ˈ <mark>s</mark> kjuld̥ə]
[ậҟɑˌlı͡ġ]	[ˈģʁٜäf <mark>ik</mark> ]	[ģựɑˈfi <mark>k</mark> ]	[ˈåʁ଼ɑfi̊g]		[ˈģʁٜäf <mark>ik</mark> ]
[ˈ <mark>s</mark> ʲœwnˌlø <mark>s</mark> ]	[ˈ <mark>s</mark> ʲœ <sup>u</sup> nˌlø <code>s]</code>	[ˈ <mark>s</mark> ʲœu̯nˌlø <mark>s</mark> ]	[ˈsœwnˌløs]		[ˈ <mark>s</mark> ʲœᢩ <sup>u</sup> nˌlø <code>s]</code>
[vio'li²n]	[vioˈlin]	[vioˈlin]	[vio'lin]		[vioˈlin]
[ĝ <sub>p</sub> riĝi   sé <sub>3</sub> <sup>*</sup> ]	[Ĝ <sub>p</sub> ŘĠďi <mark>Ž</mark> EÝ]	[kƙi <mark>t</mark> iˌ <mark>siƙə</mark> ]	[g̃µŘiďi, <mark>ŝc</mark> v]		[ĝ <sub>p</sub> Řěďi <mark>8</mark> £Ý]
[ˈjysģ]	[ˈjy <mark>s</mark> ġ]	[ˈj <mark>ʉٖs</mark> ġ]	[ˈjysģ]		[ˈjy <mark>s</mark> ģ]
['uheldi]	[u'hel <mark>d</mark> i]	[' <mark>u</mark> ,hel <mark>d</mark> i]	['uheldi]		[u'hel <mark>d</mark> i]
[biˈlisd]	[biˈli <mark>s</mark> d]	['bili <mark>s</mark> d]	['bilisd]		[ <mark>b</mark> iˈli <mark>s</mark> d]
[øġoˈno²misġ]	[økoˈ <mark>n</mark> omisk]	[əkəˈnomisk]	[ <mark>eĝo</mark> 'nomisĝ]		[økoˈ <mark>n</mark> omisk]
[ˈbäːn <mark>d</mark> ʌm]	['bäːn <mark>d</mark> ʌm]	[ˈbäːn <mark>d</mark> ʌm]	[ˈb̥äːnˌ <mark>d</mark> ʌᢩm]		[ˈbäːnd̪ʌm]
[ˈj <mark>uː</mark> ləˌfesd]	[ˈjuləˌfe <mark>st</mark> ]	[ˈjuləˌfe <mark>st</mark> ]	[ˈjuləˈfesd]		[ˈjuləˌfe <mark>st</mark> ]
[d͡sywʌˌki›]	[ <mark>t</mark> iyveˈʁi]	[tjyve'ķi]	[ģsāmv, ki]		[tʲyveˈʁi]
[uˈmuːlid]	[ʊˈmuːlʲit̪]	[ˈʉmu̞lʲi͡t̪]	['uˌmulit]		[ʊˈmuːlʲit̪]
[ˈbɛj efdʌ]	[ˌb̥ɛˈ <mark>ɛf̪t</mark> ̪ʌ]	[ˈb͡ɛjˈefᠯᢩʌ]	[ˈþɛjˈeftʌ]		[ˌb̥ɛˈɛf̪t̪ʌ]
[ˈjewnˌal <mark>d</mark> ʁənə]	[ˌjawnˈaldənə]	[ˈjæwnˌal <mark>d̥ʁ</mark> ənə]	[ˈj <mark>æ</mark> wnˌạl <mark>dʁ</mark> ənə	)]	[ˌjawnˈaldənə]
[ˈfemʌ̯ˈd͡sʁ̞að̪ˠᢩvə]	[ˌfemʌˈ <mark>tɹal</mark> və]	[ˈfemɔˈ <mark>t̪ʁ</mark> að̯və]	[ˈfemʌ̯ˈd͡sʁ <mark>ɛlj</mark> ə]		[ˌfemʌˈ <mark>tɹal</mark> və]
['yd̥̃u̥eŋsˌf͡uɨhəð̪ˠ]	[vઁ <mark>t</mark> ˈʁ <mark>v</mark> ̆sfʁ઼iˌhil]	[ˈ <mark>u़t̪</mark> ʁฺŋsfwiˌhəð]	[ <mark>yyy</mark> 'd <b>ķi</b> ŋs fķi	h <mark>ɛ</mark> ð̪ˠ]	[v <mark>̃́t</mark> ˈʁṽsfʁฺiˈhil]
[dɨˌkagdə]	[dirˈɛkt̪ə]	[ <mark>d̪iˈɣɛ</mark> g̊də]	[ˈdɨˌkaşdə]		[dirˈɛkt̪ə]
[ˌå]əҟ <mark>э</mark> ğ <sup>,</sup> ə]	[ð], Ř <mark>SQ</mark> Ə]	[ð], ƙ <mark>ç</mark> ə]	[ <mark>ˌa</mark> ]ˌ <b>ř</b> e͡ð <sub>Å</sub> ə]		[ð], ƙ <mark>ç</mark> ð
[ˈ <mark>b</mark> äːnəˌvə̯wn]	[ˈb̥äːnəˌvou̯n]	[ˈb̥äːnəˌvou̯n]	['b <mark>än</mark> vəwn]		[ˈb̥äːnəˌvou̯n]

o'gik]	o, <mark>8</mark> ik] [' <mark>å</mark> bəğ <sub>i</sub> "8p̀ko8beqs8	ogiy] [,åbdəg <sub>i</sub> "8pkobeds'8	<code>[ˈandəd̥sˈb̥ko<sup>u</sup>s̪b̥ʰed̥aˈ</code> [ˈandəd̥sˈb̥ko <sup>u</sup> s̪b̥ʰed̥aˈ	['åuðaga''' spåtosbedas o, sik]
[ˈləwˌġiwnɪŋ]	[ˈl <mark>owˌg</mark> iwnɪŋ]	[ˈl <mark>ownˌg</mark> iwnɪŋ]	[ˈləwˌ <mark>g</mark> iw <sup>n</sup> iŋ]	['low,giwnŋ]
['ewnəˌsv <mark>ɛ</mark> ]	[ˈ <mark>ɛv</mark> nəˌs̪vɛ <sup>j</sup> ə]	[ˈ <mark>ɛv</mark> nəˌ <mark>s</mark> vɛ]	[ˈewnəˌsvɛj]	[ˈ <mark>ɛv</mark> nəˌ <mark>s</mark> vɛ <sup>j</sup> ə]
[ˈb̥ʰẹ <mark>b</mark> ̥ʌᢩ]	[ˈpɛb̥ə]	[ˈpɛb̥ə]	[ˈb̥ʰẹ̣wʌ]	[ˈpɛɣə]
[äudso'medsisg]	[äu <mark>to</mark> 'm <mark>atisk</mark> ]	[äu <mark>to</mark> 'm <mark>atisk</mark> ]	[äudso'medsisg]	[äu <mark>t</mark> o'm <mark>at</mark> isk]
['d̥ʁiwhus]	[ˈd̥k͡ɛwhn͡ <mark>8</mark> ]	[ˈdɣiwhu <mark>s</mark> ]	[ˈ <mark>ds</mark> ʁiwˌhus]	[¦ẳẳɛ͡whu <mark>ɐ</mark> ]
[ˌۿmŔv]	[ˈøwrə]	[ˈœvə]	['ǿmŕť]	[ˈøwrə]
[ˈflʌ̯ʲəð̪ˠ]	[ˈf <mark>lə<sup>j</sup>t</mark> ə]	[ˈfl <mark>ə<sup>j</sup>t</mark> ]	[ˈflʌjət]	[ˈf <mark>lə<sup>j</sup>tə</mark> ]
[ˈ <mark>d</mark> äjli]	[ˈdäjli]	[ˈdäjli]	[ˈdäjli]	[ˈdäjli]
[ˈhujə]	[ˈhujə]	[ˈh <mark>u</mark> jə]	['huj]	[ˈhujə]
[ˈġʰiɐ̯ġəli]	[ˈ <mark>kiək</mark> əli]	[ˈ <mark>kiək</mark> əli]	[ˈ <mark>kiə</mark> kəli]	[ˈ <mark>kiək</mark> əli]
[ˈb̥ʰ <mark>e</mark> ɐ̯]	[ˈɡ3q']	['peɐ̯]	[ˈŷʰɛ¤]	[ˈyʒq']
[ˈfɛɐ̯sɡ̊ən]	[ˈfɛɐ <mark>̯sk</mark> ən]	[ˈfɛɐ <mark>̯sk</mark> ən]	[ˈfɛɐ̯sɡ̊ən]	[ˈfɛɐ <mark>̯s</mark> kən]
[ˈ <mark>dy<sup>ɐ</sup>k</mark> ə]	[ˈ <mark>dy<sup>ɐ</sup>k</mark> ə]	[ˈ <mark>dyk</mark> ə]	[ˈdyɐ̯g̊ə]	[ˈ <mark>dy<sup>ɐ</sup>k</mark> ə]
[ˈk <mark>ø</mark> ːsəl]	[ˈk <mark>øːs</mark> əl]	[ˈk <mark>øɐ̯s</mark> əl]	[ˈģʰøɐ̯səl]	[ˈk <mark>øːs</mark> əl]
[ˈ <mark>œ</mark> ĕðæː]	[ˈœəl]	[ˈ <mark>œɣə</mark> ðᢩv]	[ˌૡ <mark></mark> ႜႜႜ႞ᢒg <u></u> ၨ	[ˈœəl]
[ <mark>ʊbɐ</mark> niˈ <mark>s</mark> ʲi²ʌ]	[ <mark>obe</mark> ni'ze <sup>*</sup> ]	[ <mark>ʊbɐ</mark> niˈzeʌ̯]	[uɐ̯b̥aniˈseʌ̯]	[ <mark>obɐ</mark> niˈzeʌ]
[ˈb <mark>̥o</mark> d̥ˈfa̯lə]	[ˌb <mark>ot</mark> ˈfalə]	[ˌb <mark>ot</mark> ˈfalə]	[ˈb <mark>̥ot</mark> ˌfa̯lə]	[ˌb <mark>ot</mark> ˈfalə]
[ˈjạnuˌä]	[j <mark>a</mark> nu'ä]	[j <mark>a</mark> nu'ä]	[ˈj <mark>a</mark> nuˌä]	[j <mark>a</mark> nu'ä]
[ˈjœːnə]	[ˈhj <mark>æ</mark> ːnə]	[ˈj <mark>ឝ</mark> ːnə]	[ˈjឝฺnə]	[ˈhjœːnə]

## Appendix 23. T-test on the statistical significance of the difference in mean scores of the D- and R-groups.

#### 2/14/12 Student's t-test: Results

The results of an unpaired t-test performed at 11:34 on 14-FEB-2012 t= 2.04 sdev= 0.463 degrees of freedom = 25 Null hypothesis: SIPT does not play a foreign accent-mitigating role. The probability of this result, assuming the null hypothesis, is 0.053

#### Group A: Number of items= 15

1.38 1.50 1.62 1.75 2.00 2.12 2.25 2.25 2.38 2.38 2.50 2.50 2.50 2.62 2.75 Mean = 2.17 95% confidence interval for Mean: 1.921 thru 2.413 Standard Deviation = 0.427 Hi = 2.75 Low = 1.38 Median = 2.25 Average Absolute Deviation from Median = 0.333

#### **Group B: Number of items= 12**

1.25 1.38 1.38 1.50 1.50 1.50 1.62 2.00 2.00 2.12 2.50 2.88 Mean = 1.80 95% confidence interval for Mean: 1.527 thru 2.077 Standard Deviation = 0.504 Hi = 2.88 Low = 1.25 Median = 1.56 Average Absolute Deviation from Median = 0.385

# Appendix 24. Levels of the Danish language programme for adult foreigners according to the Common European Language Framework



### Danskuddannelse til voksne udlændinge Niveaumodel

Danskuddannelse 1 Danskuddannelse 2 Da

Danskuddannelse 3

Appenix 25. Recordings